

No. S-236214 Vancouver Registry

# IN THE SUPREME COURT OF BRITISH COLUMBIA

**BETWEEN** 

1392752 B.C. LTD.

**PETITIONER** 

**AND** 

SKEENA SAWMILLS LTD., SKEENA BIOENERGY LTD.
AND ROC HOLDINGS LTD.

RESPONDENTS

FIFTH SUPPLEMENTAL REPORT TO
THE FOURTH REPORT OF THE RECEIVER
ALVAREZ & MARSAL CANADA INC.
April 16, 2024



# TABLE OF CONTENTS

| 1.0 | PURPOSE                               | - 2 - |
|-----|---------------------------------------|-------|
| 2.0 | REVISED KITSUMKALUM OFFER AND THE PSA | - 2 - |
| 3.0 | RECOMMENDATIONS                       | - 6 · |

Appendix A - the Purchase and Sale Agreement between the Receiver and Kitsumkalum First Nation

### 1.0 PURPOSE

- 1.1 This fifth supplemental report (the "Fifth Supplemental Report") has been prepared by the Receiver to provide this Honourable Court with additional information in respect of the revised offer (the "Revised Kitsumkalum Offer") received by the Receiver from the Kitsumkalum First Nation ("Kitsumkalum").
- The Fifth Supplemental Report should be read in conjunction with the First Confidential Report of the Receiver dated January 19, 2024 (the "First Confidential Report"), the Fourth Report of the Receiver dated February 29, 2024 (the "Fourth Report"), the Supplemental Report to the Fourth Report of the Receiver dated March 6, 2024 (the "Supplemental Report"), the Second Supplemental Report to the Fourth Report of the Receiver dated March 11, 2024 (the "Second Supplemental Report"), the Affidavit #2 of Anthony Tillman sworn on March 21, 2024 (the "Second Tillman Affidavit"), the Third Supplemental Report to the Fourth Report of the Receiver dated March 27, 2024 (the "Third Supplemental Report"), and other materials filed in the Receivership Proceedings (collectively, the "Filed Materials"). Background information, including capitalized terms not defined herein, are contained in the Filed Materials.
- 1.3 All monetary amounts contained in this Fifth Supplemental Report are expressed in Canadian dollars unless otherwise noted.

### 2.0 REVISED KITSUMKALUM OFFER AND THE PSA

- 2.1 Subsequent to the Third Supplemental Report, at the April 5, 2024 hearing (the "Hearing"), Kitsumkalum advised this Honorable Court (the "Court") that it would be amending its offer for the purchase of the assets of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd. (collectively, the "Skeena Entities").
- 2.2 Accordingly, following the Hearing, the Receiver, at the Court's direction, engaged with Kitsumkalum to finalize the terms of the Revised Kitsumkalum Offer and develop a form of agreement to be presented to the Court.
- 2.3 On April 16, 2024, the Receiver and Kitsumkalum finalized the form of Purchase and Sale Agreement (the "PSA") for the purchase of the Skeena Entities' assets, a copy of which is attached hereto as Appendix "A".
- 2.4 Tabled below is a summary comparison between Cui Family Holdings Ltd.'s ("Cui Holdings")

  Payment and Retention Agreement (discussed in the Fourth Report and the Third Supplemental

Report, the "PRA") and the PSA. Capitalized terms not defined herein are as defined in the PRA and the PSA, respectively:

|  | Cui Holdings  | Kitsumkalum  |
|--|---|--|
| Bid price                                      | \$13,460,000  Combination of a credit bid and cash, as detailed in the Second Supplemental Report   | \$14,047,000 cash bid  |
| Form of transaction                            | Reverse Vesting Order ("RVO")   | Sale Approval and Vesting Order ("SAVO")   |
| Assets included                                | All of the assets, undertaking and properties of the Skeena Entities, with exceptions (discussed below)                                   | All of the assets, undertaking and properties of the Skeena Entities   |
| Liabilities to be paid out from purchase price | Financing leases, stumpage, source deduction remittances, property taxes (pre-2024)   | Promissory notes due to 1392752 B.C. Ltd., Receiver's Certificates, financing leases, stumpage, source deduction remittances, property taxes (pre-2024)                                    |
| Assumed liabilities/contracts                  | Silviculture liabilities, environmental liabilities, permits and licenses associated with the operations, Collective Bargaining Agreement | Silviculture liabilities, environmental liabilities, permits and licenses associated with the operations, Collective Bargaining Agreement  Future Obligations of the Replaceable Contracts |
| Excluded contracts                             | Replaceable Contracts   | Nil  |
| Deposit received by<br>Receiver                | \$400,000 (approximately 10% of the cash bid portion)   | \$1,320,000  |
| Closing Date                                   | By April 19, 2024   | Two stage closing:   |

|  | Cui Holdings    | Kitsumkalum  |
|--|-----------------|--|
|  |                 | 1) Closing of the purchase and sale of the "Phase 1 Purchased Assets", i.e. all assets except the Tenures, the associated permits and licences and the Replaceable Contracts: two weeks after the granting of the SAVO (the "Phase 1 Closing Date"); and  2) Closing of the purchase and sale of the Tenures, the associated permits and licences and the Replaceable Contracts (the "Phase 2 Purchased Assets"): five (5) days after the Province of British Columbia provides notice of whether or not it has approved the transfer of the Tenures to Kitsumkalum. |
| Payment of the cash portion of the bid   | On Closing Date | On the Phase 1 Closing Date  |
| Obligation to fund<br>the Receivership<br>Proceedings until<br>Tenures can be<br>transferred |                 | Receivership costs to be funded from the funds paid by Kitsumkalum   |

2.5 As noted in the Third Supplemental Report, the Receiver requested that Kitsumkalum provide an indication of their financial wherewithal to close a transaction. Written confirmation was provided by Kitsumkalum on March 28, 2024.

## 2.6 The Receiver notes the following:

- the PSA provides a two-stage closing wherein the Phase 1 Purchased Assets will vest in Kitsumkalum (or its nominee company) two weeks after the SAVO is granted, while the Phase 2 Purchased Assets will be held by the Receiver while the Ministry of Forests conducts the Tenure transfer approval process. By doing so, the cost of receivership will be reduced significantly as the real estate, sawmill, bioenergy plant, and various equipment, among other things, will have been transferred to Kitsumkalum shortly after Court approval. As noted in the Second Supplemental Report, monthly holding costs for the Skeena Entities, including insurance, contractor costs, utilities and other site expenses total approximately \$200,000 \$225,000/month. These costs will be largely (if not totally) eliminated after the Phase 1 Closing Date:
- b) the Receiver expects future professional fees to decline significantly upon the transfer of the Phase 1 Purchased Assets as its remaining role would be to hold the Tenures, assist the Ministry of Forests with its approval process, facilitate the closing of the sale of the Phase 2 Purchased Assets, and plan and apply for a distribution from the sale proceeds;
- c) the payment of the purchase price under the PSA is not dependent on the Ministry's decision whether to approve the transfer of Tenures. This arrangement removes the closing risk under the Kitsumkalum's earlier offer which was identified by the Receiver in the Third Supplemental Report;
- d) Kitsumkalum is to assume all future obligations under the Replaceable Contracts but not any obligations owing thereunder as at the Phase 2 Closing Date. The Receiver understands that Kitsumkalum and the parties holding the Replaceable Contracts, being Timber Baron Contracting Ltd. and Terrace Timber Ltd. (collectively the "Contractors"), have entered into agreements reflecting the foregoing, and that the Contractors are supportive of the PSA;
- e) Under the PSA, Court approval is sought in the form of a SAVO. Due to the change of ownership under the proposed structure in the PSA, the transfer of Tenures will be subject to regulatory approval, including the duty to consult, pursuant to the *Forest Act* and associated regulations. While the RVO proposed by Cui Holdings was opposed by the Department of Justice of Canada, the Province of British Columbia and various First Nations, the Receiver believes the proposed SAVO will address the concerns expressed by the aforementioned stakeholders; and
- f) under the PSA, the Petitioner will be paid the full amount owing in respect of the funds advanced by it under the Receiver's Borrowing Certificates, and the Petitioner's secured debt pursuant to the promissory notes, totaling approximately \$7.4 million plus interest, is

anticipated to be largely paid from the sales proceeds. According to the Receiver's preliminary calculation and certain assumptions, the Receiver projects a recovery exceeding 90% for the Petitioner's proven pre-filing secured debt (including interest). The Receiver is waiting to hear from the Petitioner whether, in light of the Revised Kitsumkalum Offer, it does not oppose the PSA.

- 2.7 The Receiver also notes the following when considering the *Soundair* test:
  - a) The Receiver is of the view that it has made sufficient effort to get the best price and has not acted improvidently. As described in the Filed Materials, the Sale Process was fair and transparent and resulted in the Skeena Entities' assets being marketed extensively. The Receiver initially contacted 146 parties on October 31, 2023. 31 parties signed confidentiality agreements and were provided access to the dataroom set up by the Receiver. 28 parties accessed the dataroom, and several site visits were conducted at the Skeena Entities' facilities in Terrace, B.C. Nine expressions of interest were submitted and four parties were invited to submit definitive bids. Further efforts to invite offers from potentially interested parties were made by the Receiver, at this Honourable Court's direction, in March and April 2024;
  - b) The Receiver has considered the interests of all parties:
    - i. as noted above, the Receiver is of the view that the concerns of the stakeholders who opposed the Cui Holdings' offer and proposed RVO (i.e. the Contractors, the Department of Justice of Canada, the Province of British Columbia and various First Nations) will have been addressed by the terms of the PSA and the SAVO;
    - ii. the Petitioner will recover approximately90% of its proven secured claim (including interest);
    - iii. no other stakeholder will be materially worse-off under the PSA than under the PRA, and the Receiver expects that equipment lessors and the first lien holder over current assets will receive higher realizations under the PSA;
  - c) the Receiver believes the Sale Process was properly and fairly conducted and believes in the efficacy and integrity of the process by which offers were obtained; and
  - d) the Receiver believes that at all times it has conducted itself fairly and reasonably.

### 3.0 RECOMMENDATIONS

3.1 In light of the Revised Kitsumkalum Offer, the Receiver is no longer able to recommend that the Court grant the reverse vesting order sought under its Notice of Application filed February 29, 2024. Among other things, with reference to the factors enumerated in *Harte Gold Corp (Re)*, 2022 ONSC 653, the Receiver notes that there is now before the Court a viable alternative which is both

financially superior to the Cui Holdings' offer and better addresses the financial or other concerns of many stakeholders. In the circumstances, the Receiver recommends that this Honourable Court approve the Revised Kitsumkalum Offer.

3.2 Given the above considerations, the Receiver respectfully recommends that this Honourable Court grant the SAVO and approve the PSA.

\*\*\*\*

All of which is respectfully submitted to this Honourable Court this 16th day of April, 2024.

# Alvarez & Marsal Canada Inc.,

in its capacity as Receiver of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd. and not in its personal capacity

Per: Anthony Tillman

Senior Vice President

Per: Pinky Law

Vice President

# Appendix A -

the Purchase and Sale Agreement between the Receiver and Kitsumkalum First Nation

# Appendix A -

the Purchase and Sale Agreement between the Receiver and Kitsumkalum First Nation

### **PURCHASE AND SALE AGREEMENT**

THIS AGREEMENT is dated for reference April 15, 2024 and is made

#### AMONG:

ALVAREZ & MARSAL CANADA INC., in its capacity as Receiver of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd.

(the "Receiver")

AND:

KITSUMKALUM FIRST NATION

("Kitsumkalum")

AND:

1355392 B.C. LTD.

("392")

AND:

1355390 B.C. LTD.

("390")

AND:

1355387 B.C. LTD.

("387" and together with Kitsumkalum, 392, 390 and 387, the "Purchaser")

#### BACKGROUND:

- A. Skeena Sawmills Ltd. ("Sawmills"), Skeena Bioenergy Ltd. ("Bioenergy") and ROC Holdings Ltd. ("ROC", and together with Sawmills and Bioenergy, the "Companies") own certain assets, undertakings and property, including real property located in Terrace, British Columbia, on which they operated a sawmill (including an industrial sawmill, a certified weight log scale, a natural gas kiln, two bay garage mobile shop, a millwright shop and various tools and equipment) and a bioenergy operation (including a pellet plant and various tools and equipment) and including certain forest licenses under which they conducted forest management and woodlands activities (collectively, the "Business");
- B. On September 20, 2023, on the application of 1392752 B.C. Ltd. (the "Petitioner"), the Supreme Court of British Columbia (the "Court") made an order (the "Receivership Order")

- appointing Alvarez & Marsal Canada Inc. as receiver of the assets, undertakings and properties of the Companies (the "**Property**") and authorized the Receiver to, among other things, sell the Property;
- C. The Purchaser has agreed to purchase, and the Receiver has agreed to sell, the Phase 1 Purchased Assets and the Phase 2 Purchased Assets (each as hereinafter defined) in consideration for the payment of the Purchase Price all on the terms and conditions of this agreement (the "Transaction").

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are acknowledged by each of the parties, the parties agree as follows:

#### ARTICLE 1 - INTERPRETATION

- **1.1 Definitions**. In this Agreement:
  - (a) "Accounting Standards" means at any time the accounting standards for private enterprises so described and established by the Accounting Standards Board which are applicable at such time.
  - (b) "Approved Contracts" means the following Contracts:
    - (i) the CBA; and
    - (ii) all of the Permits and Licenses which are Contracts, including, but not limited to, any licence of occupation granted by a Governmental Authority to any one or more of the Companies.
  - (c) "Approved Licences" has the meaning given to it in Section 2.8(d).
  - (d) "Assignment Order" means an order or orders of the Court, in form and substance satisfactory to the Purchaser, acting reasonably: (i) assigning any Consent Required Contract for which a consent, approval or waiver necessary for the assignment of such Consent Required Contract has not been obtained, (ii) preventing any counterparty to such Consent Required Contracts from exercising any right or remedy under such Consent Required Contracts by reason of any defaults arising from the receivership of the Receiver and (iii) the vesting in the Purchaser of all right, title and interest of the Receiver in such Consent Required Contracts.
  - (e) "Balance" has the meaning given to it in Section 2.4(a)(ii).
  - (f) "Bioenergy" has the meaning given to it in Recital A.
  - (g) "Buildings" means all building(s) and improvements located on the Lands.
  - (h) "Business" has the meaning given to it in Recital A.
  - (i) "Business Day" means any day that is not a Saturday, Sunday, Boxing Day, Easter Monday or statutory holiday in British Columbia.

- "Business Records" means all documents, files, records, reports, agreements, plans, (j) specifications, drawings, surveys, correspondence, licenses and permits in the possession or control of the Receiver or the Companies relating to the Business or any of the Purchased Assets, including without limitation: all digital records; client data; copies of all Contracts; copies of the Permits and Licenses; tax notices and assessments for the Lands; plans and surveys of the Lands and Buildings; copies of all Warranties: a non-itemized description of all Machinery and Equipment; a list of all Intellectual Property including registration details in respect of all Intellectual Property for which registration in any public office has been made; operating statements and financial statements for the most recent two fiscal periods; detailed general ledgers; detailed sales reports relating to the Business; building condition reports, structural reports, mechanical and life safety reports, environmental reports or assessments; notices or orders received from any agency having authority over the Property, the Purchased Assets or the Business; reasonable evidence of the Receiver's insurance relating to the Property, the Purchased Assets or the Business; current accounts receivable reports in respect of the Business.
- (k) "Canter Line" means the 1999 Optimil canter line with serial/VIN/DOT Number 63286 secured by a loan agreement and specific security agreement dated August 17, 2021 in favour of Dynamic Capital Equipment Finance Inc..
- (l) "CBA" means the collective bargaining agreement between the Union and Sawmills.
- (m) "Closing" means the successful completion of the Transaction.
- (n) "Closing Documents" has the meaning given to it in Section 9.4.
- (o) "Closing Payment" has the meaning given to it in Section 9.9.
- (p) "Companies" has the meaning given to it in Recital A.
- (q) "Consent Required Contract" has the meaning set out in Section 6.2.
- (r) "Contaminant" means any substance that when released into the Environment is likely to cause, at some immediate or future time, material harm or degradation to the Environment or material risk to human health. Without restricting the generality of the foregoing, "Contaminant" includes radioactive material, microwaves, urea formaldehyde foam insulation, asbestos, PCBs (polychlorinated biphenyls), PCE (perchloroethylene), TCE (trichloroethylene) and any other substance or materials declared or deemed to be hazardous, toxic or an atmospheric contaminant in or pursuant to any Environmental Laws.
- (s) "Contracts" means all contracts or agreements relating to the use or operation of the Property, the Purchased Assets, or any part thereof or the operation of the Business to which any of the Companies is a party, including, without limitation, purchase and sale agreements, options to purchase, material supply contracts, contracts relating to the operation, maintenance, cleaning, security, signage, fire protection or servicing of the Property or any part thereof made by or on behalf of one or more of the Companies.

- (t) "Court" has the meaning given to it in Recital B.
- (u) "Deposit" means the sum of One Million Three Hundred Twenty Thousand (\$1,320,000) Dollars paid to the Receiver pursuant to Section 2.4(a) and held in accordance with the terms of Section 2.7.
- (v) "Employees" means an individual who is or was employed by the Receiver or the Companies, whether on a full-time or a part-time basis, whether active or inactive as of the Closing Date, and includes an employee on short term or long term disability leave.
- (w) "Encumbrance" means any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, trusts or deemed trusts (whether contractual, statutory, or otherwise), liens, executions, levies, charges, or other financial or monetary claims, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured or otherwise including, without limiting the generality of the foregoing:
  - (i) any encumbrances or charges created by the Receivership Order;
  - (ii) all charges, security interests or claims evidenced by registrations pursuant to the *Personal Property Security Act* of British Columbia or any other personal property registry system;
  - (iii) any legal notation, charge, lien, interest or other encumbrance or title defect of whatever kind or nature, regardless of form; and
  - (iv) any agreement, lease, license, option or claim, easement, right of way, restriction, execution or other encumbrance (including any notice or other registration in respect of any of the foregoing) affecting title to or the ownership of the Purchased Assets or any part thereof or interest therein.
- (x) "Environment" means the environment or natural environment as defined in any Environmental Law and includes air, land, surface water, ground water or other water, land surface, soil, subsurface strata, or any sewer or water system.
- "Environmental Laws" means all applicable federal, provincial, municipal or local statutes, regulations, by-laws, rules, orders-in-council, and codes having the force of law, in each case pertaining in full or in part to the Environment or any Contaminant and includes those relating to the storage, generation, use, handling, manufacture, processing, transportation, treatment, releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, depositing, injecting, escaping, leaching, disposing or dumping, disposal, remediation, monitoring, investigation or other action in relation to any substances, including Contaminants, which are in effect from time to time.
- (z) "Excluded Assets" means: (i) Contracts which are not Approved Contracts; (ii) the rights of the Companies and the Receiver under this Agreement; (iii) corporate income taxes receivable and GST refunds; (iv) cash and cash equivalents; (v) the Canter Line; and (vi) all shares of capital stock or other equity interest in any entity, including the Companies.

- (aa) "Excluded Liabilities" means any and all Liabilities of the Companies that are not expressly assumed by the Purchaser under this Agreement, including without limitation:
  - (i) any taxes of or relating to the Business or the Purchased Assets, including statutory deductions and remittances, GST, and sales taxes, in respect of any period up to and including the Closing Date (including penalties, fines and interest);
  - (ii) any Liabilities or Encumbrances in respect of any claims, demands, actions, complaints, causes of action, proceedings, charges, assessments or litigation involving the Receiver, the Companies, the Purchased Assets, the Excluded Assets, or the Business commenced or threatened or resulting from any event or circumstance prior to the Closing Date;
  - (iii) any other Encumbrances made, filed, claimed, perfected or otherwise arising or resulting from any event or circumstance prior to the Closing Date charging the Purchased Assets or the Excluded Assets;
  - (iv) any Liabilities owing to or Encumbrances claimed by or in favour of any Employees or the Union that relate to any period prior to and including the Closing Date, whether pursuant to the CBA or otherwise, including without limitation outstanding salaries, wages, vacation pay and bonuses owing to any Employees, any severance or other termination obligations, including payment in lieu of notice, and any Liability for employer health tax payable;
  - (v) all executive personnel agreements, officer or director agreements, employee benefit plans or payments, pension obligations, employee tax withholding obligations, employee health or dental plan obligations, all employee complaints or claims, labour relations board actions or other employee proceedings and similar obligations of the Companies;
  - (vi) any Liabilities of the Companies arising prior to the Closing Date including, but not limited to, Liabilities owed to lenders, service contractors or third parties of any kind including all Liabilities under all Contracts other than the Approved Contracts, and subject to Section 2.8(d), the Approved Licences and the Replaceable Contracts;
  - (vii) any Liabilities relating to or arising out of the Property or Business which are not being assumed by the Purchaser, including, without limitation, Liabilities for terminating, not complying with or defaulting under any Contract other than the Approved Contracts, and subject to Section 2.8(d), the Approved Licences and the Replaceable Contracts;
  - (viii) all Liabilities for payment of fees for operation of the Purchased Assets or Excluded Assets up to the Closing Date;
  - (ix) any Liabilities for a breach or non-compliance with any applicable law; and

- (x) any other obligations which pursuant to the terms and conditions of this Agreement, remain the Liability of the Receiver after the completion of the transactions contemplated herein including, without limitation, the obligations of the Receiver under this Agreement.
- (bb) "Execution Date" means the date that this Agreement is executed by both the Receiver and the Purchaser.
- "Governmental Authority" means (i) any domestic or foreign government, whether national, federal, provincial, state, territorial, municipal or local (whether administrative, legislative, executive or otherwise), (ii) any agency, authority, ministry, department, regulatory body, court, central bank, bureau, board or other instrumentality having legislative, judicial, taxing, regulatory, prosecutorial or administrative powers or functions of, or pertaining to, government, (iii) any court, tribunal, commission, individual, arbitrator, arbitration panel or other body having adjudicative, regulatory, judicial, quasi-judicial, administrative or similar functions, and (iv) any other body or entity created under the authority of or otherwise subject to the jurisdiction of any of the foregoing, including any stock or other securities exchange or professional association.
- (dd) "GST" means goods and services tax and harmonized sales tax payable under the Excise Tax Act (Canada).
- (ee) "GST Certificate" has the meaning given to it in Section 10.1.
- "Intellectual Property" means all intellectual property and proprietary rights of any (ff) kind currently owned by any of the Companies pertaining to the Business, including the following: (i) trademarks, service marks, trade names, slogans, logos, designs, symbols, trade dress, internet domain names, uniform resource identifiers, rights in design, brand names, any fictitious names, d/b/a's or similar filings related thereto, or any variant of any of them, and other similar designations of source or origin, together with all goodwill, registrations and applications related to the foregoing; (ii) copyrights and copyrightable subject matter (including any registration and applications for any of the foregoing); (iii) trade secrets and other confidential or proprietary business information (including manufacturing and production processes and techniques, research and development information, technology, intangibles, drawings, specifications, designs, plans, proposals, technical data, financial, marketing and business data, pricing and cost information, business and marketing plans, customer and supplier lists and information), know how, proprietary processes, formulae, algorithms, models, industrial property rights, and methodologies; (iv) computer software, computer programs, and databases (whether in source code, object code or other form); (v) all rights to sue for past, present and future infringement, misappropriation, dilution or other violation of any of the foregoing and all remedies at law or equity associated therewith; and (vi) all websites and all telephone and facsimile numbers.
- (gg) "Interim Period" means the period commencing on the Execution Date until and including the Phase 1 Closing Date.

- (hh) "Inventory" means the inventory of the Companies located on the Lands or at the logging sites operated by any of the Companies as more particularly described in Schedule C.
- (ii) "Lands" means the lands and premises owned by ROC and used by the Companies in the Business and which are more particularly described in Part 1 of Schedule A (and which include the Stump Dump).
- (jj) "Liability" means, any debts, claim, liability, duty, responsibility, obligations, commitment, assessment, cost, expense, loss, expenditure, charge, fee, penalty, fine, contribution or premium of any kind or nature whatsoever, whether known or unknown, asserted or unasserted, absolute or contingent, direct or indirect, or due or to become due and regardless of when sustained, incurred or asserted or when the relevant events occurred or circumstances existed.

### (kk) "Licences" means

- (i) Tree Farm Licence 41, Forest Licence A16882, Forest Licence A16885 and cutting permits associated with the foregoing licences; and
- (ii) Road and Bridge permits R03012, R03036, R04088, R06006, R06007, R06207, R06273, R07185, R07521, R07570, R07799, R08791, R09609, R09697, R12655, R12656, R13212, R13213, R13578, R13750, R14658, R14664, R14742, R16154, R18835, R19749, R19867, R21229, R21237, R21396, R21564, R21604, R21723, R21793, R21962, R22071, R22094, R22191, R22824, R22948, R23124, R23143, R23213, R23276, R23279, R23281, R23436, R23700, R23721, R23735, R23736, R23850, R23885, and R23912.
- (II) "Machinery and Equipment" means, to the extent the same is in the possession of the Receiver, all of the Companies' machinery, equipment (including trucks and other motor vehicles), transportation equipment, spare parts, tools, jigs, dies, office equipment, furniture whether moveable or built-in, computer hardware, fixtures, furnishings and accessories and supplies of all kinds and any other personal property owned by any of the Companies used in the maintenance, management or operation of the Property or any part thereof or the operation of the Business, as more particularly set out in Schedule D.
- (mm) "Mutual Conditions" has the meaning given to it in Section 8.2.
- (nn) "Non-Assignable Interests" means any Phase 1 Purchased Assets which, by their nature cannot be legally or practically sold and assigned by the Receiver to the Purchaser hereunder, including without limitation any Consent Required Contracts for which an Assignment Order or counterparty consent has not been obtained.
- (00) "Parties" means the Receiver and the Purchaser, and "Party" means either one of them.
- (pp) "Permits and Licenses" means the following licenses, approvals, authorizations, permits, consents or other rights entered into or obtained by any of the Companies

or the Receiver from any Governmental Authority, and used in connection with the Business or in respect of any of the Purchased Assets:

- (i) Licence of Occupation SK926021;
- (ii) Licence of Occupation SK929701;
- (iii) Licence of Occupation SK932307;
- (iv) Air Discharge permit 109209;
- (v) Certificate of Inspection permit 718765 (BC Hydro);
- (vi) Gas Operating Permit GA-900885-2019;
- (vii) Air Permit PA-3008;
- (viii) Landfill Permit 3360;
- (ix) Boiler and Pressure Vessels Permits 4878750, 4878751, 4878754, 4878755, 4878756, 4878757, 4878758, 4878759, 4878760, 4878761, 4878765, 4878762, 4878763, 4878764, 4878753, 4878752, BPV-885770-2019, BPV-885770-2019; and
- (x) Electrical Permit 5148516,

but for certainty, excluding the Licences.

- (qq) "Permitted Encumbrances" means the Encumbrances set out in Part 2 of Schedule A.
- (rr) "Petitioner" has the meaning given to it in Recital B.
- (ss) "Phase 1 Closing Date" means the date that is fourteen (14) days after the date the Vesting Order is pronounced by the Court, provided that if the Land Title Office is not open on such date, the Closing Date will be on the next Business Day, or any other date as may be agreed in writing by the Receiver and the Purchaser.
- (tt) "Phase 1 Purchased Assets" means all the Companies' right, title and interest, in and to the following:
  - (i) the Approved Contracts;
  - (ii) the Business Records;
  - (iii) the Intellectual Property;
  - (iv) the Inventory;
  - (v) the Machinery and Equipment;
  - (vi) the Permits and Licenses;

- (vii) the Lands and Buildings; and
- (viii) the Warranties.
- (uu) "Phase 2 Closing Date" means the first Business Day that is five (5) days after the date the Province of British Columbia provides notice of whether or not it has approved the transfer of the Licences to the Purchaser.
- (vv) "Phase 2 Purchased Assets" means all the Companies' right, title and interest, in and to the following:
  - (i) the Licences (but not including any Rejected Licences); and
  - (ii) the Replaceable Contracts.
- (ww) "Property" has the meaning given to it in Recital B.
- (xx) "PST" has the meaning given to it in Section 10.2
- (yy) "Purchase Price" has the meaning given to it in Section 2.3.
- (zz) "Purchased Assets" means collectively, the Phase 1 Purchased Assets and the Phase 2 Purchased Assets.
- (aaa) "Purchaser" is as defined in the preamble hereto.
- (bbb) "Purchaser's Condition" has the meaning given to it in Section 8.1.
- (ccc) "Purchaser's Solicitors" means ATAC Law Corporation or such other firm of solicitors or agents as are retained by the Purchaser from time to time and written notice of which is provided to the Receiver.
- (ddd) "Receiver" is as defined in the preamble hereto.
- (eee) "Receivership Order" has the meaning given to it in Recital B.
- (fff) "Receiver's Solicitors" means Fasken Martineau DuMoulin LLP.
- (ggg) "Rejected Licences" has the meaning given to it in Section 2.8(d).
- (hhh) "Replaceable Contract Holders" means collectively, Terrace Timber Ltd. and Timber Baron Contracting Ltd..
- (iii) "Replaceable Contracts" means:
  - (i) the replaceable coast stump to dump harvest timber harvesting contract (five year) made as of January 1, 2020 between Sawmills and Terrace Timber Ltd.; and
  - (ii) the replaceable interior timber harvesting contract (five year) made as of January 1, 2021 between Sawmills and Timber Baron Contracting Ltd..

- (jjj) "ROC" has the meaning given to it in Recital A.alvar
- (kkk) "Sawmills" has the meaning given to it in Recital A.
- (III) "Stump Dump" means the lands located at 863 Kalum Lake Road, Terrace, BC and legally described as PID: 009-426-833 District Lot 1398 Range 5 Coast District Except Plan 11735.
- (mmm) "Transaction" has the meaning given to it in Recital C.
- (nnn) "Union" means United Steelworkers Local 1-1937.
- (000) "Vendor Group Parties" has the meaning given to it in Section 2.2.
- (ppp) "Vesting Order" has the meaning given to it in Section 8.2.
- (qqq) "Warranties" means all subsisting warranties and guarantees benefiting any of the Purchased Assets or any part thereof that are assignable without consent and in effect on the Closing Date.

# ARTICLE 2 - AGREEMENT TO COMPLETE TRANSACTIONS

- **2.1 Agreement**. Subject to the terms and conditions of this Agreement and based on the representations and warranties contained in this Agreement, the Receiver agrees to sell and the Purchaser agrees to purchase:
  - (a) the Phase 1 Purchased Assets on the Phase 1 Closing Date; and
  - (b) subject to Section 2.8(d), the Phase 2 Purchased Assets on the Phase 2 Closing Date,

for the Purchase Price, free and clear of all Excluded Liabilities, and, for greater certainty, free and clear of all Encumbrances except for the Permitted Encumbrances. For avoidance of doubt, the Purchaser will not retain or assume, directly or indirectly, any liability in respect of the Excluded Liabilities.

As Is, Where Is. The Purchaser is purchasing the Purchased Assets on an "as is, where is" 2.2 basis as of the Phase 1 Closing Date and the Phase 2 Closing Date, as applicable. Neither the Receiver, nor anyone on its behalf, represents or warrants the condition or state of repair of any of the Purchased Assets. The Purchaser acknowledges and agrees that the listings and descriptions of the Purchased Assets contained herein have been prepared solely for convenience and are not warranted to be complete or accurate and are for the purposes of identification only, and that the Purchaser is responsible for satisfying itself through independent investigation and verification as to the completeness and accuracy of such listings and descriptions. The Purchaser must satisfy itself, and accept the Purchased Assets on a strictly "as is, where is" basis on the terms of this Agreement. The Purchaser does hereby release the Receiver and the Companies and each of their respective directors, officers, agents and employees (collectively, the "Vendor Group Parties") from any and all claims that the Purchaser has or may have against any of the Vendor Group Parties in connection with the condition of the Purchased Assets irrespective of whether such claim arose before or after the Purchaser's acquisition of the Purchased Assets pursuant to this Agreement.

2.3 Purchase Price. The Purchase Price to be paid by the Purchaser to the Receiver in consideration of the completion of the Transaction in accordance with this Agreement will be the sum of Fourteen Million Forty-Seven Thousand Dollars (\$14,047,000) (the "Purchase Price"), exclusive of applicable taxes.

# 2.4 Payment of Purchase Price.

- (a) The Purchase Price will be payable as follows:
  - (i) by payment of the Deposit, which has been paid to the Receiver and will be held in accordance with Section 2.7; and
  - (ii) provided that the Purchaser's Condition and Mutual Condition have been satisfied or waived in accordance with Sections 8.1 and 8.2, by the Purchaser paying cash on the Phase 1 Closing Date in the amount of Twelve Million Seven Hundred Twenty-Seven Thousand Dollars (\$12,727,000) (the "Balance").
- (b) Except as otherwise provided herein, the Parties agree that there will be no adjustments for expenses and liabilities and revenues accrued in respect of the Purchased Assets.
- (c) The Purchase Price shall be paid by wire transfer or certified cheque of immediately available funds.
- 2.5 Allocation of Purchase Price. The parties agree to use reasonable efforts to agree prior to the Phase 1 Closing Date on an allocation of the Purchase Price among the components of the Purchased Assets. However, the parties further agree that failure to agree on such an allocation prior to the Phase 1 Closing Date will not render this Agreement unenforceable or result in a termination of this Agreement, and in such case each of the Receiver and the Purchaser will make its own determination of allocation.
- 2.6 Bidding Procedures. The Receiver and the Purchaser acknowledge that this Agreement and the Transaction are subject to Court approval. The Purchaser acknowledges and agrees that other prospective purchasers may attend in Court in person or by agent at the hearing of the motion to approve this Agreement and such prospective purchasers may make competing offers which may be approved by the Court. The Receiver may be compelled to advocate that the Court consider other offers in order to obtain the highest price for the Purchased Assets. The Receiver gives no undertaking to advocate the acceptance of this Agreement. To protect his, her or its interest in the Transaction, the Purchaser acknowledges and agrees that he, she or it should attend at the Court hearing in person or by agent and be prepared there to make such amended or increased offer as the Court may permit or direct.
- **Deposit.** The Deposit will be held by the Receiver in an interest-bearing trust account with interest to the account of the Purchaser, unless otherwise stated in this Agreement. The Deposit and any accrued interest will be dealt with as follows:
  - on the Closing Date, the Deposit will be credited on account of the Purchase Price and the Receiver will pay to the Purchaser the accrued interest; or

- (b) if, after the Purchaser's Condition in Section 8.1 and the Mutual Conditions in Section 8.2 herein are satisfied or waived, the Purchaser fails to complete the purchase of the Phase 1 Purchased Assets on the Phase 1 Closing Date in accordance with this Agreement or if the Purchaser repudiates this Agreement, then the Deposit together with accrued interest will be forfeited to the Receiver as liquidated damages, as the Receiver's sole and exclusive remedy; or
- (c) if the Purchaser's Condition in Section 8.1 or the Mutual Conditions in Section 8.2 herein are not satisfied or waived within the applicable time periods contemplated in Sections 8.1 and 8.2, the Deposit, together with accrued interest will be returned to the Purchaser forthwith without any deduction;
- (d) if the Purchaser is not in default of any of its obligations under this Agreement and the Receiver fails to complete the sale of the Phase 1 Purchased Assets in accordance with this Agreement or if the Receiver repudiates this Agreement, then the Receiver will return the Deposit, together with accrued interest to the Purchaser upon demand by the Purchaser on or after the Phase 1 Closing Date, as the Purchaser's sole and exclusive remedy.

### 2.8 Licences.

- (a) The Receiver will promptly make application to the Province of British Columbia for consent to the transfer of the Licences to the Purchaser upon completion of the purchase and sale of the Phase 1 Purchased Assets.
- (b) The Purchaser covenants and agrees to provide, upon request from the Receiver, all information required by the Receiver or the Province of British Columbia with respect to the application to the Province of British Columbia for consent to the transfer of the Licences to the Purchaser upon completion of the purchase and sale of the Phase 2 Purchased Assets. The Purchaser will work cooperatively with the Receiver in making such application and answering any questions from the Province of British Columbia with respect to such application.
- (c) The Purchaser acknowledges and agrees that the consent of the Province of British Columbia to the transfer of the Licences to the Purchaser is not a condition for payment of the Balance and the Purchaser is obligated to pay the Balance on the Phase 1 Closing Date and will not be entitled to any refund of any portion of the Purchase Price even if the Province of British Columbia subsequently rejects the application for transfer of the Licences or fails to respond to the application prior to the Phase 2 Closing Date.
- (d) If the Province of British Columbia rejects the application for transfer of all of the Licences or fails to respond to the application prior to the Phase 2 Closing Date, the obligation of the Receiver to sell and the obligation of the Purchaser to purchase the Phase 2 Purchased Assets will be at an end and all of the representations, warranties and covenants contained herein with respect to the Phase 2 Purchased Assets will merge and there will be no survival of any representation, warranties or covenants contained in this Agreement with respect to the Phase 2 Purchased Assets except for any provision of this Agreement expressly contemplating obligations to be observed or performed by a Party after Closing. If the Province of British Columbia rejects the

application for transfer of only some of the Licences (the "Rejected Licences") but approves the transfer for the balance of the Licences (the "Approved Licences"), then the obligation of the Receiver to sell and the obligation of the Purchaser to purchase the Rejected Licences will be at an end, but the Receiver and the Purchaser will proceed with the purchase and sale of the Approved Licences as part of the Phase 2 Purchased Assets.

#### ARTICLE 3 - DOCUMENTS AND INSPECTION

- 3.1 Business Documents. The Receiver covenants that it has made and until the Phase 1 Closing Date will make available to the Purchaser full, accurate and complete copies of all Business Records within the Receiver's possession or control, and the Purchaser will be entitled to make photocopies of such of the material in those files as the Purchaser may reasonably request. If for any reason the Transaction contemplated in this Agreement is not completed, then the Purchaser will promptly return any such Business Records without retaining any copies.
- 3.2 Inspection. The Purchaser and its advisors will be entitled upon reasonable notice to the Receiver and in accordance with the Receiver's reasonable requirements as to security to enter the Lands and carry out tests and inspections of the Purchased Assets, provided that such access for such purposes will be at reasonable times scheduled by the Receiver at the Purchaser's request and, at the option of the Receiver, subject to the Receiver's supervision. The Purchaser will be responsible for and indemnify the Receiver for all costs, injuries or damages to the Purchased Assets, or to the Receiver, its agents or employees, directly arising out of such entry by the Purchaser and such indemnity will survive the completion of the Transaction or earlier termination of this Agreement. In carrying out such tests and inspections and entry the Purchaser will not disrupt or unduly interfere with the Business carried out on the Lands and Buildings.
- 3.3 Authorization. The Receiver hereby authorizes the Purchaser and its agents, consultants and advisors to meet with or correspond with appropriate statutory or governmental authorities having jurisdiction over the Purchased Assets, the Lands, or the Companies or the Receiver for the purposes of this Transaction, including but not limited to inquiries with respect to compliance with laws, by laws, regulations and assessments. The Receiver will promptly, at the Purchaser's request, execute and deliver any authorizations reasonably required by the Purchaser to authorize the statutory or governmental authorities to release information to the Purchaser, provided such authorizations explicitly do not authorize or request any inspections with respect to the Lands and Buildings.

#### **ARTICLE 4 - GENERAL COVENANTS**

## **4.1 Covenants of the Receiver.** The Receiver:

(a) throughout the Interim Period will notify the Purchaser of any material changes to the information delivered or made available to the Purchaser under or in connection with this Agreement, and after the expiration of the Interim Period up until the Phase 2 Closing Date, will notify the Purchaser of any material changes to the information delivered or made available to the Purchaser in under this Agreement in connection with the Phase 2 Assets;

- (b) throughout the Interim Period will maintain in full force and effect all existing policies of insurance currently maintained by the Receiver and maintain insurance on all the Phase 1 Purchased Assets at least to the levels as they are insured on the date of this Agreement;
- (c) throughout the Interim Period will not enter into any commitment or agreement or Contract, any agreement to lease, offer to lease or lease the Purchased Assets or modify any material terms of or terminate any of the Contracts, Licences, Replaceable Contracts, Permitted Encumbrances, Permits and Licenses or any mortgage or charge relating to the Purchased Assets or that would form an Encumbrance on the Purchased Assets without the prior written consent of the Purchaser, which the Purchaser may withhold in its sole discretion, or without order of the Court;
- (d) throughout the period from the expiration of the Interim Period until the Phase 2 Closing Date, will not modify any material terms of or terminate any of the Licences or Replaceable Contracts without the prior written consent of the Purchaser, which the Purchaser may withhold in its sole discretion, or without order of the Court;
- (e) will continue to maintain the Approved Contracts, Licences and Replaceable Contracts in their current status and, as would be expected for a receivership of a non-operating business, honour the obligations of any of the Companies or the Receiver under the Approved Contracts, Licences and the Permitted Encumbrances, and enforce the terms of all Approved Contracts, Licences and Permitted Encumbrances as would a prudent owner, subject to the Receiver's powers and obligations under the Receivership Order, and any other Court order or statute;
- (f) will promptly notify the Purchaser if the Receiver becomes aware that, after the date of this Agreement, any covenants, terms or conditions in this Agreement are breached or cannot be performed; and
- (g) will promptly forward to the Purchaser any search results from government offices which are directed to the Receiver in response to any due diligence inquiries made by or at the request of the Purchaser.

### ARTICLE 5 - RISK AND CONDITION

5.1 Risk. The Phase 1 Purchased Assets will be at the risk of the Receiver until completion of the purchase and sale of the Phase 1 Purchased Assets on the Phase 1 Closing Date and thereafter at the risk of the Purchaser. In the event of damage to the Phase 1 Purchased Assets by reason of fire, tempest, lightning, earthquake, flood or other Act of God, fire, explosion, riot, civil commotion, insurrection or war, then the Purchaser will complete the purchase and sale of the Phase 1 Purchased Assets and the Receiver will assign to the Purchaser the proceeds of any insurance claim related to such damage. For certainty, the Receiver is under no obligation to repair any damage to the Phase 1 Purchased Assets that arises as a result of any such damage.

### 5.2 Site Profile and Environmental Condition.

(a) The Purchaser hereby waives and releases the Receiver from any obligation to deliver a site profile or site disclosure statement to the Purchaser for the Lands and the

Buildings as contemplated by the *Environmental Management Act*(British Columbia) or any regulation in respect thereto.

- (b) Without limiting the generality of Section 2.2, the Purchaser acknowledges and agrees that the Receiver makes no representations or warranties with regard to the environmental condition of Lands and the Buildings and the Purchaser is relying solely on in its own investigations and inspections in regard to the environment and the existence on the Lands and the Buildings of any hazardous materials as it deems necessary in satisfying itself as to the environmental condition of the Lands and the Buildings.
- (c) The Purchaser does hereby release the Vendor Group Parties from any and all claims that the Purchaser has or may have against any of the Vendor Group Parties in connection with the environmental condition of the Lands and the Buildings, any contamination or Contaminants located on the Lands and the Buildings or migrating onto the Lands and the Buildings or from the Lands or the Buildings or any breach of any Environmental Laws irrespective of whether such claim arose before or after the Purchaser's acquisition of the Phase 1 Purchased Assets pursuant to this Agreement.
- (d) The Purchaser agrees that it shall not directly or indirectly commence or assert or pursue or threaten to commence, assert or pursue any type of claim (including an order issued by a Governmental Authority) against the Vendor Group Parties or any of the them relating to the environmental condition of or any environmental matter or issue involving the Lands and the Buildings including the location of contaminants or Contaminants thereon or migrating thereon or therefrom.
- (e) Without limiting the generality of the foregoing, the parties acknowledge and agree that any and all costs in any way related to the environmental remediation of the Lands and the Buildings (including any and all costs associated with the disposal of Contaminants or contaminated soil) or in connection with any adjacent property contaminated by Contaminants or contamination migrating from the Lands and the Buildings shall be for the Purchaser's sole account and neither the Receiver nor the Companies shall have any liability or responsibility in connection with any such remediation costs.
- (f) The parties acknowledge and confirm that the provisions of this Section 5.2 constitute an agreement between them that is a private agreement respecting liability for Contaminants and contamination on, in, at or under or released to, at or from Lands and the Buildings or otherwise associated with the Lands and the Buildings and any contamination of adjacent properties and waters resulting from such Contaminants and contamination or remediation of the Lands and Building.

The terms of this Section 5.2 shall survive completion of the sale of the Phase 1 Purchased Assets to the Purchaser.

### ARTICLE 6 - POSSESSION

**Possession Date.** The Purchaser will, upon completion of the purchase and sale of the Phase 1 Purchased Assets have possession of the Phase 1 Purchased Assets as of the Phase 1 Closing Date free and clear of all Encumbrances subject only to Permitted Encumbrances.

- **Non-assignable Assets.** In the event that there are any Approved Contracts which are not assignable in whole or in part without the consent, approval or waiver of another party or parties to them and such consents, approvals or waivers have not yet been obtained as of the Phase 1 Closing Date, then:
  - (a) nothing in this Agreement will be construed as an assignment of any such Contract (each a "Consent Required Contract");
  - (b) until the Vesting Order is granted, the Receiver shall use its commercially reasonable efforts to obtain any such consent, approval or waiver and the Purchaser shall provide its reasonable cooperation to assist the Receiver in obtaining any such consent, approval or waiver;
  - (c) if any consent, approval or waiver is not obtained for any Consent Required Contract prior to the service of the application for the Vesting Order, the Purchaser may request, at the Purchaser's own cost, that the Receiver bring an application to the Court for issuance of an Assignment Order with respect to such Consent Required Contracts together with the motion for the Vesting Order, or at such later date as may be designated by the Purchaser;
  - (d) pending obtaining consent or Assignment Order, the Receiver shall hold the Consent Required Contract in trust for the exclusive benefit of the Purchaser as a Non-Assignable Interest, and shall use commercially reasonable efforts to continue to perform its obligations under the Consent Required Contract and to continue to seek consent, approval or waiver; and
  - (e) once the consent, approval or waiver to the assignment of a Consent Required Contract is obtained or the assignment of such Contract has been ordered by the Court, such Consent Required Contract shall be deemed to be assigned to the Purchaser on the Phase 1 Closing Date.

### ARTICLE 7- REPRESENTATIONS AND WARRANTIES

- 7.1 Purchaser's Representations and Warranties. The Purchaser represents and warrants to the Receiver, regardless of any independent investigation that the Receiver may cause to be made that:
  - (a) Kitsumkalum is an Indian Band within the meaning of the *Indian Act*, R.S.C. 1985 c. I 5, as amended, and enters into this Agreement for and on behalf of itself and all the members of the Kitsumkalum First Nation, as represented by its Chief and Council having an address at PO Box 544, Terrace, British Columbia, V8G 4B5;
  - (b) each of 392, 390 and 387 is a corporation incorporated and existing under the laws of British Columbia and is a subsidiary of Kitsumkalum;
  - (c) each of the entities comprising the Purchaser has the power and authority to enter into this Agreement and to perform its obligations under this Agreement;
  - (d) neither the Purchaser's entering into this Agreement nor the performance of its terms will result in the breach of or constitute a default under any term or provision of any

indenture, mortgage, deed of trust or other agreement to which any party comprising the Purchaser is bound or subject.

#### ARTICLE 8 - CONDITIONS PRECEDENT

- **8.1 Closing Condition Precedent in favour of the Purchaser.** The obligation of the Purchaser to complete the purchase and sale of the Phase 1 Purchased Assets is subject to following condition ("Purchaser's Condition"):
  - (a) the Receiver having performed in all material respects each of its obligations under this Agreement to the extent required to be performed at or before the Phase 1 Closing Date, including the delivery of each of the items required pursuant to Section 9.2.

The foregoing condition is for the exclusive benefit of the Purchaser. The condition in this Section 8.1 may be waived by the Purchaser in whole or in part, without prejudice to any of its rights of termination in the event of non-fulfillment of any other condition in whole or in part. Any such waiver shall be binding on the Purchaser only if made in writing.

- **8.2 Mutual Condition**. The obligation of the parties to complete the Transaction will be subject to the following mutual conditions (collectively, the "**Mutual Conditions**"), for the benefit of both the Receiver and the Purchaser:
  - (a) that on or before the date that is forty-five (45) days following execution of this Agreement by both Parties, the Receiver will have obtained (at the sole cost of the Receiver) an order or orders of the Court substantially in the form set out in **Schedule B** and satisfactory to the Receiver and the Purchaser (collectively, the "**Vesting Order**"):
    - (i) approving the Transaction on the terms of this Agreement;
    - (ii) on completion of the purchase and sale of the Phase 1 Purchased Assets, vesting title to:
      - (A) all Lands except the Stump Dump in and to 392;
      - (B) the Stump Dump in and to 390; and
      - (C) the balance of the Phase 1 Purchased Assets in and to Kitsumkalum,

free and clear of all Encumbrances except the Permitted Encumbrances

- (iii) subject to Section 2.8(d), on completion of the purchase and sale of the Phase 2 Purchased Assets under this Agreement, vesting title to:
  - (A) the Licences in and to 387; and
  - (B) the balance of the Phase 2 Purchased Assets in and to Kitsumkalum,

free and clear of all Encumbrances except the Permitted Encumbrances; and

- (iv) authorizing and directing the Receiver, as vendor, to complete the Transaction; and
- (b) that as of each of the Phase 1 Closing Date and the Phase 2 Closing Date, as applicable, there will be no applicable law or Court order in effect that prohibits the applicable Closing or the applicable part of the Transaction.

The Mutual Conditions are for the mutual benefit of the Receiver and the Purchaser and may not be waived unilaterally by either party. Both parties agree that they will use all reasonable commercial efforts to satisfy the Mutual Conditions. If the Mutual Conditions have not been satisfied by the applicable deadline provided for in this Section 8.2, then the Purchaser's and Receiver's obligation to complete the Transaction (or applicable portion thereof) pursuant to this Agreement will be an end.

If the Vesting Order is made by the Court, then the Parties will complete the Transaction without regard to any appeal or application for leave to appeal to vary or set aside the Vesting Order by any person, unless the Vesting Order has been stayed by further Court order.

#### **ARTICLE 9- CLOSING**

## **9.1 Closing.** The closing of:

- (a) the purchase and sale of the Phase 1 Purchased Assets will take place on the Phase 1 Closing Date; and
- (b) subject to Section 2.8(d), the purchase and sale of the Phase 2 Purchased Assets will take place on the Phase 2 Closing Date,

each by electronic exchange of documents between the Purchaser's Solicitors and the Receiver's Solicitors in accordance with this ARTICLE 9.

# 9.2 Receiver's Closing Documents.

- (a) On or before the Phase 1 Closing Date, the Receiver will deliver, or cause the Receiver's Solicitors to deliver, to the Purchaser's Solicitors in trust to be held in escrow as provided in this Agreement, the following documents duly executed as applicable and all in a form satisfactory to the Purchaser, acting reasonably:
  - (i) Court certified copy of the Vesting Order and any other orders of the Court as are necessary, all in a form registerable in all necessary offices required to effect the Transaction:
  - (ii) an assignment and assumption of Permitted Encumbrances with 392 (as to all Lands except the Stump Dump) and 390 (as to the Stump Dump);
  - (iii) a general conveyance transferring the Business Records, Intellectual Property, Inventory and the Machinery and Equipment to Kitsumkalum;
  - (iv) an assignment of all of the Receiver's and the Companies' rights under any and all Warranties;

- (v) a certificate dated as of the Phase 1 Closing Date of a senior officer of the Receiver having knowledge of the facts certifying, on behalf of the Receiver and without personal liability, that the Receiver's covenants and agreements to be observed or performed on or before the Phase 1 Closing Date pursuant to the terms of this Agreement have been duly observed and performed in all material respects;
- (vi) a statutory declaration by an authorized officer of the Receiver that the Receiver is not a non-resident of Canada for the purposes of the *Income Tax Act* (Canada);
- (vii) a notice from the Receiver to each of the other parties under the Approved Contracts giving notice of the assignment of such Approved Contract;
- (viii) all Business Records that are in the possession or control of the Receiver;
- (ix) all access cards, security cards, keys (including master keys) relating to the Phase 1 Purchased Assets or the Lands and Building which are in the possession or control of the Receiver; and
- (x) such other documents and assurances as may be reasonably required by the Purchaser to give full effect to the intent and meaning of this Agreement or contemplated by the Vesting Order as they relate to the Phase 1 Purchased Assets.
- (b) Subject to Section 2.8(d), on or before the Phase 2 Closing Date, the Receiver will deliver, or cause the Receiver's Solicitors to deliver, to the Purchaser's Solicitors in trust to be held in escrow as provided in this Agreement, the following documents duly executed as applicable and all in a form satisfactory to the Purchaser, acting reasonably:
  - (i) an assignment and assumption of the Licences (not including any Rejected Licences) wherein 387 assumes the rights and obligations under the Licences (not including any Rejected Licences) and the Purchaser indemnifies the Receiver and the Companies for all Liability under the Licences (not including any Rejected Licences) (but which for certainty will not include an indemnity from the Receiver to the Purchaser);
  - (ii) an assignment and assumption of the Replaceable Contracts wherein the Kitsumkalum assumes the rights and obligations under the Replaceable Contracts and the Purchaser indemnifies the Receiver for all Liability under the Replaceable Contracts (but which for certainty will not include an indemnity from the Receiver to the Purchaser);
  - (iii) if required by the Province of British Columbia, an assignment and consent agreement with respect to the transfer of the Licences (not including any Rejected Licences);
  - (iv) a certificate dated as of the Phase 2 Closing Date of a senior officer of the Receiver having knowledge of the facts certifying, on behalf of the Receiver

- and without personal liability, that the Receiver's covenants and agreements to be observed or performed on or before the Phase 2 Closing Date pursuant to the terms of this Agreement have been duly observed and performed in all material respects; and
- (v) such other documents and assurances as may be reasonably required by the Purchaser to give full effect to the intent and meaning of this Agreement or contemplated by the Vesting Order as they relate to the Phase 2 Purchased Assets.

# 9.3 Purchaser's Closing Documents.

- (a) In addition to payment of the Initial Payment, on or before the Phase 1 Closing Date, the Purchaser will deliver, or cause the Purchaser's Solicitors to deliver, to the Receiver's Solicitors in trust to be held in escrow as provided in this Agreement, the following duly executed as applicable:
  - (i) an assignment and assumption of Approved Contracts and Permits and Licenses;
  - (ii) assignments and assumption of Permitted Encumbrances;
  - (iii) the GST Certificates;
  - (iv) a certificate dated as of the Phase 1 Closing Date of a senior officer of the Purchaser having knowledge of the facts certifying, on behalf of the Purchaser and without personal liability, that the representations and warranties set out in Section 7.1 are true and correct in all material respects as at the Phase 1 Closing Date and that the Purchaser's covenants and agreements to be observed or performed on or before the Phase 1 Closing Date pursuant to the terms of this Agreement have been duly observed and performed in all material respects; and
  - (v) such other documents and assurances as may be reasonably required by the Purchaser to give full effect to the intent and meaning of this Agreement as it relates to the Phase 1 Purchased Assets.
- (b) In addition to payment of the Balance, subject to Section 2.8(d), on or before the Phase 2 Closing Date, the Purchaser will deliver, or cause the Purchaser's Solicitors to deliver, to the Receiver's Solicitors in trust to be held in escrow as provided in this Agreement, the following duly executed as applicable:
  - (i) an assignment and assumption of the Licences (not including any Rejected Licences);
  - (ii) an assignment and assumption of the Replaceable Contracts;
  - (iii) if required by the Province of British Columbia, an assignment and consent agreement with respect to the transfer of the Licences (not including any Rejected Licences);

- (iv) a certificate dated as of the Phase 2 Closing Date of a senior officer of the Purchaser having knowledge of the facts certifying, on behalf of the Purchaser and without personal liability, that the representations and warranties set out in Section 7.1 are true and correct in all material respects as at the Phase 2 Closing Date and that the Purchaser's covenants and agreements to be observed or performed on or before the Phase 2 Closing Date pursuant to the terms of this Agreement have been duly observed and performed in all material respects; and
- (v) such other documents and assurances as may be reasonably required by the Purchaser to give full effect to the intent and meaning of this Agreement as it relates to the Phase 2 Purchased Assets.
- 9.4 Preparation and Form of Documents. The closing documents contemplated in Sections 9.2 and 9.3 (other than the Vesting Order) (collectively, the "Closing Documents") will be prepared by the Purchaser's Solicitors and delivered to the Receiver's Solicitors at least five (5) Business Days before the Phase 1 Closing Date and at least two (2) Business Days before the Phase 2 Closing Date, as applicable. The Closing Documents (including the Vesting Order) will be in a form and substance reasonably satisfactory to the parties and their respective solicitors.
- **Payment into Trust.** On or before the Phase 1 Closing Date, the Purchaser will pay to the Purchaser's Solicitors in trust, by way of certified cheque, bank draft, or wire transfer, funds in an amount equal to the Balance.

### 9.6 Closing Procedure.

- (a) All Closing Documents, funds, and other items delivered by the Parties for the Phase 1 Purchased Assets will be held in trust by the Receiver's Solicitors and the Purchaser's Solicitors until completion of closing on the Phase 1 Closing Date in accordance with this Agreement (except that the Vesting Order may be submitted for registration in accordance with this Section). Forthwith following the payment set forth in Section 9.5 and after receipt by the Purchaser's Solicitors of the documents and items referred to in Section 9.2(a), the Purchaser will cause the Purchaser's Solicitors to file the Vesting Order (together with such other documents as are required to be filed) in the Land Title Office. Upon the Purchaser's Solicitors obtaining a post application title search of the Lands which indicates that in normal Land Title Office routine, title to the Lands will be registered in the name of 392 and 390, as applicable, subject only to the Permitted Encumbrances and any documents filed by the Purchaser, and the state of title is in accordance with the terms and conditions of this Agreement, the Purchaser will cause the Purchaser's Solicitors to pay to the Receiver's Solicitors, in trust by wire transfer or certified cheque, the Balance and upon payment of the Balance, the Closing Documents as they relate to the Phase 1 Purchased Assets will be released to the appropriate parties.
- (b) All Closing Documents, funds, and other items delivered by the Parties for the Phase 2 Purchased Assets will be held in trust by the Receiver's Solicitors and the Purchaser's Solicitors until completion of closing on the Phase 2 Closing Date in accordance with this Agreement and thereafter released to the appropriate parties.

- Purchased Assets that all matters of payment, execution and delivery of documents by each Party to the other pursuant to the terms of this Agreement as it relates to the Phase 1 Purchased Assets will be deemed to be concurrent requirements and it is specifically agreed that nothing will be complete until everything required as a condition precedent at the Closing of the purchase and sale of the Phase 1 Purchased Assets has been paid, executed and delivered. It is a condition of Closing of the purchase and sale of the Phase 2 Purchased Assets that all matters of payment, execution and delivery of documents by each Party to the other pursuant to the terms of this Agreement as it relates to the Phase 2 Purchased Assets will be deemed to be concurrent requirements and it is specifically agreed that nothing will be complete until everything required as a condition precedent at the Closing of the purchase and sale of the Phase 2 Purchased Assets has been paid, executed and delivered.
- 9.8 Delivery of Business Records. The Receiver will table at Closing for the purchase and sale of the Phase 1 Purchased Assets and, on release of escrow, after completion of the purchase and sale of the Phase 1 Purchased Assets, will deliver originally executed copies of the Business Records, if in possession or control of the Receiver, to the Purchaser, to the extent not previously delivered.
- 9.9 Late Payment at Closing. Notwithstanding anything else contained herein, the Purchaser will make all commercially reasonable efforts to ensure that the Balance payable in cash under Section 2.4 (each a "Closing Payment") will be paid to and received by the Receiver's Solicitors, in trust on or before 5:00 p.m. (Vancouver time) on the Phase 1 Closing Date. If for any reason out of the control of the Purchaser (which, for greater certainty, will not include any event which is a default by the Purchaser under this Agreement), the Purchaser cannot ensure that such amount will be received by the Receiver's Solicitors, in trust on or before 5:00 p.m. (Vancouver time) on the Phase 1 Closing Date, then provided the Purchaser's Solicitors confirm in writing to the Receiver's Solicitors on the Phase 1 Closing Date that the Purchaser's Solicitors are holding the Balance in their trust account, they may make the payment to the Receiver's Solicitors, in trust on or before 5:00 p.m. (Vancouver time) on the next Business Day following the Phase 1 Closing Date, and the Purchaser will pay to the Receiver, in addition to the Balance, simple interest on such amount at a rate equal to three percent (3.0%) per annum for each day after the Phase 1 Closing Date until the Balance is received by the Receiver's Solicitors, in trust (and if it is received after 5:00 p.m. (Vancouver time) on any day, then an additional day's interest will be added). For example, if the Phase 1 Closing Date occurs on a Friday, the funds are wired on the following Monday and the Balance is received by the Receiver's Solicitors at 6:00 p.m. (Vancouver time) the day after wiring, the Purchaser will pay the Receiver four days' interest on the Balance.

# **9.10 Termination**. Notwithstanding any other provision of this Agreement:

(a) if the purchase and sale of the Phase 1 Purchased Assets contemplated by this Agreement do not complete on or prior to the day that is sixty (60) days after the Court pronounces the Vesting Order other than as a result of the default of the Purchaser, then the Purchaser may, in its sole discretion, terminate this Agreement with written notice delivered to the Receiver without any further liability, and the Receiver will forthwith return any part of the Deposit then remaining to the Purchaser; and

(b) this Agreement will automatically terminate upon Closing of both the purchase and sale of the Phase 1 Purchased Assets and the Phase 2 Purchased Assets, upon which all of the representations, warranties and covenants contained herein will merge and there will be no survival of any representation, warranties or covenants contained in this Agreement except for any provision of this Agreement expressly contemplating obligations to be observed or performed by a Party after Closing.

#### **ARTICLE 10- TAXES**

- 10.1 GST. The Purchaser represents and warrants to Receiver that each of the entities comprising the Purchaser is and will be, as of the closing on each of the Phase 1 Closing Date and the Phase 2 Closing Date, registered for GST in accordance with the requirements of Subdivision D of Division V of the Excise Tax Act (Canada) (the "ETA"). On the Phase 1 Closing Date, the Purchaser will pay to the Receiver GST calculated at the applicable GST rate on the Purchase Price. Notwithstanding the foregoing, on the Phase 1 Closing Date, each of 392 and 390 will deliver to the Receiver a certificate (each a "GST Certificate" and together, the "GST Certificates") of a senior officer of each of 392 and 390 certifying, on behalf of the Purchaser and without personal liability:
  - (a) that each of 392 and 390 is registered under Part IX of the ETA as of the Phase 1 Closing Date;
  - (b) its registration number; and
  - (c) that 392 will account for, report and remit any GST payable in respect of the purchase of the Lands and Buildings (other than the Stump Dump) in accordance with the ETA;
  - that 390 will account for, report and remit any GST payable in respect of the purchase of the Stump Dump and the Buildings thereon in accordance with the ETA.

If the Purchaser delivers such GST Certificates, then the Purchaser will not be required to pay to the Receiver, and the Receiver will not be required to collect from the Purchaser nor report or remit, any GST in connection with the portion of the Purchase Price allocated to the Lands and the Buildings. The Purchaser will indemnify and hold the Receiver and its directors, officers, employees, advisors and agents harmless from any liability under the ETA arising as a result of any breach of this Section 10.1, the GST Certificate or any declaration made therein and such indemnity will survive the completion of the transactions contemplated herein.

10.2 Provincial Sales Tax ("PST"). The Purchaser acknowledges that it is liable to pay provincial sales tax in respect of its purchase of some or all of the Inventory and Machinery and Equipment. The Receiver and the Purchaser will use commercially reasonable efforts to reach an agreement on the amount of PST payable. On the Phase 1 Closing Date, the Purchaser will pay to the Receiver PST calculated at the applicable PST rate on the portion of the Purchase Price allocated to any Inventory and Machinery and Equipment subject to the payment of PST. The Purchaser will indemnify and hold the Receiver and its directors, officers, employees, advisors and agents harmless from any liability related to the Receiver's failure to account for, or report and remit such PST and such indemnity will survive the completion of the transactions contemplated herein.

- 10.3 Tax Elections. Notwithstanding the above, the Receiver will cooperate with the Purchaser to execute any election available under applicable law that may reduce or defer the amount or due date of any GST, PST, or other tax payable by the Purchaser provided such election will not result in any increased cost or tax liability for the Receiver.
- 10.4 Other Taxes. The Purchaser will be responsible for any transfer taxes, fees and expenses in connection with the registration of the Vesting Order, and the Receiver will be responsible for any taxes or fees in respect of the effective disposition of the Phase 1 Purchased Assets and the Phase 2 Purchased Assets by the Transaction, including, without limitation, income tax.

#### **ARTICLE 11 - GENERAL**

- 11.1 Post-Closing Access to Business Records. After each of the Phase 1 Closing Date and the Phase 2 Closing Date, the Purchaser will provide access to any Business Records reasonably required by the Receiver in order to prepare and file any tax returns or to perform its obligations as Receiver of the Companies. This provision will survive the completion of the Transaction.
- **11.2 Further Assurances**. Each of the parties will execute and deliver all such further documents and do such further acts and things as may be reasonably required from time to time to give effect to this Agreement.
- 11.3 Entire Agreement. This Agreement constitutes the entire agreement between the Receiver and the Purchaser pertaining to the Transaction and supersedes all prior agreements and undertakings, negotiations and discussions, whether oral or written, of the Receiver and the Purchaser and there are no representations, warranties, covenants or agreements between the Receiver and the Purchaser except as set out in this Agreement.
- **11.4 Amendment**. Subject to Section 11.5, this Agreement may only be altered or amended by an agreement in writing executed by all of the Parties.
- Solicitors as Agents. Any notice, approval, waiver, agreement, instrument, document or communication permitted, required or contemplated in this Agreement may be given or delivered and accepted or received by the Purchaser's Solicitors, on behalf of the Purchaser, and by the Receiver's Solicitors, on behalf of the Receiver, and any tender of Closing Documents and the Purchase Price may be made upon the Receiver's Solicitors and the Purchaser's Solicitors, as the case may be.
- **11.6 Notices**. Any notice, document or communication required or permitted to be given under this Agreement will be in writing and delivered by hand or electronic transmission as follows:

### (a) if to the Purchaser:

Kitsumkalum First Nation as represented by Kitsumkalum Band Council PO Box 544 Terrace, BC V8G 4BS

Attention:

Steve Roberts, Kitsumkalum Band Manager

E-mail

sroberts@kitsumkalum.com

with a copy to the Purchaser's Solicitors:

ATAC Law Corporation 308-8988 Fraserton Court Burnaby, BC V5J 5H8

Attention:

Mike Stewart and Gurpal Sandhu

E-mail:

mstewart@ataclaw.ca, gsandhu@ataclaw.ca

### (b) if to the Receiver:

Alvarez & Marsal Canada Inc. 902 – 925 West Georgia Street Vancouver, BC V6C 3L2

Attention:

Anthony Tillman

E-mail:

atillman@alvarezandmarsal.com

with a further copy to the Receiver's Solicitors:

Fasken Martineau DuMoulin LLP 2900 – 550 Burrard Street Vancouver, BC V6C 0A3

Attention:

Kibben Jackson and Sarah Batut

E-mail:

kjackson@fasken.com, sbatut@fasken.com

or to such other address in Canada as either Party may in writing advise. Any notice, document or communication will be deemed to have been given on the Business Day when delivered by hand if delivered prior to  $5:00~\mathrm{p.m.}$  (Vancouver time), otherwise will be deemed to be delivered and received on the next Business Day; or, if made by email, will be deemed to have been given on the Business Day when transmitted if it is so transmitted prior to  $5:00~\mathrm{p.m.}$  (Vancouver time) on the day of transmittal, otherwise will be deemed to be given and received on the next Business Day.

11.7 Fees. Subject to Section 2.3, each of the Parties will pay its own legal fees and fees of its consultants. The Purchaser will pay all registration costs and property transfer tax payable in connection with the Transaction and the taxes identified ARTICLE 10 as being the Purchaser's responsibility, and the Receiver will be responsible for the taxes or fees identified in ARTICLE 10 as being the Receiver's responsibility. For certainty, the Purchaser will have no liability to fund any of the costs of the receivership.

- **11.8 Accounting Terms.** Accounting terms used herein and not expressly defined will be deemed to have such meanings as may apply on the application of the Accounting Standards.
- 11.9 Time. Time is of the essence of this Agreement.
- **11.10 Tender**. Unless otherwise set out herein, any tender of documents or money may be made upon the party being tendered or upon its solicitors and money will be tendered by certified cheque, bank draft, or wire transfer.
- **11.11 Enurement.** This Agreement will enure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns.
- 11.12 Assignment. The Purchaser may not assign its rights and obligations under this Agreement without the prior written consent of the Receiver. If the Receiver consents to any proposed assignment, the assignee must enter into an agreement pursuant to which the assignee agrees to be bound by all of the obligations and Liability of the Purchaser under this Agreement as if it was the original party and the Purchaser will not be released from its obligations and Liability under this Agreement until the completion of the Transaction, at which time the assignor will be automatically released from all of its obligations and Liability under this Agreement without the need for any further deliveries or instruments of release.
- **11.13 Governing Law.** This Agreement will be governed by and construed in accordance with the laws of the Province of British Columbia and the laws of Canada applicable in it.
- **11.14 Waiver.** No waiver of any of the provisions of this Agreement will be deemed or will constitute a waiver of any other provision nor will any waiver constitute a continuing waiver unless otherwise expressed or provided. Subject to Section 11.5, no waiver of any provision of this Agreement is binding unless it is executed in writing by each Party to be bound.
- 11.15 Currency. All dollar amounts referred to are Canadian dollars.
- **11.16 Construction**. The division and headings of this Agreement are for reference only and are not to affect construction or interpretation.
- **11.17 Joint and Several**. If any Party is comprised of more than one party, then all of the covenants, agreements, representations and warranties of each of the parties comprising such Party, will be joint and several covenants, agreements, representations and warranties.
- 11.18 Replaceable Contracts. Notwithstanding that the Purchaser is assuming the Replaceable Contracts, the Receiver acknowledges having been advised by Kitsumkalum that pursuant to separate agreements negotiated between the Purchaser and the Replaceable Contracts Holders, the Replaceable Contracts Holders have agreed directly with the Purchaser that the Purchaser will have no liability for any Liabilities under the Replaceable Contracts that arose prior to the Phase 2 Closing Date.
- **11.19 Counterparts and Execution**. This Agreement may be executed electronically and in counterparts and delivered by electronic transmission including by PDF format, and each such counterpart will constitute an original and all such counterparts together will constitute one and the same agreement.
- 11.20 Schedules. The following schedules are attached to and form a part of this Agreement:

Schedule A – Legal Description and Permitted Encumbrances Schedule B - Form of Vesting Order Schedule C – Inventory Schedule D – Machinery and Equipment

[Signature page follows]

The Parties are signing this Agreement as of the date set out above.

ALVAREZ & MARSAL CANADA INC.., in its capacity as Receiver of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd.

| By:         |                             |
|-------------|-----------------------------|
| -           | Name: Anthony Tillman       |
|             |                             |
|             | Title: Authorized Signatory |
|             |                             |
|             |                             |
|             |                             |
|             |                             |
| KITS        | UMKALAM FIRST NATION        |
|             |                             |
|             |                             |
|             |                             |
| By:         |                             |
| <i> j</i> . | Name: Steven W. Roberts     |
|             |                             |
|             | Title: Band Manager         |
|             | _                           |
|             |                             |
|             |                             |
|             |                             |
| 1255        | 392 B.C. LTD.               |
| 1333        | 372 D.G. LLD.               |
|             |                             |
|             |                             |
| D.,,        |                             |
| By:         |                             |
|             | Name:                       |
|             | Title:                      |
|             | Title.                      |
|             |                             |
|             |                             |
| 1255        | 390 B.C. LTD.               |
| 1333        | 370 B.C. LI D.              |
|             |                             |
|             |                             |
| D           |                             |
| Ву:         |                             |
|             | Name:                       |
|             | Title:                      |
|             | itte.                       |
|             |                             |
|             |                             |
| 1255        | 387 B.C. LTD.               |
| 1993        | 30/ B.C. LIV.               |
|             |                             |
|             |                             |
| ъ.          |                             |
| Ву:         |                             |
|             | Name:                       |
|             | Title:                      |
|             | Tiuc.                       |

## SCHEDULE A LEGAL DESCRIPTION AND PERMITTED ENCUMBRANCES

#### Part 1 - Legal Description of Lands

#### The following:

- 5330 Highway 16 W, Terrace, BC
   PID: 011-691-042 Lot A District Lot 616 Range 5 Coast District Plan 3986, Except Plan
   PRP47978 and PID: 011-691-051 Lot B District Lot 616 Range 5 Coast District Plan 3986
- 5402 Highway 16 W, Terrace, BC PID: 030-631-700 Lot A District Lots 616 and 1745 Range 5 Coast District Plan EPP78423
- 76 Kalum Lake Road, Terrace, BC
   PID: 011-768-398 Lot 3 District Lot 616 Range 5 Coast District Plan 3700
- 863 Kalum Lake Road, Terrace, BC
   PID: 009-426-833 District Lot 1398 Range 5 Coast District Except Plan 11735

#### Part 2 - Permitted Encumbrances

• PID: 011-691-042

Legal Notations:

 This title may be affected by a permit under Part 14 of the Local Government Act, see CA8395265

Charges, Liens and Interests:

- o 2024 Property Taxes
- PID: 011-691-051

Legal Notations:

 Hereto is annexed Easement CA7183199 over Lot A Plan EPP78423 as TP Art shown on Plan EPP82278

Charges, Liens and Interests:

- o 2024 Property Taxes
- PID: 030-631-700

**Legal Notations:** 

- o Hereto is annexed Easement BB1131385 over Lot 1 Plan BCP43227
- This title may be affected by a permit under Part 14 of the Local Government Act, see CA6825838

Charges, Liens and Interests:

- o Statutory Right of Way BB1131384
- o Easement CA7183199
- o Statutory Right of Way CA7224665
- o 2024 Property Taxes
- PID: 011-768-398

Legal Notations:

o NIL

Charges, Liens and Interests:

o 2024 Property Taxes

- PID: 009-426-833

Legal Notations:

o NIL

Charges, Liens and Interests:

o 2024 Property Taxes

## SCHEDULE B FORM OF VESTING ORDER

See attached.

#### IN THE SUPREME COURT OF BRITISH COLUMBIA

BETWEEN:

1392752 B.C. LTD.

**PETITIONER** 

AND:

SKEENA SAWMILLS LTD. SKEENA BIOENERGY LTD. and ROC HOLDINGS LTD.

**RESPONDENTS** 

# ORDER MADE AFTER APPLICATION APPROVAL AND VESTING ORDER

| BEFORE | ) | THE HONOURABLE MR. JUSTICE WALKER | ) | April 16, 2024 |
|--------|---|-----------------------------------|---|----------------|
|        | ) | WALKER                            | ) |                |
|        | ) |                                   | ) |                |

ON THE APPLICATION OF Alvarez & Marsal Canada Inc. in its capacity as court-appointed receiver (in such capacity, the "Receiver") of Skeena Sawmills Ltd. ("Sawmills"), Skeena Bioenergy Ltd. ("Bioenergy") and ROC Holdings Ltd. ("Roc", and together with Sawmills and Bioenergy, the "Companies"), coming on for hearing at Vancouver, British Columbia on April 2-5, 10 and 16, 2024, AND ON HEARING Kibben Jackson and Mishaal Gill, counsel to the Receiver, Mike Stewart and Gurpal Sandhu, counsel to Kitsumkalum First Nation, and those other counsel listed in Schedule "A" hereto, AND NO ONE ELSE APPEARING, although duly served.; AND UPON READING the material filed, including the First Confidential Report of the Receiver dated January 19, 2024, the Receiver's Fourth Report dated February 29, 2024, the Supplemental Report to the Fourth Report of the Receiver dated March 6, 2024, the Second Supplemental Report to the Fourth Report of the Receiver dated March 27, 2024, the Fourth Supplemental Report to the Fourth Report of the Receiver dated March 27, 2024, the Fourth Supplemental Report to the Fourth Report of the Receiver dated March 27, 2024, the Fourth Supplemental Report to the Fourth Report of the Receiver dated March 27, 2024, the Fourth Supplemental Report to the Fourth Report of the Receiver dated April 3, 2024, the Confidential Supplemental Report to

the Fourth Report of the Receiver dated April 3, 2024 and the Fifth Supplemental Report to the Fourth Report of the Receiver dated April 16, 2024.

#### THIS COURT ORDERS AND DECLARES that:

#### THE TRANSACTION

- 1. The sale transaction (the "Transaction") contemplated by the Purchase and Sale Agreement dated April 15, 2024 (the "Sale Agreement") between the Receiver, as vendor, and 1355387 B.C. Ltd., 1355390 B.C. Ltd., 1355392 B.C. Ltd. and Kitsumkalum First Nation (collectively, the "Purchaser"), as purchasers, a copy of which is attached hereto as <u>Schedule "B"</u>, is hereby approved, and the Sale Agreement is commercially reasonable. The execution of the Sale Agreement by the Receiver is hereby authorized and approved, and the Receiver is hereby authorized and directed to take such additional steps and execute such additional documents as may be necessary or desirable for the completion of the Transaction and for the conveyance to the Purchaser of the Purchased Assets (as that term is defined in the Sale Agreement).
- 2. Upon delivery by the Receiver to the Purchaser of a certificate substantially in the form attached as Schedule "C" hereto (the "Receiver's Phase 1 Certificate"), all of the Companies' right, title and interest in and to the Phase 1 Purchased Assets (as that term is defined in the Sale Agreement) shall vest absolutely in the Purchaser in fee simple, free and clear of and from any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, trusts or deemed trusts (whether contractual, statutory, or otherwise), liens, executions, levies, charges, or other financial or monetary claims, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured or otherwise (collectively, the "Claims") including, without limiting the generality of the foregoing: (i) any encumbrances or charges created by the Order of this Court dated September 20, 2023 (the "Receivership Order"); (ii) all charges, security interests or claims evidenced by registrations pursuant to the Personal Property Security Act of British Columbia or any other personal property registry system; and (iii) those Claims listed on Schedule "D" hereto (all of which are collectively referred to as the "Encumbrances", which term shall not include the permitted encumbrances, easements and restrictive covenants listed on Schedule "E" hereto), and, for greater certainty, this Court orders that all of the Encumbrances affecting or relating to the Purchased Assets are hereby expunged and discharged as against the Purchased Assets.

- 3. Upon presentation for registration in the Land Title Office for the Land Title District of Prince Rupert of a certified copy of this Order, together with a letter from counsel for the Receiver authorizing registration of this Order, the British Columbia Registrar of Land Titles is hereby directed to:
  - enter the Purchaser as the owner of the lands (the "Lands"), as identified in <u>Schedule "F"</u> hereto, together with all buildings and other structures, facilities and improvements located thereon and fixtures, systems, interests, licenses, rights, covenants, restrictive covenants, commons, ways, profits, privileges, rights, easements and appurtenances to the said hereditaments belonging, or with the same or any part thereof, held or enjoyed or appurtenant thereto, in fee simple in respect of the Lands, and this Court declares that it has been proved to the satisfaction of the Court on investigation that the title of the Purchaser in and to the Lands is a good, safe holding and marketable title and directs the BC Registrar to register indefeasible title in favour of the Purchaser as aforesaid; and
  - (b) having considered the interest of third parties, to discharge, release, delete and expunge from title to the Lands all of the registered Encumbrances except for those listed in <a href="Schedule">Schedule "E"</a> hereto.
- 4. Upon delivery by the Receiver to the Purchaser of a certificate substantially in the form attached as Schedule "G" hereto (the "Receiver's Phase 2 Certificate"), all of the Companies' right, title and interest in and to the Phase 2 Purchased Assets (as that term is defined in the Sale Agreement) which can and are to be transferred to the Purchaser under the terms of the Sale Agreement (if any) shall vest absolutely in the Purchaser in fee simple, free and clear of and from any and all Encumbrances, and, for greater certainty, this Court orders that all of the Encumbrances affecting or relating to such Phase 2 Purchased Assets are hereby expunged and discharged as against such Phase 2 Purchased Assets.
- 5. For the purposes of determining the nature and priority of Claims, the net proceeds from the sale of the Purchased Assets shall stand in the place and stead of the Purchased Assets, and from and after the delivery of the Receiver's Certificate all Claims shall attach to the net proceeds from the sale of the Purchased Assets with the same priority as they had with respect to the Purchased Assets immediately prior to the sale, as if the Purchased Assets had not been sold and remained in the possession or control of the person having had possession or control immediately prior to the sale.

- 6. The Receiver is to file with the Court a copy of the Receiver's Phase 1 Certificate and, if applicable, the Receiver's Phase 2 Certificate forthwith after delivery thereof.
- 7. Pursuant to Section 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act or Section 18(1)(o) of the Personal Information Protection Act of British Columbia, the Receiver is hereby authorized and permitted to disclose and transfer to the Purchaser all human resources and payroll information in the company's records pertaining to the Companies' past and current employees. The Purchaser shall maintain and protect the privacy of such information and shall be entitled to use the personal information provided to it in a manner which is in all material respects identical to the prior use of such information by the Companies.
- 8. Subject to the terms of the Sale Agreement, vacant possession of the Purchased Assets, including the Lands, shall be delivered by the Receiver to the Purchaser at 12:00 noon on the Phase 1 Closing Date and, if applicable, the Phase 2 Closing Date (as those terms are defined in the Sale Agreement), subject to the permitted encumbrances enumerated in Schedule "E".
- 9. The Receiver, with the consent of the Purchaser, shall be at liberty to extend each of the Phase 1 Closing Date and, if applicable the Phase 2 Closing Date to such later dates as those parties may agree without the necessity of a further Order of this Court.

#### 10. Notwithstanding:

- (a) these proceedings;
- (b) any applications for a bankruptcy order in respect of the Companies now or hereafter made pursuant to the *Bankruptcy and Insolvency Act* (the "BIA") and any bankruptcy order issued pursuant to any such applications; and
- (c) any assignment in bankruptcy made by or in respect of the Companies,

the vesting of the Purchased Assets in the Purchaser pursuant to this Order shall be binding on any trustee in bankruptcy that may be appointed in respect of the Companies and shall not be void or voidable by creditors of the Companies, nor shall it constitute or be deemed to be a transfer at undervalue, fraudulent preference, assignment, fraudulent conveyance or other reviewable transaction under the BIA or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

- 11. In addition to and without limiting the rights and protections afforded to the Receiver pursuant to the Receivership Order made herein, the Receiver and its employees and representatives shall not incur any liability as a result of acting in accordance with this Order save and except for any gross negligence or wilful misconduct on the part of any such parties. All protections afforded to the Receiver pursuant to the Receivership Order, any further order granted in these proceedings or the BIA shall continue to apply.
- 12. Upon delivery of the Receiver's Phase 1 Certificate, all persons shall be absolutely and forever barred, estopped, foreclosed and permanently enjoined from pursuing, asserting, exercising, enforcing, issuing or continuing any steps or proceedings, or relying on any rights, remedies, claims or benefits in respect of or against the Receiver, its directors, officers, employees, counsel, advisors and representatives, the Purchaser, the Companies or the Purchased Assets, in any way relating to, arising from or in respect of the insolvency of the Companies, the commencement or existence of these receivership proceedings or the completion of the Transaction.
- 13. The Receiver, the Purchaser, or any other party have liberty to apply for such further or other directions as may be necessary or desirable to give effect to this Order.
- 14. Endorsement of this Order by counsel other than counsel for the Receiver is hereby dispensed with.
- 15. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body, wherever located, to give effect to this Order and to assist the Receiver and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Receiver, as an officer of this Court, as may be necessary or desirable to give effect to this Order or to assist the Receiver and its agents in carrying out the terms of this Order.

THE FOLLOWING PARTIES APPROVE THE FORM OF THIS ORDER AND CONSENT TO EACH OF THE ORDERS, IF ANY, THAT ARE INDICATED ABOVE AS BEING BY CONSENT:

Signature of Kibben Jackson
Lawyer for ALVAREZ & MARSAL CANADA
INC., in its capacity as Receiver of Skeena
Sawmills Ltd., Skeena Bioenergy Ltd. and ROC
Holdings Ltd.

| BY THE COURT |  |
|--------------|--|
|              |  |
|              |  |
| REGISTRAR    |  |

## Schedule "A"

(List of Counsel)

## Schedule "B"

(Sale Agreement)

#### Schedule "C"

(Receiver's Phase 1 Certificate)

No. S236214 Vancouver Registry

#### IN THE SUPREME COURT OF BRITISH COLUMBIA

BETWEEN:

1392752 B.C. LTD.

**PETITIONER** 

AND:

SKEENA SAWMILLS LTD., SKEENA BIOENERGY LTD. and ROC HOLDINGS LTD.

RESPONDENTS

#### RECEIVER'S CERTIFICATE

- A. By order made September 20, 2023, this Court appointed Alvarez & Marsal Canada Inc. (the "Receiver") as receiver and manager of the assets, undertakings and property of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd. (collectively, the "Companies").
- B. Pursuant to an order of the Court dated April 16, 2024, the court approved the Sale Agreement entered into by the Receiver and Kitsumkalum First Nation ("Kitsumkalum"), and provided for the vesting of the Phase 1 Purchased Assets (as defined in the Sale Agreement) effective upon delivery by the Receiver to the Purchaser of a certificate confirming: (i) receipt of the full amount of the Purchase Price (as defined in the Sale Agreement); and (ii) the transaction in respect of the Phase 1 Purchased Assets as contemplated by the Sale Agreement has been completed to the satisfaction of the Receiver.

#### THE RECEIVER HEREBY CERTIFIES the following:

1. The Receiver has received the full amount of the Purchase Price.

| at on                                       | , 2024.   |
|---|---|
| ity as Receiver of S<br>na Bioenergy Ltd. a | CANADA INC., in its<br>Skeena Sawmills Ltd.,<br>nd ROC Holdings Ltd., and<br>ty |
| ame:  |   |
| c   | AREZ & MARSAL   |

## Schedule "D"

(Encumbrances to be discharged from title to Purchased Assets)

| SECURED PARTY     | LAND TITLE REGISTRATION NO. |  |  |
|-------------------|-----------------------------|--|--|
| 1392752 B.C. LTD. | CB458433                    |  |  |
| 1392752 B.C. LTD. | CB458434                    |  |  |

| SECURED PARTY                             | PERSONAL PROPERTY REGISTRY<br>REGISTRATION NO. |
|---|--|
| Delta Cedar Specialties Ltd.              | 282094N  |
| Timber Baron Contracting Ltd.             | 903702N, 903703N                               |
| Deuce Creek Contracting Ltd.              | 350830P, 350833P                               |
| Infinity West Enterprises Inc.            | 427844P, 427848P                               |
| Antler Creek Contracting Ltd.             | 507680P, 507684P, 757548P, 757558P             |
| L & J Logging Ltd.                        | 771796P, 771821P                               |
| D.R. Holtom Ltd.                          | 777413P, 777418P                               |
| Daudet Creek Contracting Ltd.             | 788822P, 788843P, 788873P, 788895P,            |
|   | 788922P, 788941P, 788955P, 788978P             |
| Silvicon Services Inc.                    | 795469P, 797967P                               |
| K'Alii AKS Timber Corporation             | 795470P, 797968P                               |
| Round Lake Transport Ltd.                 | 830654P, 830659P                               |
| Dynamic Capital Equipment Finance Inc.    | 144761N, 154648N, 184908N, 184963N             |
| Caterpillar Financial Services Limited    | 007610M, 984133L, 632901M, 941146M             |
| CWB National Leasing Inc.                 | 170285N, 545347N                               |
| Corley Manufacturing Company              | 820477P  |
| Microtec Inc.                             | 120833Q  |
| The Bank of Nova Scotia                   | 436815P, 608762K                               |
| Her Majesty the Queen in the Right of the | 313007P  |
| Province of British Columbia              |  |
| Timber Tracks Inc.                        | 541647P  |
| Bank of Montreal/Banque de Montreal       | 660141P  |
| Stardust Contracting Ltd.                 | 323463P  |
| 1392752 B.C. LTD.                         | 294187P  |

#### Schedule "E"

#### (Permitted Encumbrances)

• PID: 011-691-042

Legal Notations:

o This title may be affected by a permit under Part 14 of the Local Government Act, see CA8395265

Charges, Liens and Interests:

- o Property taxes imposed in 2024
- PID: 011-691-051

Legal Notations:

 Hereto is annexed Easement CA7183199 over Lot A Plan EPP78423 as TP Art shown on Plan EPP82278

Charges, Liens and Interests:

- o Property taxes imposed in 2024
- PID: 030-631-700

Legal Notations:

- o Hereto is annexed Easement BB1131385 over Lot 1 Plan BCP43227
- o This title may be affected by a permit under Part 14 of the Local Government Act, see CA6825838

Charges, Liens and Interests:

- o Statutory Right of Way BB1131384
- o Easement CA7183199
- o Statutory Right of Way CA7224665
- o Property taxes imposed in 2024
- PID: 011-768-398

Legal Notations:

o NIL

Charges, Liens and Interests:

- o Property taxes imposed in 2024
- PID: 009-426-833

Legal Notations:

o NIL

Charges, Liens and Interests:

o Property taxes imposed in 2024

#### Schedule "F"

(Real Property)

- PID: 011-691-042; LOT A DISTRICT LOT 616 RANGE 5 COAST DISTRICT PLAN 3986, EXCEPT PLAN PRP47978
- PID: 011-691-051; LOT B DISTRICT LOT 616 RANGE 5 COAST DISTRICT PLAN 3986
- PID: 030-631-700; LOT A DISTRICT LOTS 616 AND 1745 RANGE 5 COAST DISTRICT PLAN EPP78423
- PID: 011-768-398; LOT 3 DISTRICT LOT 616 RANGE 5 COAST DISTRICT PLAN 3700
- PID: 009-426-833; DISTRICT LOT 1398 RANGE 5 COAST DISTRICT EXCEPT PLAN 11735

#### Schedule "G"

(Receiver's Phase 2 Certificate)

No. S236214 Vancouver Registry

#### IN THE SUPREME COURT OF BRITISH COLUMBIA

BETWEEN:

1392752 B.C. LTD.

**PETITIONER** 

AND:

SKEENA SAWMILLS LTD., SKEENA BIOENERGY LTD. and ROC HOLDINGS LTD.

RESPONDENTS

#### **RECEIVER'S CERTIFICATE**

- A. By order made September 20, 2023, this Court appointed Alvarez & Marsal Canada Inc. (the "Receiver") as receiver and manager of the assets, undertakings and property of Skeena Sawmills Ltd., Skeena Bioenergy Ltd. and ROC Holdings Ltd. (collectively, the "Companies").
- B. Pursuant to an order of the Court dated April 16, 2024, the court approved the Sale Agreement entered into by the Receiver and Kitsumkalum First Nation ("Kitsumkalum"), and provided for the vesting of such of the Phase 2 Purchased Assets (as defined in the Sale Agreement) which can and are to be transferred to the Purchaser under the terms of the Sale Agreement (if any) effective upon delivery by the Receiver to the Purchaser of a certificate confirming that the transaction in respect of the Phase 2 Purchase Assets contemplated by the Sale Agreement has been completed to the satisfaction of the Receiver.

### THE RECEIVER HEREBY CERTIFIES the following:

1. The transaction in respect of the Phase 2 Purchase Assets contemplated by the Sale Agreement has been completed to the satisfaction of the Receiver.

| This Certificate was delivered by the Receiver at | on, 2024.  |
|---|--|
| capacit <sub>i</sub><br>Skeena                    | REZ & MARSAL CANADA INC., in its y as Receiver of Skeena Sawmills Ltd., Bioenergy Ltd. and ROC Holdings Ltd., and ts personal capacity |
| Per:<br>Nar                                       | <br>ne:  |

## SCHEDULE C INVENTORY

| Location     | Category | Specific Location | Item Specs                           | Volume (M3)  |
|--------------|----------|-------------------|--------------------------------------|--------------|
| Mill         | Log      |                   | Sawlogs Hembal at Mill               | 1,986.7      |
| Mill         | Log      |                   | Spruce at Mill                       | 397.8        |
| Mill         | Log      |                   | Cedar at Mill                        | 176.3        |
| Mill         | Log      |                   | Pulp Logs (Hembal) at Mill           | 1,095.7      |
| Bush         | Log      | CP 523 blk NIS001 | decked HemBal (pulp and S/L (60/40)  | 300.0        |
| Bush         | Log      | CP 523 blk NIS002 | Decked HemBal                        | 2,419.0      |
| Bush         | Log      | CP 523 blk NIS002 | Roadside unprocessed HemBal          | 1,910.0      |
|              |          | CP 517 blks.      |                                      |              |
| Bush         | Log      | Paw001/003        | Decked HemBal pulpwood               | 3,435.0      |
| Bush         | Log      | CP 40 WAT001      | R/W (HBCw) unprocessed               | 357.0        |
| Location     | Category | Status            | Item Specs                           | Volume (ODT) |
| Mill         | Fibre    |                   | Chips at Mill                        | 11.0         |
| Pellet Plant | Fibre    |                   | Chips at Skeena Bioenergy            | 107.3        |
| Pellet Plant | Fibre    |                   | Pellet Fine material at Bioenergy    | 12.6         |
| Pellet Plant | Fibre    |                   | Hog material at Bioenergy            | 1,884.1      |
|              |          |                   | Pellet fibre material at Bioenergy   |              |
| Pellet Plant | Fibre    |                   | (material is a mixed belnd of fibre) | 984.8        |
| Pellet Plant | Fibre    |                   | Shavings material at Bioenergy       | 162.3        |
| Pellet Plant | Fibre    |                   | Belend sawdust and hog material at B | 875.0        |
|              |          |                   |                                      |              |

# SCHEDULE D MACHINERY AND EQUIPMENT

See attached.

| 1985   1986      |       | No. P  |
|--|-------|--|
| Company  | INT.  | Hen/Kena   |
| ## 15   10   10   10   10   10   10   10   |       | 1  |
| 1  |       |  |
| Section   Company   Comp   |       |  |
| Septiments and provided in the |       | <del> </del>                                     |
| Part   Company   Part   |       |  |
| Services of the control of the contr |       | <del></del>                                      |
| Company  |       | $oxed{oxed}$                                     |
| Column   C   |       | $\vdash$   |
| 1  |       | -  |
| 1  |       |  |
| Section of the control of the contro |       | Τ=   |
| Proceedings   Proceedings   Process   Proces   |       |  |
| Part   March   Part   Part   Part   March   Part   Pa   |       |  |
| 1.5    |       |  |
| Medical Control Contro |       | 28.254 h   |
| March   Marc   |       | 1  |
| And Indiented  |       |  |
| ## 1 Fig Special complete and 1, Clip on the State Management and 1, Clip of the State Management and 1, Clip on the State Management and 1, Clip of the State Management and 1, Clip on the State Management and  |       |  |
| 1. September 1. Se |       |  |
| Services of the services of th |       |  |
| Services of the control of the contr |       | _  |
| Section of the comment of the commen |       |  |
| One Mark Life Content of the back Adea the conditioners.  One Mark Life Content of the back Adea the conditioners.  One Mark Life Content of the Content of  | a     |  |
| COUNTY Toward and Section County of the County of County |       | =  |
| 3.1 1900 Bellet    1902  |       | Ε  |
| Merche dept. 19 17 17 17 and record level on unit 75 12 and an overal PS 12 and an ove |       |  |
| Window provided a boat filed framework (1) the company that deep (1) and the company of the comp |       | +  |
| ## 1.00 of 37 household with the control of the con |       | T  |
| Color   Colo   |       | <del> </del>                                     |
| Section 1. Corint September 1. Sept. 1. Corint September 1. Sept. 1. Corint September 1. Sept. 1. Sept | KI.50 |  |
| Content   Cont   |       |  |
| Security   Company   Com   |       |  |
| Montane   Mont   |       | <u> </u>   |
| Section Development and services, Appears  Option Development and Section Company (1997)  27 5 1 pages (Lember Leading Active Ac |       | <u> </u>   |
| AcCorden Management of the Control o |       |  |
| service to comment of the comment of the comment of the comment independent, comment independent of the comment of |       |  |
| Silvenum Hei delnis careful. His Steller Hei Steller Hei Steller committe (septen)  197 A. E. process grant and earth both has been process, process and septen described (septen)  29. A. E. Process and septen described (septen)  29.  |       |  |
| Sectional bridge on the same with droy and present ingree develor pation   |       |  |
| 97   Set 10   The better with drive   Set 10   The first of the control of the co |       |  |
| 25 Setting 105, Se |       | -  |
| Section   Sect   |       |  |
| Subtract  Subtra |       |  |
| path before conducted with drive  37 X X X X but conveyer 7 3 has followed orders, 10 ams presumate pickes forget diseased station, 17 (5) and 52° release with 5 hip drive  48.6.5 Section 20 perch 10  48.6.5 Section 20 perch 1 |       |  |
| Proc.   Decide Hammer  |       | <del></del>                                      |
| After Broder, 11E, Bendeuw destricts, 30 for temporary and legislating period.  After 100000  After 100000  After 100000  After 1000000000000000000000000000000000000  |       | <del> </del>                                     |
| West part of some part of the property   |       |  |
| G. 4 shared log but broader docks, abolite, Reduced other G. 14 shared log but broader docks, abolite, Reduced other G. 15 shared log but broader docks, abolite, Reduced other G. 16 shared log but broader docks, abolite, Reduced other G. 17 shared log but broader docks, abolite, Reduced other G. 18 shared log but broader docks, abolite, Reduced abolite, abolite docks, abolite, a |       | -  |
| 65 Cut of service (66 puts librarilla contents). But produce converse, 26 puts per librarilla contents. But produce contents and per librarilla contents. But produce contents and per librarilla contents. But produce contents and per librarilla contents. But per librarilla bender and 5 to Reduce drive.  65 Cut of the per librarilla contents and per librarilla contents. But per librarilla bender deck. 4 per librarilla bender deck. 5 per li |       | <b>†</b>   |
| Price   Pric   |       |  |
| 18 P. styrond boy have foreign decks, if a min to get participation, a first work in England on the control of the control o   |       | 1  |
| 18 P. styrond boy have foreign decks, if a min to get participation, a first work in England on the control of the control o   |       | <b>_</b>   |
| 18   Service   |       | +  |
| Wide      |       | <del>                                     </del> |
| Bit   Cap   Bit   Cap   Bit   Cap   Bit   Cap   Bit    |       |  |
| Bit   Cap   Bit   Cap   Bit   Cap   Bit   Cap   Bit    |       |  |
| 18   4 street log Duried deck, 4 am los planels deck, 2 am los planels deck, 3 am los planels deck, 4 am los planels deck, 4 am los planels deck, 4 am los planels deck, 3 am los planels, 3 am lo   |       |  |
| 100    |       |  |
| Fig. 1 stands log had larged edeck, sindle pin high relation drive. Relation drive.  |       | 1  |
| Concest  |       | -  |
| indic citysipin hands 150 hp. Twin 1.3 (17 feet onls, top 21 107 150 hp chipping hand, bottom splan removal hand, twin 167 ode ext preys note, that is repet to the control of the control |       | T  |
| 101  |       | 1  |
| 100  |       | <del> </del>                                     |
| 105  Yushera   Melicu U1000   Vid (obtension)   West   Section 23 texts Allen Bredery (e., that collection within from layer)  |       |  |
| (05) Code Hamme Merc, Tection 2) orbit, Alen Brodgreys, and collection exists from based (07) State of the Code of |       | 1  |
| 109 [ Forano 1290ES ] Twin band resaw system, 5 single cut, double column, mechanical strain, hydrautic leadworks, variable speed to 327 FPM, 32 6" X 9" [2R-7699]   |       |  |
| If the groupe blacket both 32 177 betten mile beite burben die einfannteren beaut mile fall 187 bed unter Turke 2 and 177 mile familie   |       | <del></del>                                      |
| A is gaiged access, servi, su the drope receiption in control and proper facilities (set, it is to see facilities access, management of patient, semantic positions, demonstrate back public back control back provide access and driven, Copy page 18 backs.  |       |  |
| 110 18 (Strate of the strate o |       | 旨  |
| 112   Six to hotsess cover coxis (secretally, 55 to hotsesse post (pet) from bord (sec)  |       | +  |
| 114 OT X 20" but conveys, diverging gala, Radicing data, Radicing data   |       |  |
| 15   45-55   1907 1-18 1/07 that converts, Rediction from   Rediction from   116   34-9   17 4 strong incoder deck, 4-sthrs, 3 tom bumber deck 6-sthrs, 3    |       |  |
| 117 47 105 mg 27 mbees Rection des, bit 3 mm presentals tights  119 119 119 119 119 119 119 119 119 11   |       | <del> </del>                                     |
| 199   175 Stated Dangles deck 1 with Mobile stackages, Ruddon drive   199   175 Stated Dangles deck 1 with Mobile stackages, Ruddon drive   190   19   |       | 1  |

|   | Y - EKEENA SA  | YMA.       | ŞLII            | )  | U-2-2  | Destalia   | and the t                                       | Libraria. |
|---|----------------|------------|-----------------|--|--|--|---|-----------|
| 121   | Unit #         | O.V.       | Year<br>1973    | Haice  |  | Description<br>Transformer, 1,500 kva, 3 ptuse, 60 hz, 25,000 -480V  | Sariat Number                                   | Hes/Kens  |
| 122   |                | F          | F               |  |  | 5 strand 5 1/2' camelback landing lable, Radicon drive, 4' (4) livin strand flyted unscrambler with drive<br>5 strand camelback cross transfer deck 3 hp Radicon, 5 arm roller stophower station, 18' (7) roll 33' 1/2' rollcase, pop up positioners   |   |           |
|   |                | <b>!</b> — | <u> </u>        | OH .   |  |  | 74  | *****     |
| 124   |                | 上          | <u>_</u>        | UM   | 1  | Pony edger, 30°, 3 saw, spined arbour, bvin lop hold down rolls, 5 bed rolls, lop opening, Elworthy set works, 150 hp 1790 rpm drive,<br>posh button control, (215 hp hydraulic power packs  | 1145  |           |
| 125<br>_126   | 45-57<br>45-59 | $\pm$      | $\vdash$        |  |  | 16° X 25° belt convence Replican drive, single presentatic pictor finger directing station. 16° X 23° 17° belt convence shall mount drive.   |   |           |
| 127   | 45-50<br>45-51 | _          |                 |  |  | 5 5 strand carriet back banding deck, 5 hp Radicon offys   |   |           |
| 128<br>129  | 10-01          |            |                 |  |  | 50 (13) 193 SS rokszes. Repticon obne.<br>Elektrics in swes. Coffing 2 jon skectric chein hold. Coffing 1/2 jon electric hold, ext tadopy.<br>Double cut band saw bitales, edger guides, (7) circular saw hold cradies, Coffing 1/2 ion electric hold, circular saw blades, 1/2 ion  |   |           |
| 130   |                |            |                 |  |  | Double cut band saw blades, edger guides, (2) circular saw hofst cradies, Coffing 1/2 ton electric hoist, circular saw blades, 1/2 ton electric hoist, circular saw blades, 1/2 ton electric hoist, edpour   |   |           |
| 131   |                | Г          |                 | ****   |  | Williams & White circular saw stratcher roll, 27 throat s/n 7285D, Armit, edger guides, IAMV guide grinder, splined rolls, Armstrong   |   |           |
| 132   |                | 1-         | ├──             |  |  | #18 shear, Colling 1/2 ton electric horst. Position sylinders, sembledes with horst carriers, (2) 62° & 72" insent tooth circular sem blades, circular sem blades, artours, colling 1/2  |   |           |
| 130   |                | ⊢          |                 |  |  | (on else holss, pellet jack, edger guides<br>Allen Bradley bull edger ple (old) Lincoln electine RDR-400 welder, Kilo 500 kg else helst, edger guides, arbeur rolls, Barnhardt grinder,  |   |           |
| 134   |                | ╙          |                 |  |  | HAV guide grades   |   |           |
|   |                | L          |                 |  |  | (2) Torch sets, 5' bend mill spaked wheels, (2) Kito 1/2 ton electric hoists, Amistrong weld clamp, CM loadstar 1 ton electric hoist, saw<br>cradles, fell restraint.  |   |           |
| 136   | NCC-19         | -          | -               | ILAV   |  | Filing troom electrics, 25 kvs transformer<br>Guide grinder  |   |           |
| 137   |                |            |                 |  |  | Set up stand, bebbit pot, 17 & 17 U7 emits, quides, step ladder  |   |           |
| 130   |                |            |                 | I <del>MAY</del>   | 240  | Edger guides, babbit, micrometers & diel gevoes, orneits surface block<br>Guide grinder  |   |           |
| 140   |                | -          | -               | Wright<br>Armstrong  | Taion TF-3   | Circular saw sharpener  Band saw leveling bench, #5 stretcher roll s/n 6735, (2) S X 14° emits and clamp   | Q114021F3                                       | 6.383 Ns  |
| 142   |                | 1          | _               | LAM  |  | Mood size chain bara   |   |           |
| 143   |                |            |                 | Avmskrong.   |  | Roth hand circular say sharpener<br>Grinding wheels begrings and parts   |   |           |
| 145   |                | -          | -               | Armstrong  | <b>315</b>   | Circular serv pherpener Circular serv blades, First LCN-14 drill press, King 4" X 36" belt sender, set up stand, jigs, chipper loufe clamps, Armstrong tension   |   |           |
| i   |                | <u> </u>   |                 |  | L  | gauge grinder  |   |           |
| 147   |                | 1          | -               |  |  | heelson post grinder, 23' circular saw armi, 10' X 6' amil, chain saw post grinder<br>Acme junker post sharpsmer, hand swedges and shapers, Waliama & White circular saw looth setter  |   |           |
| 149   | MCC-198        |            |                 |  |  | Microd chipper knives, jig, DeWalt grinder, hand looks<br>Microd efections   |   |           |
| 150<br>151  |                |            |                 | Wright   | W-350HD  | Circular saw sido grinder  | 19510   |           |
| 152   |                | -          | 1996            | Reform   | AR21/51  | Traveling head Infle grinder, (2) 4 1/4" X 5" magnetic clamps Pneumatic babbit (ig., babbit pot, pouring (ig., chipper Innives   | 589496  |           |
| 154   |                |            |                 | VIGOR  | Cortex   | Knife grinder 75 hp, with knives   | 0418010TX1                                      |           |
| 156<br>156  |                |            |                 | Armstong<br>Henshelt   | n  | Post grinder, hose, saw blades<br>Circulat saw sharpener, circulat saw tipping bench with 9" X 15" anvil   |   |           |
| 157   |                | +-         | -               | Wright   | W-350 HQS  | Coffing 1/2 Ion electric hoist, (2) 1/2 Ion chain hoists, overhead beam, Irollys Circular saw side grinder, Burton pro 9/3 affachment  | A9519   |           |
| 159   |                |            |                 | Wiright  | W-350 HO B   | Circulat saw top grinder   | E9627   |           |
| 150   |                |            |                 | Wright   | W-350 HD B   | Carular sew top grinder<br>Set up bench, 24° & 23° anvils, lipping station, sowe circular saw blades<br>Bund sew sharpener, andro feed, clamp and stands   | E9625   |           |
| 162   |                | Γ,         | F               | Armstrong Armstrong  | #8 R.H. 5242   | Band saw sharpener, aido food, clamp and stands<br>Hand Swedges with (3) hand shapers  | 51186   |           |
| 154   |                |            |                 |  |  | Cat walk and cransvars above for had infeed 42" w  |   |           |
| 165<br>166  |                | $\vdash$   |                 | Amstrong<br>Amstrong   | MILH<br>MRH  | Band saw sharpener, auto feed, 858 clamp and stands, shape up attachment. Band saw sharpener, auto feed, 858 clamp and stands, shape up attachment.  | 8028  |           |
| 167   |                | _          |                 | Witterns & White   |  | Circular saw strotcher roll 17 1/2" throat sin 35070, Williams & White saw stretcher roll sin 351450F, grinder, cart   |   |           |
| 189   |                | 1          |                 | Williams & White   |  | Bandsaw leveling bench, 17 1/2" stretcher s/n 17608 clamp, (Z) 12" X 46" anvils<br>Splined circular saw blades, Milar Multimatic 215 welder  |   |           |
| 170   |                | <u> </u>   | $\vdash$        |  |  | 15 kva transformer and electrics Makyta guide grinder, hand tools, vise, guides and parts  |   |           |
| 172   | 33-25          | _          |                 |  |  | 90' X 14" belt conveyor, s-drive, fiberglass insert, MEX metal detector with drive   |   |           |
| 173   | 35-12          |            |                 | Cencer   |  | Chipper, R. H., side feed, bottom discharge, 150 hp 1780 rpm, with starter<br>10" box chain clean out conveyor, 5 hp Radicon   | 202   |           |
| 175   | 35-40          | -          | _               | Rawlings   |  | 120' X 48' belt conveyor, troughed roll, s-drive, shaft mount drive (hog infeed)  Hammar hog top feed, bottom discharge, plw hvin 350 hp. 1770 rpm drives, 10 hydraufic assist opening, (2) starters, spare screan   | 900510  |           |
|   |                | L          |                 |  |  |  |   |           |
| 177   | 35-12          | -          |                 |  |  | 6 dip 6, 15 dia cyclones, common stoel stand, blowpine<br>235 X45° bell conveyor, licusphed rys, shed ineme, cahrallung, sprintders, metal clad, steel supports, 25 hp 1775 rpm drivs  |   |           |
| 178   |                |            |                 |  |  | Electrics and blower   |   |           |
| 180   | 35-19          | 匚          |                 |  |  | 12 Da cyclone, 5 die cyclone (top of bine rough shape) 25 X 10° belt conveyor, sheft mount dive  |   |           |
| 183   | 35-72          |            | H               |  |  | 45' hein strand box chein clean out common, 5 ho I folhopar reducer 75 kvs transformer, electrics, Uncoln R3R-400 welder   |   |           |
| 164   |                | 1          |                 |  |  | 25 livs transformer, etectrics, Uncoh POR-400 welder<br>(4) lock blocks, (5) ghyms of incinsusc oil, lube room, (2) pneumatic pumps, steel hobbing tenks and grease  |   |           |
| 105   | 33-26<br>34-58 | -          | $\vdash \vdash$ |  |  | 35 X 17" bell conveyor, s-drive, 10 hp Hghgear reducer 35 shole strand box chain clean out conveyor, s-drive with motor  |   |           |
| 107   | 33-44          |            |                 |  |  | 37 pinds stand box (her) stem out conveyor, I done with motor<br>10X33 blower hips RSST9 100 hip blower, Rader 20X2S feeder 5 hip and control board  |   |           |
| 189   | 34-80          |            | 1973            | Holmes   |  | 10X33 blower type 1085 19 100 hip blower, Heder 20X25 feeder 5 hip and control board.  50 elogie strand box chain clean out conveyor, e-dive with motor.   |   |           |
| 190   | 35.20          | -          |                 | Edem<br>Harsel   |  | 40 X 18" vibratory conveyor, MOI metal detactor  | EJ717   |           |
| 182   |                |            |                 | DIST.  |  | Chieper, tida feed, bottom discherne, 8 kmis, 300hp, contex kmis protern, player, brake and ports (drive disconnected)<br>Refuse protern posit button control  |   |           |
| 193   | 35-1<br>36-35  | ├          | $\vdash$        |  |  | 100 X 38" bell conveyor, troughed raft, convert, a-drive<br>15" 4 strand h78 (monter deck, Radicon drive (re-entry)  |   |           |
| 195   |                | _          |                 | 2.07.11  | L  | Spare electric perts   |   |           |
| 197   | 33-24          |            |                 | Cutter Hummer  |  | Macc 14 paction, 69 protot, Allen Bradley plc<br>BS X 18" ball convensy, a-stine, Radicon  |   |           |
| 198   | 34-95<br>34-37 | -          | -               |  |  | 130' box chain clean cul conveyor, a-drive, Redicon  |   |           |
| 200   | 34-30          |            |                 |  |  | 30 bax chain olden out conveyor, s-drive, Radicon<br>30 bax chain delen out conveyor, s-drive, 5 po  |   |           |
| 201   | 34-43          |            |                 |  | 1 1  | 16 box chain clean out conveyor, a-drine. 5 hp<br>15 box chain clean out conveyor, a-drine, 5 hp   |   |           |
| 202   | 34-24<br>33-27 |            |                 |  |  | 27 pou chaîn chen out convenor, a drye. 5 hp. 45 bou chaîn chen out conveyor, Radicon drye   |   |           |
| 205   |                | $\vdash$   |                 |  |  | 50 km transformer with electroics, Lincoln RSR-400 welder  |   |           |
| 206<br>207  | 33-44          | +-         | -               |  |  | 50° box chain clean out conveyor. Radicon drive. 125° box chain clean out conveyor. Radicon drive.   |   |           |
| 208 L   | 33-40          |            |                 |  |  | 42, 8 1/2 long link clasm out conveyor, p-d/ye, Radicon  |   |           |
| 209<br>210  | 27-20          | $\vdash$   |                 |  |  | 160', 6 1/7' long link clean out conveyor, a dave, Radicon 65', 6 1/7' long link clean out conveyor, a dave, Radicon   |   |           |
| 211   | 33-57          |            | $\equiv$        |  |  | 85', 8 UZ' long link class out conveyor, s-drive, Radicon  |   |           |
| 212   | 33-58<br>33-78 |            |                 |  |  | 255 (6 VZ long link clean out conversor, a-drive, Gadison 12 X 17 bell conversor with drive  |   |           |
| 214   | 33-32          | 1          | $\vdash$        |  | 1  | 120', 6 1/2" long linit clean gut conveyor, a-drive, Radicon 60', 6 1/2" long linit clean gut conveyor, a-drive, 5 ho Hollond reducer  |   |           |
| 215   | 33-23          | F-         |                 |  |  | BU, 6 LIT long link chien out commyor, s-drive, 5 hp drive   |   |           |
| 217   | 33-31          |            |                 |  |  | 45 box chain clean out conveyor with drive 312 X 30' boughed rell bait conveyor, e-drive with drive (main chip)  |   |           |
| 219   | 33-18<br>33-18 | F          |                 |  |  | 120 pas chain clean out common, a dinn, 10 hp<br>45 bas chuin clean out common, a chine, Radicon dine  |   |           |
| 221   | 33-17          |            |                 |  |  | 25 box chain class put conveyor, s-drive, Radicon drive  |   |           |
| 223   | 34110          | <u></u>    |                 |  |  | 87 X 16 1/2" bet conveyor with office  |   |           |
| 224   |                |            |                 | Time and   |  | Electric phop, neir parts, connectors, offices, computers, screens, scrollers bar tricks, computer rack<br>Electrics   |   |           |
| 226<br>226  |                | 1          | -               | Siemera  |  | Pipe shop, (3) chain saws, Atles Copco parts, pipe fittings, figings birs. Victaulic fittings, frangers, födgid threader, pipe culturs, fire   |   |           |
|   |                |            | <u> </u>        |  |  | hose station, vise, valves<br>1 section 2 switch pdc, capacitors and electrics   |   |           |
|   |                | =          | 1973            | GE   |  | Transformer, 1,500 kva, 3 ohase, 60 kz, 25 000 -480V   |   |           |
| 227<br>228  |                | $\vdash$   | 1983            | Carte  |  | Transformer, 2,000 kvs. 3 phase, 60 hz, 25,000 kW, 2400Y/1386 LV Main 25,000v incoming enclosed switch, WR & Wostinghouse 2 section main drive control 25,000V (log chipper) Westinghouse 2  | L0255-1   |           |
| 228<br>229  |                | 1          |                 | A 11:-   |  | section 1 switch DC drive control (log chipper) and electrics  |   |           |
| 728<br>729<br>230   |                | ⊢          |                 | Cuttler Hammer   |  | Mec 3 section 10 switch 60' box chain clean out conveyor with s-drive, Radicon   |   |           |
| 228<br>229<br>230<br>231  |                | -          |                 |  |  | Progracks nine filtrons Richard \$35 threader make lack hose compressor of   |   |           |
| 228<br>230<br>230<br>231<br>232<br>232  |                |            |                 |  |  |  |   |           |
| 228<br>229<br>230<br>231<br>232   |                |            |                 |  |  | Ima P12C mill drill vises, new electric parts, fitings, wire rolls, (4) badders<br>Electric shop, hand tools, snakes, hole saw bits, Helt DX450, Milwaukee band sew, Brady lebel printers, Attings bins, Milwaukee Fuel  |   |           |
| 220<br>230<br>230<br>231<br>232<br>232<br>233<br>234<br>235<br>235  |                |            |                 |  |  | bil, connectors  |   |           |
| 229<br>230<br>231<br>232<br>232<br>233<br>234<br>235<br>235<br>236<br>237   | 35-11/2        |            |                 |  |  | Ist, connectors<br>(7) affectoolers<br>45 Astand Fimber dock transfer, with drive  |   |           |
| 278<br>279<br>230<br>231<br>232<br>233<br>234<br>236<br>236<br>237<br>238<br>237<br>238   | 38-110         |            |                 | Subsir<br>Atlan coord  | 25-15QL  | M, consoleto<br>(), affeccolet<br>45 Astand innbe deck transfe, with drive<br>Raday saters wit compressor, 1900b, shafe  | 000-69112                                       | 39.091 h  |
| 228<br>229<br>230<br>231<br>232<br>233<br>234<br>235<br>236<br>237<br>238<br>239  | 35-110         |            |                 | Alba cooco<br>Sufant   | 25-150L<br>GA 1107<br>258-150L WCAC  | M, connection (7) affectedors 45 4 shand inher deck transfer, with drive Ralay sterer air compensar, 190 hp, Bullet   | 50278701-2<br>32351-CGG                         |           |
| 228<br>229<br>230<br>221<br>222<br>234<br>234<br>235<br>236<br>237<br>238<br>239<br>240<br>240<br>241   | 35-110         |            |                 | Atlas copco  | 25-15QL<br>GA 1107   | M, connection (7) affectodois 45 4 shand inher deck transfer, with drive Relays sterne air compensar, 150 hp, Mulle Relays sterne air compensar, 150 hp, Mulle Relays sterne air compensar, 150 hp, dates Relays sterne air compensar, 150 hp, dates Relays sterne air compensar, 150 hp, dates Air drive mystem, (7) LT ar reservoir quinte 40° A 60° Relay without set den data but Cack with but set den data but Cack with but set den data but Cack without set den data but  | 50278701-2                                      |           |
| 278<br>239<br>230<br>231<br>232<br>232<br>233<br>234<br>235<br>236<br>239<br>240<br>240<br>240<br>240<br>242  | 35-110         |            |                 | Alba cooco<br>Sufant   | 25-150L<br>QA 1107<br>259-150L WCAG<br>A7SSFARE1   | M, connection (2) districtions 45 Ashand imbar deck lansfer, with drive Reasy server will compenses, 150 to, Budde Robert server all compenses, 150 to, Bu | 50278701-2<br>32351-CGG                         |           |
| 220<br>220<br>220<br>221<br>232<br>232<br>233<br>234<br>235<br>237<br>239<br>240<br>240<br>241<br>242   | 3£110          |            |                 | Alles septre<br>Suitain<br>Pail<br>Bridesport                      | 25-150L<br>GA 1107<br>259-150L WCAG<br>A7SSFARE1   | M, convenion ( ) affectedors  45 Astruct imber deck transfer, with drive  85 Astruct imber deck transfer, with drive  Richary screw air compensor, 150 hp, Marter  Air drive myter, 17 12 ar recenting tubes 45° à 60°  Ceck with has scele or fait her  Million methods, 15, 26. 17 and 18 hp, 18  | 50278701-2<br>32351 CGG<br>(0361-72             |           |
| 226<br>229<br>230<br>231<br>232<br>233<br>234<br>235<br>236<br>236<br>240<br>240<br>241<br>242<br>241<br>242<br>241<br>242<br>243<br>244<br>245<br>246<br>246   | 35-110         |            |                 | Alles cocco Sugai Fast  Brickecoot  Hypertherm Powermatic          | 25-150.<br>QA 1107<br>258-150, WCAC<br>A7SSFARE 1<br>Privarias: 105 pha<br>PSD-620                             | M, convenion ( ) affectedors  45 Astruct imber deck transfer, with drive  85 Astruct imber deck transfer, with drive  Richary screw air compensor, 150 hp, Marter  Air drive myter, 17 12 ar recenting tubes 45° à 60°  Ceck with has scele or fait her  Million methods, 15, 26. 17 and 18 hp, 18  | 50278701-2<br>32351 CGG<br>(0361-72             |           |
| 220<br>229<br>230<br>231<br>232<br>233<br>234<br>235<br>239<br>239<br>240<br>241<br>241<br>242<br>241<br>242<br>241<br>245<br>245<br>246<br>247   | 38-110         |            |                 | Altes socce<br>Sugar<br>Page<br>Brisbesood<br>Hypertherm           | 25-150.<br>GA 1107<br>259-150. WCAC<br>A7SUFABE 1<br>Provember 105 sha<br>PTS-020<br>Golden 452                | M, connection (2) affectedors 45 Astront imber deck transfer, with drive 85 Astront imber deck drive 86 Astron | 50278701-2<br>33351-CGG<br>00501-72<br>J167882  |           |
| 228<br>239<br>231<br>231<br>232<br>233<br>234<br>235<br>236<br>239<br>249<br>249<br>241<br>242<br>241<br>242<br>241<br>242<br>243<br>244<br>245<br>246<br>246<br>246<br>248<br>248<br>248<br>248<br>248<br>248<br>248<br>248<br>248<br>248        | 35-110         |            | 1965            | Alba socco Subai Pai  Snisaccot  Hypertherm Povermatic Make Buttoo | 25-150a<br>GA 1107<br>259-150, WCAC<br>A7SEASE1<br>Powermax 105 sha<br>PRO-920<br>Gobber 452<br>ZY5050A        | Nel, connection (C) affectories (C) affectorie | 50278701-2<br>372361-CGG<br>(0361-72<br>J187882 |           |
| 226<br>229<br>230<br>231<br>232<br>233<br>234<br>235<br>237<br>236<br>237<br>240<br>240<br>241<br>241<br>241<br>241<br>241<br>241<br>245<br>246<br>246<br>246<br>246  | 35-110         |            | 1965            | Alba socco Subin Pal Bindoscot Hypertherm Poversmatic Male         | 25-150.<br>GA 1107<br>259-150. WCAC<br>A7SSFABE 1<br>PSWMMBX 105 pha<br>PSS-920<br>Golden 452                  | Nel, connection Conference of the Conference of  | 50278701-2<br>33351-CGG<br>00501-72<br>J167882  |           |
| 220<br>220<br>230<br>231<br>232<br>233<br>234<br>235<br>236<br>237<br>240<br>240<br>241<br>242<br>241<br>242<br>242<br>243<br>244<br>244<br>245<br>246<br>246<br>247<br>248<br>249<br>249<br>249<br>249<br>249<br>249<br>249<br>249<br>249<br>249 | 35-110         |            | 1965            | Alba socco Subai Pai  Snisaccot  Hypertherm Povermatic Make Buttoo | 25-150a<br>GA 1107<br>259-150, WCAC<br>A7SEASE1<br>Powermax 105 sha<br>PRO-920<br>Gobber 452<br>ZY5050A        | Nel, connection (C) affectories (C) affectorie | 50278701-2<br>33351-CGG<br>00501-72<br>J167882  | 39.091 hr |
| 270<br>270<br>270<br>271<br>272<br>273<br>273<br>275<br>275<br>275<br>275<br>276<br>276<br>276<br>276<br>277<br>276<br>276<br>276<br>276<br>276   | 35-110         |            | 1965            | Alba socco Subai Pai  Snisaccot  Hypertherm Povermatic Make Buttoo | 25-150a<br>GA 1107<br>250-1501 WCAG<br>A7S0FARE1<br>Powermax 105 ship<br>PCONTROL 50<br>Godstar 502<br>ZY5950A | M, connection (2) distraccion (2) distraccion (3) distraccion (4) distraccion (5) distraccion (6) distraccion  | 50278701-2<br>33351-CGG<br>00501-72<br>J167882  |           |

| UNIVAR<br>RID.#   | Y - SKEENA SA<br>Unit # | men.         | Year     | Make  | Model  | Description  | Secial Number                           | Hra/Kr   |
|---|-------------------------|--------------|----------|---|--|--|---|--|
| 258<br>258  |                         | Ε-           | F        |   | <del>                                     </del>     | Fatings bing, sockers, shim bits, bearings, lest banch, vise, splety sitros, hose, crámbra, y bolls<br>Chain holpts, bram clamos, sángs  |   |  |
| 757   |                         |              |          |   |  | Communic pumps, by steel busings, drive chain, Postboring cylinders, parts washer, steel bench, bearings   |   |  |
| 250   |                         | <del> </del> |          |   |  | Positioning cylinders, parts washer, steet bench, bearings Higgivens, air valves,  |   | ├  |
| 260   |                         | -            | $\vdash$ |   |  | Spare machine parts, edger's, ethecker, screen, bench with vite  |   | 1  |
| 262   |                         | 1            |          |   |  | Bushings, spare machine parts canter, bearing seals, stabber, vibratory conveyor, him band, chippens Oil seals, bearings   |   |  |
| 263   | L                       | L            |          |   |  | Electric chain holes, and manual with trollys, hand loots, but splicing litts, wrenches, banch grinder, Eastman K24 hydraulic crimper,<br>bearing warner, flammable cabinet, vise, drill crimse. Energic power suspies, jorda  |   | 1  |
| 264   |                         | -            |          |   |  | Hydraulic lighnos, hydraulic dylws carts and supplies, bar fridde<br>Hand foots, torch sets, torque wranches, Milwaukee band saw, abill saw, Nove-Pro 500 kil, Office, CPU, monitors, salety harnesses.  |   | -  |
|   |                         | ┡            | ļ        |   | ļ  | greese guns  |   | <u> </u>   |
| 205<br>267  |                         |              |          | BMM   | <u> </u>   | Lunch morn conferes Chip classifier, grafe, even   |   | ├─   |
| 266   | 34.29                   | $\vdash$     |          |   |  | (3) offices, decks, file cabinets, chairs, couls flat screens, babiets, radio's, BX T3-19 rechargeable bender  |   | =  |
| 270   | Li-E                    |              |          |   | 1  | 35 A strand box chain conveyer, 7.5 hp Redicon drive  Motor stores area, small drives, (4) pallets 2 hp-5 hp, (2) pallets 7.5 hp-10 hp, (2) pallets 15-20 hp, (2) pallets 25 hp, (3) pallets 30 hp.  |   | <del>                                     </del> |
| 271   |                         | -            | -        |   | <del> </del>   | (24) motors 25 hp-400 hp. Spare used parts by vibratory conveyor, winch, positioning cylinders, chain, shines, limitys, hyd drives, wheels, power pack, ice melt   |   | ├-   |
| 272   |                         | -            | -        |   |  | (3) midiors, spoods of wire  |   | ├  |
| 273   |                         |              |          |   |  | (2) 5 kvs transformers, Hammond cebinet, 25 kvs transformer, (2) System direction soft start cabinets 400 hp, (2) Hammond control  |   | $\vdash$   |
| 27.4  | 34-81                   | $\vdash$     | -        | Eden .  | ł  | Sabinata<br>130 X 18" wibrationy converser, 7,5 hp, MDL metal detector   | E12577                                  |  |
| 275<br>276  | 3494                    |              |          | CAE   |  | GO pingle strang box chain conveyor, 5 hp<br>Chipper, 85°, 8 kmile, side feed, side deschange, 250 hp 1700 rpm chive, stanter  | 93A097                                  | =  |
| 277   | 3462                    |              |          | νs  |  | 75 hein strand box chain conveyor, s-drive, 10 hp Rasicon  | SSC(97                                  |  |
| 279   | 3480<br>3477            |              |          |   |  | 15' 4 strand h78 transfer deck, 10hp Radicon drive<br>12' 4 strand came/back transfer deck, Radicon drive  |   |  |
| 200   | 34.78                   |              |          |   | 1  | 10' (3) twin strand flytod unscrambler, Radicon drive  |   |  |
| 281<br>282  | 3474                    | 1-           | _        | QM .  | <b> </b>   | 25' (5) roll 20' rollcase Radicon drive, pop up stops.  Roman edger, 30', 100 hp, 3 sew, splined arbour, he'n top hold down rolls, 5 bed rolls, top opening, 10 hp feed drive, 3 lesser (ghts,   |   | ├-   |
| 283   | 34-69                   | -            | _        |   | ļ  | push button control, efectrics, Allen Bradley pic  |   | ļ  |
| 284   |                         |              |          |   |  | 20 X 16" belt conveyor, 5 hp drive, picker hipe diverting station, 6 1/2" X 24" belt conveyor 5 hp Pdc #1, Westinghouse switch, TCY transformer, Square D & Allen Bradley electrics  |   |  |
| 285<br>286  | 34-11<br>34-75          | $\vdash$     |          |   | ļ  | 35 X 24" belt conveyor with drive  |   |  |
| 207   | 35-11                   |              |          |   | (  | -sc a 15 mile terreport, 3 rip resource once 47 X 17 bet compeyor, shaft mount drive 10 5 strand H100 cameback brander deck, Rudson drive  |   |  |
| 75  | 34-113                  |              |          |   |  | 145' 4 strand H78 return transfer deck, Radicon dine, steel support standards and railings   |   | <del> </del>                                     |
| 290   | 34-24                   | <b></b>      |          |   |  | 52 9 strand H130 came back transler deck, Radicon drive  |   |  |
| 291   | 34-27                   | "            |          |   |  | Disconnects in area 16 T strand mixed chain transfer dock, Radicon drive   |   | _  |
| 203   | 34-26<br>35-10          |              |          |   |  | 12' 6 strand H130 cameback transfer dock   |   | -  |
| 205   | 35-8                    |              |          |   |  | 35' 6 strand transfer dock 10 hp Radicon drive<br>12' 6 strand transfer dock ultrimw lined common drive  |   | $\vdash$   |
| 290<br>297  | 35-12<br>PDC 18         | $\vdash$     | -        |   | <b></b>  | 9 (5) strand lynn Byled unscrambler, 1/2 ton bearn hoist, 20 hp drive Westinghouse and Square D electrics  |   | -  |
| 298   |                         |              |          | C. Aller Van  |  | 6 (6) tolk evenending rollicase, 6 but ending station, 8 6 strand transfer dock twin drives  |   |  |
| 300   |                         |              |          | Cuttler Hammer  |  | Mcc 5 section 22 switch VId drive controls, (?) Yaskawa Matrix U1000 (unscramble) , Toshiba ESP-130 (backlog storage)  |   | <del> </del>                                     |
| 301   | 35-200, 35-201          |              |          |   |  | via one cannots, (g) i seema main too (enteramon), i territories 1.50 (estend) to i cannot starge)  6 6 strand 80 for terrater dect, 6 6 strand 80 for transfer dect with gives Lug leader station with 8 6 strand dynw ined transfer det with gives Lug leader station with 8 6 strand dynw ined transfer det are singulator, 5 6 strand throughlend, 8 arm singulator with are tickback, 1  17 6 mm with this mit deal formation of the formation of the singulator.   |   | <u> </u>   |
| 1   |                         |              |          |   | <u> </u>   |  |   |  |
| 303   |                         | H            |          | Hernes  | <b> </b>   | Posh button controls with Eston ewitch.  10' 6 strand lugged transfer, 10' 9 strand lugged utmove level sciencer throughleed transfer (6) even ending roles, 23' dia over length and   |   | -  |
|   |                         | <u> </u>     |          |   |  | Um serv station  |   | <u> </u>   |
| - 308   |                         | Н            | $\vdash$ | Newnes/USHR<br>Newnes   | BEALQS 1208P   | 20' page scapner station USAFI couls  15 6 strand lugged uhmer lined pre-position transfer (3) pneumatic positioned and paddle positioners, (6) 10 1/2' even ending rells, (4)   | F93-024                                 | <del> </del>                                     |
|   | ~ ~ ~                   | -            |          |   |  | sets of 5 arm pop up singulators   | *************************************** | <u> </u>   |
| 30  | 35-1B                   |              |          | Newnes  | TA:20-U+09   | Trimmer, 20 pass, 11 pay line shall drive, approx 25 hp, 19 10 strand topped throughlead transfer  Spare sprockets, say bladge, electrics, lorch set, Lincoln RCR-400 welder, 25 hys transformer with penal  | 93-006-KA                               |  |
| 309   |                         |              |          |   |  | Spare sprochats, seer blades, electrics, borth set, Lincoln R3R-00 resider, 25 has bandionner with people.  3 102 6 arm roller chain transferiferenting gate, 18 10 strand lugged ohner lined transfer with (10) 15 even ending rolls and drive  |   |  |
| 310   |                         | $\vdash$     | _        |   |  | 2022 ST carn systems 5.13.4 board length scarning raw data station, Northern Mittach Incorporated moisture meter frame (no sensors)  |   | <del> </del>                                     |
| 311   | 35-35                   | $\vdash$     | -        |   |  | # 4 strand lugged throughteed immire with price 50° 5 strand lugged 90 degree transfer coronion drive  |   | ├  |
| 312   |                         |              |          |   |  | Tremmer control room, spere parts, Alten Bradley data kner, (2) Alten Bradley plc's, monitors. J bar sorter system, 71 bin with 22'S etrend kugged transfer deck, Siemens cabinet with Yaskawa Matrix U1000 vid., 12 sets of bin.  |   | _  |
| 313   |                         |              |          | Moore/Heymes  |  | controls, 67 Moore birm (3) Newnes, 38 1/4" bin width, Opcon & Banner photo eyes, _reader boards, (9) sets of 35 5 strand a-drive  |   | l  |
|   |                         |              |          |   |  | ultime fined take every transfer decks with 7.5 hp. drives, 10' 4 strand transfer deck, 30 hp hydrautic poverpack, air receiving tards, Win  |   | l  |
| 314   |                         |              |          | Cuttler Hammer  |  | Tally 6.1 cou.  5 section 16 switch mcc, (2) 2.5 kvs transformers with switch's, Cuttler Havarrer 4 section 13 switch mcc.   | *************************************** |  |
| 315   | 35-57                   | -            |          |   |  | 2Z 5 strand ultrave fined a drive transfer deck, 7 5 hp<br>6 1/Z (d) hyrin strand Byled unscrambles with drive, orenhead beam and bothy  |   | -  |
| 217   | 36-77                   |              | =        |   |  | 45 bot chain clean auf convenor 3 ho<br>Hoore 20' stacker system mech/hyd with 16' 5 strand roller chain heed transfer ultraw fined, 3 1/2' but even ending station, 4 arm   |   | _  |
| 318   | 33-17                   | 1            |          |   |  |  |   | l  |
|   |                         | ll           |          |   |  | with driver, 25 top trydrautic power pack, push button control, 60" twin strand 80 dx sticker bin transfer with drive  |   | 1  |
| 318   |                         |              |          |   |  | (7) 1/2 ton chain hoists, 80° of 1 been  |   |  |
| 320   |                         |              |          | Cultier Hammer  |  | Micc 4 section 14 switch, 25 kvs transformer with switch, Yaskawa Malric U1000 vfd (incline chain) (2) Yaskawa Malric U1000 vfd's (stacker)  |   | l  |
| 321   |                         |              |          |   |  | Trimmer/sorter building 560' L. X 40' W. steel frame, metal clad bott locativer, sprividers, biolics   |   |  |
| 쮨   |                         | 70           |          |   |  | (5) 4' X 10' steel stickes secte. (2) demande recks. Smolth cyl reck, severall, baroling, steel words tim, (20) perts bith carts<br>18', 18' X 12' yerd tembers  |   | $\vdash$   |
| 324<br>325  |                         | -            | -        |   |  | Microst stylet in area stong wall<br>(2) steel lyngth cut racks, durviage rack; (2) bending casts, (15) rofs of bending  |   | -  |
| 320   |                         |              |          |   |  | Reducers and motors in gree  |   | _  |
| 327   |                         | Н            | 2003     |   |  | 42 shipping container, welded steel feel Kiin sate parts and steel in site, beling and rail on container rool  |   | <del>-</del>                                     |
| 320   |                         |              | 2010     |   | I  | 40 stripping combines, prelided at ear foot.<br>28 sections of 4 Mar shahme, shives, pallow block beautings, bouldings, spruckate mitted pilch and replaces parts.   |   | _  |
|   |                         |              |          |   | ļ  |  |   | -  |
| 330   |                         |              |          |   |  | Sorter storage area, apitted rolls, (S) doubte cut saw bladge, shalls, debarter ring parts   | *********                               | F  |
|   |                         | H            |          |   |  | Sortier storage area, spilland nots. (3) doubte rus says labeles, thatis, debander ring parts.<br>2 section patient mick two bier, cylimiders, reducers, sprockets, chain , conveyor rolls, carriage cable, Hydrostar hyd drive, blower  |   | F  |
| 233<br>225<br>221   |                         |              |          |   |  | Sotiet Kirppe illines, spilland (1995, (3) double upt serv lifesjeer, pitulis, debarter ring parts.  Zusckin paller nich hor bier, spillands, nebusans, sproculates, chein , conveyor nals, carriege cable, Hydrostar hyd diner, blower  Conveyor nals, charte, cyferders, reducers.   |   |  |
| 335<br>331  |                         |              |          |   |  | Social Koroga larsa, spiked offs, (3) double out are higher, fields, debater for goats,<br>currier paids not be better position to be the control of the company of the compa |   |  |
| 334<br>334<br>335   |                         |              |          |   |  | Social storage area, spiked offs, (3) double out any basice, storage, or considering out of the considering out of   |   |  |
| 224<br>233<br>233<br>233  |                         |              |          |   |  | Sofie Korosa area, spiked offs, (3) Ooder on any layee, prouts, dehard ving parts Zuckinn palled mit he heir, dynfarin, shooten, sprochests, chain, conveyor nels, carriege cable, Hydrostal hyd drive, blower Conveyor note, during, prinders, reduces. 3 soction palled mitch, reduces, 1900 day valve, bossile pump, edgar valve system pump (ii) (120), lead note, shives, postdoning ordinders. Wood bin, lead note, hand new steel, conversor pole, best note. 1 section palled mitch, mind new steel, conversor pole, best note. 1 section palled mitch, mind new steel, conversor pole, best note. 1 section palled mitch, mind new steel, conversor pole, best note. 1 section palled mitch mind not on the note of conversor pole.   |   |  |
| 22<br>22<br>23<br>23<br>23<br>23<br>23<br>23<br>23<br>23<br>23<br>23<br>23<br>2   |                         |              |          |   |  | Social Kinggal area, spiked offs, (3) double out are higher, fields, debatin ring parts outcompation from their cylindrics, modern, sponsober, button, commanders, ceremique cable, hydrodial hydr drive, belower Conveyor or their, desting, principles, modern outcompation for chances, the condition of control or control or control outcompation for chances, the condition of control or rindrich.  J. socion failed rick, mixed crew the control or right, shert rick J. socion failed rick, mixed crew the control or right, shert rick J. socion failed rick, mixed crew the control or right. J. socion failed rick is with crew these root control or right. J. socion failed rick is with crew these root control or right. J. socion failed rick is with crew these root controls or right. J. socion failed rick is with crew these root controls or right. J. socion failed rick is with crew these root controls or right. J. socion failed rick is with crew these root controls or right. J. socion failed rick is with crew these root controls or right. J. socion failed rick is with crew these roots of rick is recommended by the rick of rick is recommended by the rick is recommended.  J. socion failed rick is recommended by the rick is recommended by the rick is recommended by the rick is recommended.  J. socion fail rick is recommended by the rick is re   |   |  |
| 331<br>332<br>334<br>336<br>336<br>337<br>336<br>336  |                         |              |          |   |  | Social science area, spiked offs, (3) Octobe out are things, pitches, debtain ving parts carbon paids not be the cylindrics, moderate, spondows of their, conveyor note, certified cable, hydrodizal hydridise, belower Conveyor on the, chaffer, pitches, moderate a control paid not chances, flore day valves, bosales pump, adjes water system pump (ii) (120), feed not s, allows, positioning cylindrics.  2 best (in) and moderate press press, common prids, pitches, and 2 best (in) and moderate press, pitches, common prids, pitches, 2 best (in) and moderate press, pitches, common prids, pitches, 2 best (in) and moderate press, pitches, common prids, pitches, 2 best (in) and moderate press, pitches, common prids, pitches, 2 best (in) and moderate press, pitches, in) paidest of chairs, signed bin, (1) and conveyor prids, in, (2) paidest of chairs, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, in, (3) paidest of chair, signed bin, (1) and conveyor prids, (3) paidest (3) paidest (4) paide   |   |  |
| 23 23 23 23 23 23 23 23 23 23 23 23 23 2  |                         |              | 2001     |   |  | Social science area, subbot offs, (3) double out are things, (but all, debut and recognition of the control of    |   |  |
| 22 22 22 22 22 22 22 22 22 22 22 22 22  |                         | 51           | 2001     |   |  | Social science area, subbot offs, (3) double out are things, (but all, debut and recognition of the control of    |   |  |
| 201<br>202<br>203<br>205<br>205<br>205<br>205<br>205<br>205<br>205<br>205<br>205<br>205   |                         | \$1          | -        |   |  | Social siderage area, spiked offs, (3) double out any budger, that is, debrarier ring parts current parts must be all right offs, (3) double out any budger area controlled to this, debrarier, incheses 3 contino parts must, duffin, geleder, incheses 3 contino parts must, duffin, geleder, incheses 3 contino parts must, exclusive, lippe dry rather, bootaler pump, edger wrater system pump (1) (120), fixed roll is, all-lives, positioning cyclicides. 3 contino parts must reserve their common (rolls, literal dick). 3 lession after rick with more chain and governor (rolls, inchese). 5 lession after rick with more chain and governor (rolls, inchese). 5 lession after rick with more chain and governor (rolls, inchese). 5 lession after rick with more chain and governor (rolls, inchese). 5 lession after rick with more chain and governor (rolls, inchese). 5 lession and rick with conveyor chain is, (3) parties of chain, spel bin. (1) Profession of company of their 5 lession series (rolls to be profession). After its to bloover 125 hp. 45 dis blower (mill mod ) 507 of steal 107 blow pips, 107 of 407 abhorities) disc.  5 continued in the control of their contro   |   |  |
| 332<br>332<br>333<br>334<br>335<br>337<br>339<br>340<br>341<br>343<br>343   |                         | 51           | 2001     |   |  | Societ along area, spiked offs, (3) double out any larger, that is, debrarin ring parts carbon paid mich which, cylinder, inchocen 2 section paid mich which, cylinder, inchocen 3 section paid mich, challen, fine der inchocen 3 section paid mich, challen, fine dry rinke, booaler pump, edger writer system pump fill (120), fixed roll is, shiven, positioning cylinder, cylinder, cylinder, 2 section paid mich, inchocen 3 section paid mich, inchocen 3 section paid mich mich rever steels, set, () paidest of chain, spellish (13)-challen of company chain (14)-challen of company chain (15)-challen of company chain (15)-challen of company chain (15)-challen of company chain (16)-challen of company chain (16)-challen of company chain (17)-challen of company chain (18)-challen of chain (18)-   |   |  |
| 331<br>332<br>333<br>334<br>335<br>337<br>339<br>340<br>341<br>342<br>343<br>343<br>345<br>346  |                         | 51           | -        |   |  | Social Kingga area, spiked offs, (3) Goodie sus are higher, fields, debants frog parts contempted in the bill register, industrial contempted in the bill, cylindrian suchases, spocotos, of the company one, cereinge cable, hydroidal hydrian, belower Contract or this, define, priceder, spikeder, the contempted in the c   |   |  |
| 22 22 22 22 22 22 22 22 22 22 22 22 22  |                         | \$1          | -        |   |  | Social sideous area, subhall offs, 10, 00 dode, out are higher, thatas, debharts for goats of a school path of the higher shall, offseder, indicated by a school path offseder offseder, indicated by a school path offseder offs   |   |  |
| 331<br>332<br>333<br>335<br>336<br>337<br>339<br>340<br>341<br>343<br>345<br>346<br>346<br>346<br>346<br>346<br>346<br>346<br>346<br>346<br>346 |                         | 51           | 2010     | S delair  | 20-100LACAC  | Social science area, subbled offs, (3) double out any larger, that is, debrarier for gents of the control of th   | 503-50921                               |  |
| 331<br>332<br>333<br>334<br>335<br>337<br>338<br>337<br>338<br>340<br>341<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343        |                         | \$1          | 2010     | Pak   | 20-100LACAC  | Social sideogal area, spiked offs, (1), Ododin out are things, pitches, pit   | 305 65571<br>O 197-45                   |  |
| 201<br>332<br>333<br>334<br>335<br>337<br>339<br>339<br>340<br>341<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343               |                         | 51           | 2010     | Suthir<br>Pak<br>Technologiai Inc                                 | 20-100LACAC  | Social Kingga area, spiked offs, (1), 00 odds, out are higher, fields, debative frog parti.  Z unkern patien from beiter, cylinder, modern, sporocole, othic, conveyor end, cereinge cable, hydrodial hydr drive, belower  Conveyor or title, defile, predictor, inducers.  3 socion patient for, chackers, the code of your year. booster pump, edger water system pump (ii) (12O), lead rod s, shives, povidoring  or herbots.  3 socion patient code, modern or service conveyor (nds. shert lock)  4 socion for the code of the code o   |   |  |
| 231<br>232<br>233<br>233<br>233<br>233<br>233<br>233<br>233<br>233  |                         | 51           | 2010     | Pak   | 20-100LACAC  | Social storage area, speked offs, (3) Goodes out are things, globals, globals, rich goods, commission and the commission of the commission   |   |  |
| 201<br>332<br>333<br>334<br>335<br>337<br>339<br>339<br>340<br>341<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343<br>343               |                         | \$1          | 2010     | Pak   | 70-1000 ACAG<br>14000HA40000                         | Social sizing alma, spiked offs, (3) double out any before, that is, debiant viring parts of the control of the   |   |  |
| 23 23 25 25 25 25 25 25 25 25 25 25 25 25 25  | 91:09<br>97:01:02       | \$1          | 2010     | Pak   | 30-100L ACAC<br>14000+M49000                         | Social Kingga area, spiked offs, (1), 00 dolfs out are higher, fleshs, debating frog parts contention and mobile in cylindrian induces, sporouses, belian, conveyor resis, cereinge cable, hydrosia hydrone, blower Conveyor or tide, distilla, prieder, induces it contention and mobile, cylindrian induces, sporouses, or conveyor resis, cereinge cable, hydrosia hydrone, blower Conveyor or tide, distilla, prieder, induces, by contention pump, adge water system pump (in (120), lead not it, shives, povidoring cylindrian Conveyor or tide, distilla, prieder, induces, by conveyor resis, sheri tide Wood bin, shed neck, mixed crew (rest, conveyor ride, sheri tide) Wood bin, shed neck, mixed crew (rest, conveyor ride, sheri, lead (12), Prieder and conveyor or tide, pols, (1)) palests of them, sheri bin. (13) Prieder and conveyor or tide, pols, (1)) palests of them, sheri bin. (13) Prieder and conveyor or tide, pols, (1)) palests of them, sheri bin. (13) Prieder and conveyor or tide, pols, (1)) palests of them, sheri bin. (13) Prieder and conveyor or tide, pols, (1)) palests of them, sheri bin. (13) Prieder and conveyor or tide, the conveyor or tide, pols, (1)) palests of them, sheri bin. (14) Prieder and conveyor or tide, the conveyor or tide, (1)) palests of the sheri bin. (15) Prieder and conveyor or tide, (1) Prieder and tide, (1) P   |   |  |
| 23 22 23 25 25 25 25 25 25 25 25 25 25 25 25 25   | 01:03                   | 51           | 2010     | Pak   | 30-100L ACAC<br>14000+M49000                         | Social sizing almas, spekhol offs, (3) double out any larger plates, debtant viring parts contention and mobile of printing instances, sportcode, plates, or compare note, cereinge cable, hydroidal hydr drive, belower Contract mith, distillar, principles, inducers 3 streets parts mith, distillar, principles, inducers 3 streets parts mith, distillar, principles, inducers, procedure, but only only offset printing or contract parts (A) social based mich, mixed graw yilled, contract printing (A) social based mich, mixed graw yilled, contract printing (A) social based mich, mixed graw yilled, contract printing (A) social based mich, mixed graw yilled, contract printing (A) social based mich, mixed graw yilled, contract printing (B) based on the mixed printing (B) based on the mixed printing (B) based on the mixed graw yilled, based on the mixed ben. (B) Preliated or compared graw (B) based printing    |   |  |
| 231<br>232<br>233<br>234<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235  | 01:00<br>Q (21:01:02    | 51           | 2010     | Pali<br>Technologies Inc  | 30-100L ACAC<br>14000+M49000                         | Social sizing although the control of the control o   |   |  |
| 231<br>232<br>233<br>233<br>233<br>233<br>233<br>233<br>233<br>233  | 01:03                   | 51           | 2010     | Pali<br>Technologies Inc  | 30-100L ACAC<br>14000+M49000                         | Social Kingga area, spiked offs, (1), 00 dolfs, out are higher, flatts, debants frog parts contention and mobile cylindrian, modern, sporocole, other, common and collections, common and collections.  Contract rolls, distilla, priceder, practices.  3 stordon pallet rolls, debants, priceder, practices, the collection purple, disput spiker pump (ii) (120), lead roll, private, positioning orthodology.  Wood loss, that risk with content per feet, contract rolls, kiled title.  Wood loss, that risk with content per feet, contract rolls, kiled title.  1 section lettin lettin with content per than, post, (1) paletts of them, specified.  1 section lettin lettin with content per than, post, (1) paletts of them, specified.  1 section lettin lettin with content per than, post, (1) paletts of them, specified.  1 section lettin let   |   |  |
| 231<br>232<br>233<br>234<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235<br>235  | 01:00<br>07:21, 01:02   | \$1          | 2010     | Pali<br>Technologies Inc  | 30-100L ACAC<br>14000+M49000                         | Social Moraga area, selected offs, (3) double out as yet pipe, (1945), debuting from part of the control of the control offs offs of the control offs offs offs offs offs offs offs of   |   |  |
| 231<br>232<br>233<br>233<br>233<br>233<br>233<br>233<br>233<br>233  | 91:09                   | \$1          | 2010     | Pali<br>Technologies Inc  | 30-100L ACAC<br>14000+M49000                         | Social Kingga area, spiked offs, (3), 30x66s, sxt, sare higher, fluids, debatin ring parts.  Contrage of mids, shafts, principals, reduced.  3 section patient next work of cyclings, makes.  3 section patient next high, spikeds, reduced.  3 section patient next high, spikeds, reduced, social persons, or supplied to the patient pump in (120), lead not i, shakes, positioning or checks, and social patients.  3 section patients with contrage of cyclings, and state state.  3 section patients with contrage of calls and state state.  3 section state section of contrage of calls.  3 section and state section of contrage of calls.  3 section state section of contrage of calls.  3 section state section of contrage of calls.  3 section state section of calls section of calls.  3 section state section of calls section of calls.  3 section state section of calls section of calls.  3 section of calls section of calls section of calls.  3 section of calls section of calls section of calls.  4 section of calls section of calls section of calls.  5 section of    |   |  |
| 331<br>332<br>333<br>334<br>335<br>337<br>337<br>339<br>339<br>349<br>349<br>349<br>349<br>349<br>349<br>349<br>349                             | 91:00                   | 51           | 2010     | Pair Technologies Inc  Cadtier Hammer                             | 86-1001, ACAC<br>19002-M-9000                        | Social Associations are spekted offs, 10, 00x66s out are tables, debate viring parts. Sections paid in each code, or spectrum, or spectrum, or compare on the compare of the code of the c   |   |  |
| 331<br>332<br>333<br>335<br>337<br>337<br>337<br>337<br>337<br>337<br>337<br>337  | 07-08                   | 51           | 2010     | Pair Technologies Inc  Cadtier Hammer                             | 86-1001, ACAC<br>19002-M-9000                        | Social Moraga area, subball offs, 10, 30x566 acts are higher, fleshs, debatin ring parts.  Social Moraga area, subball offs, 10, 30x566 acts are higher, fleshs, debatin ring, and acts are higher fleshs, conveyor resist, certifies cable, hydroids hydrid rine, belows  Conveyor rinds, shaftly, pfeeders, reduces, 10  Social moraga area, inches, 10  Social moraga area, 10  Social mora   |   |  |
| 331<br>332<br>333<br>334<br>335<br>336<br>337<br>340<br>341<br>341<br>341<br>341<br>341<br>341<br>341<br>341                                    |                         | 51           | 2010     | PAS Technologias Inc Technologias Inc Cutter Hammer Statison Ross | 86-1001, ACAC<br>19002-M-9000                        | Social Moraga area, spiked offs, (1), 00 dolfs, out, are higher, fluids, debants rich goats)  a school path on the Christian, processes, 100 conveyor resists, cereinge cable, hydroids hydridine, blower  Conveyor rolls, drafts, principles, reduces, 1  3 cerein path on the Christian, processes, reduces, 1  3 cerein path on the Christian, processes, reduces, 1  3 cerein path on the Christian of th   |   |  |
| 331<br>332<br>333<br>333<br>333<br>333<br>333<br>333<br>333<br>333  | 07-08                   | \$1          | 2010     | Pair Technologies Inc  Cadtier Hammer                             | 20-1003, ACAC<br>15000:445000                        | Social Associations, spekhol offs, (C) Oxides out any better, that is, debiated viring pattern for gently content and the control of the cont   | 0.192-5                                 |  |
| 331<br>332<br>333<br>333<br>333<br>333<br>333<br>333<br>333<br>333  | 07-08                   | \$1          | 2010     | PAS Technologias Inc Technologias Inc Cutter Hammer Statison Ross | 20-1003, ACAC<br>15000:445000                        | Social Associations, spekhol offs, (C) Oxides out any better, that is, debiated viring pattern for gently content and the control of the cont   | 0.192-5                                 |  |
| 331<br>332<br>333<br>333<br>333<br>333<br>333<br>333<br>333<br>333  | 07-08                   | 51           | 2010     | PAS Technologias Inc Technologias Inc Cutter Hammer Statison Ross | 20-1003, ACAC<br>15000:445000                        | Social Kingga area, spiked offs, (3), 30x66s, six are higher, fields, Sebanta ring parts.  Contract rink, dieffs, priceder, practices.  2 steelers pallet netter, chackers, proceeds, six area spiked pump, a cigar water spiken pump (ii) (120), lead rod, philosophic processors, contract rink, dieffs, priceder, practices, processors, six area spiken pump (ii) (120), lead rod, philosophic processors, chackers, proceder, processors, process   | 0.192-5                                 |  |
| 331<br>332<br>333<br>334<br>335<br>336<br>336<br>337<br>337<br>341<br>341<br>341<br>341<br>341<br>341<br>341<br>341                             | 07-08                   | 51           | 2010     | PAS Technologias Inc Technologias Inc Cutter Hammer Statison Ross | 20-1003, ACAC<br>15000:445000                        | Social siderage area, speked offs, (3) double out as the place, that a, debate viring part of the place of th   | 0.192-5                                 |  |
| 331<br>332<br>333<br>335<br>337<br>337<br>337<br>337<br>337<br>337<br>337<br>337  | 07-08                   | \$1          | 1983     | PAS Technologias Inc Technologias Inc Cutter Hammer Statison Ross | 85-1003, ACAC<br>1900/A4-0000<br>614-O1<br>Gradescen | Social Moraga area, spiked offs, (1), Oscides, out, service, before, the states, debtare, they don't show the control of the states, and the states, and the states are states are states. The states are states are states are states are states are states are states. The states are states. The states are states are states are states are states are states are states. The states are    | 0.192-5                                 |  |

| Signien                                       | Y. SKFFNA PA     | ARM 1     | 517       | 3.                                      | r            |   | ,             |             |
|---|------------------|-----------|-----------|---|--------------|---|---------------|-------------|
| Hern #  | Y - SKEENA SAY   | Qt.       | Year.     | Make                                    | Model        | Observiction  Report Video reference seal on without CD 20 knots hearts (70) confine all notes seefs hearts, bronch concises (70) change hearts recorded and mostly.  | Seriel Humber | How/Kima    |
|   | ļ                | ļ_        |           | Status Day                              |              | Bench, vise, planer set up stand. (2) 20 kmb heads, (6) profile stands with heads, bench grinder, (5) planer heads monitor and parts  |               |             |
| 372<br>373                                    |                  | =         | _         | Sisten Ross<br>Reform                   | ARI5         | Skis hand profile ginder, Spikting weld station, Jet drill press<br>Traveling hand butte grinder grinding tength 1520 mm  | 2456-82       |             |
| 374<br>373                                    |                  | _         | _         | Cultier Hammer                          |              | 1 ton electric heist, dan, apu, 55° tv, yberte<br>Mac: 10 section 27 section  |               |             |
| 378   |                  |           | _         | Cutter Herryman<br>Maleon               | 50           | Mos 5 section 19 switch Blooms, 50 hp, ghart gate (drawer roof)   |               |             |
| 378   |                  |           |           | Anderson                                | 100          | Blows: 200 to 1780 cms (planes roof) 1.6 dis profess (loop) palendard blow glos (o chip (in) 1.7 dis profess (loop) palendard blow glos (o chip (in) 1.7 dis profess (loop) palendard blow glos (o chip (in) 1.8 dis profess dud colored, (1) dis post gate, bibat doors, Anderson size 6008 150 hp motor, Gardner Denner (IAEH8SA 15   |               |             |
| 380   | L                | L         |           | Torit Dorwidson                         |              | hp blower   |               |             |
| 381   |                  | L         | L         |   |              | Planer parts room, bearings, hardware, sir cylenders, pine, hydraulic poetioning cylinders, roller chaln, shafts, sorter parts, compressor parts, chain holds, bushings   |               |             |
| 383<br>383                                    | 02-01            | E         |           |   |              | Satured 47 transfer duck with Radicon other (re-entry) 4.(4) twin strend fixed unaccambles with drive   |               |             |
| 384<br>384<br>385                             | 02-05            | F         | F         |   |              | 18' 6 strand camelback (people deck, 6' but even ending station, (3) 0' 6, (2), 4' even ending rolls with drive<br>16' 5 strand shoot chair lapspler deck   |               |             |
| 356   |                  |           | Γ         |   |              | Singulating kig loader station, 6 1/2 5 strand lugged transfer throughfeed, 6 arm singulator with artil kick back, 6 strand 16" roller  |               |             |
| 307   |                  |           |           |   |              | deverting transfer, (7) (of over som hold downs.<br>Control room, 55° hr. Bedjor opu, desks, sthere, spere monitors, printers, Buttenklap stepler, hand tods<br>27° 10 stand supped transer broughfried (resets desk (8) 75° even anding rolls, presumatic and positioners  |               |             |
| 389   |                  | Ļ         | 1983      | QM<br>Vanhaus                           |              | Trimmer, 20 pass, 11 saw line shaft drive, approx 25 hp   | 7758-201R77   |             |
| 391   | -                | <u> </u>  | _         | 7 Yes Address of Contractor             |              | VId's  18 8 stand ularma lined bugged Ingrafer deck, crustom made 4 station cut in him spikter system, (7) 10' even ending cots.  |               |             |
| 392   |                  |           |           | Z-Tec Automation Systems Inc            | Wirdell      | Ink jet printing harrode application and vision verification system, their optional and print, Microscotticle, PCID4 real time operating system, Pt. control, 2004 prints, user input locati, pursees, coatern 1-bit, ball deferrey, vision system, all tries.  1-bar sorting system, Scan mag lists detection station, 12 6 streed logged ulmer lend timeler, Neoness 22 bit, 27 bit width, soft   |               |             |
| 350   |                  |           | 1983      | Neuries                                 | 3            | CROOK, (4) SHE OF OUR CONDORS, CARDER PREMIEW 4 SECSION 14 SWECTI MICC, 1U SHOT REPORTS BY THE TEXT SECTION SHOULD VISIT TO THE SECTION OF THE PROPERTY OF THE  |               |             |
|   |                  | Ц.        |           |   | 1 :          | 25 hp hydrausic power pacta, (2) 35 6 strand uhrmw fined scor arresp transfers with Radicon drives, 12 6 strand uhrmw fined incline<br>transfer drive, disconnects  |               | <u> </u>    |
| 394<br>365                                    |                  |           | 1963      | Nernes                                  |              | 5 (4) he's strend byted unscrambler with Radicon drive<br>Stacker system 20, 15 5 strend camelback throughleed transfer, 5 arm stacker Reprord drive, 5 arm lowering hoist, Alien Bradley pic,  |               |             |
|   |                  |           |           |   |              | 3 1/2 Butt even ending station, 5 pocket placer 41 1/2" X 3/4", Hydraufic power pack with (3) 15 hp drivex, push button control, electrics, 22" 4 strend uhmw lined transfer deck   |               | ĺ           |
| 397   |                  | 5         | -         | Yaskawa                                 | Mikrix U1000 | Pit's (stacker), network switch, APC backup<br>32° 5 strand ulmiw fined transfer deck s-drive, 4 arm pop up singulator, Radicon drive, overhead hydrautic package and squeeza   |               |             |
| 396   |                  |           |           |   |              | Station 30 commo, 20 to Implematic power papels<br>32 On roll 80' referent 5 to Respicon others.  |               |             |
| 399<br>400                                    |                  |           |           |   |              | os, return for remains 39 toseway (1975).<br>Af 4 Hand Morn Bred Lysolie deck, Padicon (Area.<br>Strapping system, 25' (8) roll 60' rolliciae, Signade strapping station, top and side squeeze, auto duringe angle placer, M-HT-80 plastic  |               |             |
|   |                  |           |           |   |              | Stepping head, 20 hp hydraulic power pack, puth button control, monitors, camera's, Allen Bradley panelview 500 display, Allen<br>Bradley pic   | ]             |             |
| 401<br>402                                    |                  |           |           | Cutter Hammer                           |              | Mcc 7 section 29 serich   |               |             |
| 403<br>404                                    |                  |           |           |   |              | 30 (11) roll 4" orphity roll-rate 30 (iii) roll 4"   |               |             |
|   |                  |           | <u> </u>  |   |              | (2) Sato CL0085 label printers, Remmable storage calcinet, Acktands N-250 welder, overhead tember wrap disponser, 1/2 ton electric<br>floris.   |               |             |
| 405   |                  |           |           |   |              | Overhead furnises wrap station, 2 ion invelhead hoist, wrap spreader beir, 101 1/2" twin roll wrep post oner with flying cut off blade,<br>hydrautic drives.  |               |             |
| 406<br>407                                    |                  |           |           |   |              | 52' 3 strand fransker deck, Radiçon drive<br>15 ha hydraulic power pack, 10 ha hydraulic power sack, such button control  |               |             |
| 406<br>400                                    |                  | _3        |           | Lenmer                                  | 6552/40.52   | 32 twin strand transfer deck hydraulic drive Paint pumps, self dump hopper  |               |             |
| 410<br>411                                    |                  | 10        |           |   |              | (24) rolls of banding, Sweed scrap chopper Bundles of new stickers  |               |             |
| 412<br>413                                    |                  | 12        |           |   |              | Heading double cut handsaw blades Rolls of plastic banding  |               |             |
| 414   |                  | 97<br>27  |           |   |              | Ournage bundles 44" X 3 1/7 Reils of 54" lamber wrap  |               |             |
| 416   |                  | 16        |           |   |              | Rolls of 101 1/2" tumber wrap (6) concrete lock blocks and steel bin  |               |             |
| 417   |                  | П         |           |   |              | Custom bundle saw system, (2) 12' twin strand transfers in and out feed, 6' L&M bar type saw, laser light, 3 hp hydraulic power pack  |               |             |
| 410   |                  | _         |           |   |              | with hydraufic drings, Aften Bradtey switch<br>(umber vrap station, 101 1/2" hvin roll wrap positioner with Bying cut off blade, hydrautic drives   |               |             |
| 420<br>421                                    |                  |           |           |   |              | Lumber vrop staton, 101 1/2 Non nell stop positioner with frying cut off basis, hydraulic divises. ## (13) of 45° bytes rollishes with drive, 5° 5 shrond canadesis lending sheding shed. ## 5° 5 shrond 5° belt decline barneter decli with drive, 1° 5 shrond grapher decli 5° by Radison drive   |               |             |
| 622   |                  |           |           |   |              | 6 roll 90" even ending rollcase, 3 1/2 bull even ending station, 65 5 strand greenchain transfer deck with drive and side rollofs.  |               |             |
| 423<br>424                                    | 00-00            | - 8       |           |   |              | 20' (5) roll S2" hamber rolloffs<br>S0' X 14" (ott converyor with drive   |               |             |
| 425<br>426                                    | 07-09<br>07-11   |           |           |   |              | 50 bas chijn clean out conveyor with Rudoon drive.<br>50 % 10" vibration someyor, Rens mikit delastos, drive and push butten control<br>Crisper, 65" , 8 brille, Carles trade pystem, idde lead bottom discharge, plw 250 hp drive  |               |             |
| 427<br>428                                    | 07-17            |           |           | CAE                                     |              | 15' box chain clean out conveyor, s-drive with readcon  |               |             |
| 429<br>430                                    | 07-16            | $\exists$ |           |   |              | 75' bas chain clean out common, a drive with Radicon Eaton and Westinghouse perfiches   |               |             |
| 451<br>432                                    | 06-06<br>06-06   |           |           |   |              | 10" Twin auger, Radicsn drive Holmas 8X17 s/n 20089 high pressure 75 hg blower, Approx 24" X 30" 10 hg Neder  |               |             |
| 433<br>434                                    |                  | $\exists$ | $\exists$ | Cuttler Hammer                          | j            | Mcc 11 section 25 switch  |               |             |
| 435   |                  |           | _         | CAE                                     |              | Push baden ceréar, relar sekiral, spessions<br>Chip screen F.X.I.7. deuthir deuth. 7.5 to drive<br>Spers parts. 97 of demanistrat funds, sprodukts  | 16083         |             |
| 437   |                  | _         | =         |   |              | Holmes High pressure 150 he blower, Blancer, Rachir 25° X 30° ESW 5 he feeder.  Main pdc, 1 section 6 switch distribution panel, Skyway 1750 km 25,000 hr/480Y277 transformer, single main switch, Westinghouse   |               |             |
| 439   |                  | _         |           |   |              | 2 section 6 mich tot<br>Pipa 30 lots transformer, Beaner 45 live transformer, (7) Westinghouse 37.5 live transformers, Cuttier Hanner 8 switch pdc, (2)   |               |             |
| 440   |                  | 20        |           |   |              | Place of the designation of the Company of the Comp  |               |             |
| 441   |                  | -14       |           |   |              | Cel transmission, automobive parts, steel bin, man beskel   |               | <b></b>     |
| 442   |                  |           |           | *************************************** |              | 20' shipping container with olit and lube<br>Spare bloom Impallers, sprockets, hose, planer roll, signode strapping head, sentrores, torch sat, Miller SRI-000 webler, hydraulic  |               |             |
| 44  |                  |           |           |   |              | drive, pteel mick, Signade reel<br>Chap sper with grip-try rollopses  |               |             |
| 445   |                  |           | ]         |   |              | Parts wissher, DP gelnder, Visies, Miller SRH-333 welder, littings bins, Rammable cabinet, Malota chop sinv. steet shalls, 14" band sew,<br>chop saw, steet walking (side, hydraufic drives   |               |             |
| 445<br>447                                    |                  | 2         |           | Regel                                   |              | 75 ton educatable hydraulic shoop press, Milings bin with herdware, hydraulic cyfroder, signode parts streopers<br>6,000 L cap hyel lanks with pumpe  |               |             |
| 446   |                  |           |           | Westerd                                 |              | 24.9005, feet bink.<br>Forbild cherry pickers, warnebosses stairs, steet bin, bires with rack.  |               |             |
| 450<br>451                                    |                  | _         | 2003      |   |              | 40 shipping containe<br>Contants, portable gas pumps, lines, hose, atte boxes, remps  |               |             |
| 452   |                  | $\neg$    | $\neg$    |   |              | Torch set, Maller SRH 1333 welder, radio, steel banch, Vise, drill press, steel, hydrautic shop press adjustable, Partier crimper, Snep-On<br>MT 1560 (seter, hyd jack  |               |             |
| - 453<br>654                                  |                  | $\dashv$  | $\dashv$  |   |              | mi coorcentus process.<br>Transference and electrics.<br>Hydraufic house, Devilhica 5 hg pir compressor, hydraufic jacks, like jack, rechargeable gresse guns, jack stands, warehouse stains.   |               |             |
| 453   |                  | 4         | _         |   |              | Stong bins, bench top drift press   |               |             |
|   |                  | _         |           |   |              | Jul 12 band saw, Boos jacks, chop saw, Robing hold firms with elec chain hold, chains, visa, hand tools, parts vrasher. Flammable cabinet, florage chaiper, sides, after high power pack. Ratractable reals Note pumps. Roor jack, obs. sociats, hand lools, new filters, seets, filting bins, hydraulic cylinders, fittings bins.  |               |             |
| 456   |                  |           |           |   |              | chaina  |               |             |
| 457<br>450                                    |                  |           |           |   |              | Metal detoctor, filters, puller parts, shop vec's, file cabbriets, time clock<br>Steel rack with parts, box car, 20 locklist forks, likes, oil tanks  |               |             |
| 450<br>460                                    |                  |           | $\exists$ |   |              | Daewoo buckat, brush guard, keeder arm<br>14 new barrels BAE20SL SAE20SL, Dauton 111, 15W40, SAE 10W olls and lube  |               |             |
| 461   |                  | 2<br>5    | 1997      | New West Industries                     | 4 Kapa       | Side feeders, machanical, 24' 4 lise<br>Sheel log bunks   |               |             |
| 483<br>461                                    |                  | - 5       |           |   |              | 107 X 167 wood bridge decks<br>SPL X 126 1/2° steel bridge keens  |               |             |
| 465<br>468                                    |                  | $\exists$ | 2003      | Slinger Welding                         | NCT-98-1     | 501, X 130' steel bridge frams (13,625 b)  Modular steel bridge dack, (2) 5'W X 50' sections  |               |             |
| 467<br>468                                    |                  | _         |           | ,                                       |              | 30 X, 13" belt conveyor 10 hp Recopy), parts edger's<br>Stanking bunk   |               |             |
| 498<br>470                                    |                  | 7         | =         | Pacific                                 | PSP90114-100 | Josef Charles (1994) 3 section steel dock. (00,000 kg cap, Avery 23405 dro, cap, printer, monitor).  42 box chain convenor.   | \$336-Y       |             |
| 471   |                  |           |           |   |              | SU V-type chain transfer conveyor, Historica 20th drive, spere Michord drive  |               |             |
| 4772<br>4773                                  |                  |           |           |   |              | 17 9 strent correlated transfer deck 37 V-type shein transfer correror, 15 he flourent other  |               |             |
| 474   |                  | 士         | _         |   |              | (2) sturrange sector Nicholson parts debarter, 6 arm. (4) front and rear twin rote, 50 to rung drive  |               |             |
| 478   |                  | 21        | ]         |   |              | Bundler of dumage<br>Boneyard, hydroids power pecks, motors, electrics, comerce rate, stilespect<br>CD X 14° bet comprov with lickers, spare rature sections  |               |             |
|   | - $           -$ |           |           |   |              | 607 X 14" bell corruptor with tickers, spere return sections. Corruptor troughed rolls, conveyor sections, (coder, landing deck, control booth, blower tims, reficese, reducers, self-dump hopper.  |               |             |
| 478   |                  | - 1       |           |   |              | Sjeel log bunk, old diesel besk up fire pump, transfer renisch, transformer and electrics   |               | <b></b>     |
| 478<br>479                                    |                  | _         |           |   |              |   |               | <del></del> |
| 478   |                  |           |           |   |              | Main plant poerar ingrassion fine, power poles, stitches, fence enclosure   |               |             |
| 478<br>479<br>480<br>481<br>482<br>483        |                  |           |           |   |              | Major plant power laposprisolom (spe., power poles, stätches, fence enclosure) Tech calle (n. savenille prof plane) mill. Yard camera, lecunity cappine gratem, monitors, (popoder  Vard camera, lecunity cappine, gratem, monitors, (popoder   |               |             |
| 478<br>479<br>480<br>481<br>482<br>483        |                  |           |           |   |              | Main date (power Ungungspalan has, power priess, pickhos, livrous proclasors. Tech codes) is service and progress and Tech codes) is service and progress and Tech codes) is service (power profiless, procedure). Tech common security (power profiless, procedure). Tech common security (power profiless, procedure). Tech codes (power profiless, procedure). Tech codes (power profiless, procedure). Tech codes (power profiless). T  |               |             |
| 478<br>479<br>480<br>481<br>483<br>484<br>485 |                  |           |           |   |              | High older (power Ungungstein in the, power pietes, (EU-bes, hance procleours<br>Tech called in sent and pay (place mod ). The called |               |             |
| 478<br>479<br>480<br>481<br>482<br>483        |                  |           |           |   |              | Main date (power Ungungspalan has, power priess, pickhos, livrous proclasors. Tech codes) is service and progress and Tech codes) is service and progress and Tech codes) is service (power profiless, procedure). Tech common security (power profiless, procedure). Tech common security (power profiless, procedure). Tech codes (power profiless, procedure). Tech codes (power profiless, procedure). Tech codes (power profiless). T  |               |             |

| SUMMAR | Y - SKEENA BA | WAN I | SLT  | )         | 1           |  | ·                  |             |
|--------|---------------|-------|------|-----------|-------------|--|--------------------|-------------|
| liam # |               |       |      | Make      | Model       | Description  |                    |             |
| 470    |               | +     | -    |           | - CENTRAL   | Main plant offices, destru, chairs, cours, printers, storage cabinets, file cabinets, meeting scorn tables, fiel screen tys, paper styledders, | Secial Number      | Healtima    |
|        |               | 1     |      |           | 1           | Dall server, with rack switches and APC backup ups, copiers, boardroom, funch room, label printers, radio's binders, taminater.                | 1                  | 1 !         |
| 1      |               | 1     |      | Į.        |             | cleaning supplies, storage shelves, snowshoes, marking hammers, helmets, shelloway supplies, raco's bunders, terranger,                        | i                  | ( !         |
| 460    |               | -     |      |           |             |  |                    |             |
| 491    | 26            | 1     |      | Geria     | GS1930      | 40' shipping containers with wood inuss, metal clad roof structure<br>Scissor LR, electric   |                    | +           |
| 492    |               | +-    |      | Sudence   |             |  | 6\$30134-12006S    | IVA.        |
| 493    |               | +     | 4010 | Subertra  |             | Portable sea container loader, super type, 201-40 container cap, 4 wheel, electric, order # \$0602(03, job # 345105                            |                    |             |
| 494    |               | ┿     | -    | 3.G       | <del></del> | Tandern aufe convertor delly, 275/80R22 5 lives  | ZA9C62224/D138244  |             |
| 495    | 2116          | -     | 2005 |           | 500V        | Markft, dual fuel, 2 stage boom,   | 300090379          | 4.647.741   |
| 496    |               |       |      |           |             |  | H2X384S03906       | no meter    |
|        |               |       | 2002 | Dodge Ram |             |  | 3B7KF26Z72M289015  | INA         |
| 497    | 946           | -     | N/A  | Oodge Ram |             |  | NA                 | IVA         |
| 498    |               |       |      | Sid-Doa   |             | Snowmobile, engine 600ACE  | 28PSKCJA7JV000717  | 462 Kms     |
| 499    |               | -     |      | Artic Cat |             |  | 4LF05ATVX5T250062  | INVA        |
| 500    |               | -     |      | Danwoo    | G25E-3      | Forld ff, 5,0001b cap, 3 stage mast, cushion tire, side shift, bg  | G900404            | NYA         |
| 501    |               | -     | 1996 |           | F250 Xt     |  | 2FTHF28FGTCA49104  | NVA         |
| 502    | 910           |       |      | Ford      |             |  | NA                 | 407,000 kms |
| 500    | 954           |       |      | Dodge Ram |             | Tradesman, Crew cab, automatic transmission, 4X4, headache rack  | IC6RRFNG6: N100947 | 42,044 kms  |
| 504    | 950           | 1     |      | Dodge Ram | 2500 HD     | Pick up truck, crew cab, Vortec, automatic transmission, Hemi 5.7L, 4X4, side steps, headache rack   | 1C6RR7LT6K\$505647 | 33,723 bms  |
| 505    |               |       | 2004 |           | F2SO XLT    | Pickup truck, crew cab, automatic transmission, flat deck, yard truck  | IFTNW21L14ED43T26  | N/A         |
| 506    | 941           | L     | 1996 |           | F250        | Pickup truck, reg cab, automatic transmission, yard truck  | 2FTHF26HITCA49103  | N/A         |
| 507    | 845           | 1     | 1996 | Ford      | F250        |  | 1FTJW36H8TEB47630  | N/A         |
|        |               |       |      |           |             | portable welder 400 hrs, torch set yard truck  |                    | 4*** 1      |

| April  |  |                |          |  |  |               |   |                      |            |
|--|--|----------------|----------|--|--|---------------|---|----------------------|------------|
| The content of the    | Distr. C   | Unit #         | 017      | Year   | Maxe   | Model         | Description   | Serial Number        | Hes/Kine   |
| Company  | 1 2  |                | 70       | 2016   | 3  | 1             | Concrete lock blocks Twin steel hopper lead system approx 60 cubic meter, open too feed, 4 cylinder floor moving  |                      |            |
| 1  |  |                | 1        |  |  |               | system, (2) 30 hp 1775 rpm variable speed hydraulic powerpacks, twin overhead bin rollers with  |                      |            |
| 1  |  |                | Η,       |  |  |               | Disconnects in area, rolling stairs   |                      |            |
| 1  | 5  |                | <u> </u> | 2018   | Bruks  | CS1000X18 25M | 65 X.3 brits strand byted incluse conveyor, mid bypess drop out, Sew 7.5 bp drive   | 1821010              |            |
| March   Marc   | 7  |                | <u> </u> | 2016   | 3(   | KS1060X100    | 10' suger conveyor, 3 he shellmount drive   | 195 1030             |            |
| 1  |  |                | L        | 1  |  |               | 6 X 81 1/2" w vibratory screen twin 2.5 hp drives, 6 magnetic roll druin 1.5 hp, Twin Comeln 5.5 hp<br>mod. KHLE30-400-5H-4-2-R180 cleaner lars and disconnects.  |                      |            |
| 10   10   10   10   10   10   10   10  | •  |                |          | 2016   | Bruks  | H##800X 2000  | Hammer mill, 70" X 51" top feed bottom discharge plw livin 275 hp drives side opening, hydrautic coverneck. Riemans main discooped. Rout a D switches   | 1821030              |            |
| 1.   |  | 220(MD)        |          |  |  |               | 16" X 25" steel support structure with calvalising and steins   |                      |            |
| 1.   | 12   |                |          | 2015   | Bruks  | CS1000X15,644 | 52' Twin strand bried drag brow incline converse, covered, 5.5 to since   |                      |            |
| 10   10   10   10   10   10   10   10  |  |                | L        |  | .L   |               | lander describede couverior 3 pb grive  | L                    |            |
| 1  | 14   |                | ┼        | 2018   | Bruks  | C\$500X11.3M  | 40 fivied drag type conveyor, any losed with drive.  Nammer mill dust collection system accords 14,000 cfm, with 6 die cyclone on stand, bottom relay.  | 182070               |            |
| 1  | i l  |                | 1        |  |  | 1             | air lock feeder, 50° dis 40 hp main id fan, spark detection sensor, Square D disconnects  |                      |            |
| March   Marc   |  | 2230 1822      | 13       |  |  |               |   |                      |            |
| ## 150   | 18   | 32204403       |          |  |  |               | 307 Auger Incline conveyor, 2 hp. bottom drop out   |                      |            |
| ## Company of the Com |  |                | 1        | 2018   | Swisa Combi  |               | Low temperature bell drying system, 171 m2 (1,840 Pt) self supporting, 100° L X 20° W, Elytra   |                      |            |
| 1  |  |                | Ì        |  |  |               | diverting gate top infet module double screw leading system, inspection and cleaning doors, polyester belt with alignment, belt cleaning system, brush roller, (73) bed support rolls, (3) Elektron   |                      |            |
| March   Marc   | 1 1  |                |          |  | ľ  |               | main fars 85" dia mod CFL1 1400-00-LGO-315M volume flow 131,303 p/w Weg 100hp 1192 rpm  |                      |            |
| 1.   | 1  |                | 1        |  |  |               | chember and the educat stacks, fire detection and extinouishing. Finns continuous moisture  |                      |            |
| 1.   |  |                |          | l  | ]  |               | [burners, ignition systemitams safety burners are air admission with planum chambers, secons layer  |                      |            |
| 20   20   20   20   20   20   20   20  |  |                |          | <u>L</u>   |  |               | 1   |                      |            |
| 1  | 22   | 3520-M01       |          |  |  |               | 25' Incline auger with 5 hp drive   |                      |            |
| 1.   | 23   |                | 40       | $\vdash$   |  |               | Efectives in area Lincoln presentatic tube pump with tube, hose reets  Concrete lock blocks with (7) steel pipe stands  |                      |            |
| Company  |  |                | F        | 2018   | Hambison<br>Incersol Rand  |               |   |                      | × m-       |
| 1985      | 27   |                | <u> </u> | 2018   | Incorport Pand   | R/.5-1145     | Rotary screw air compressor, 10 hp 90 psig  |                      | 22.874 hrs |
| 20   200     | 29   |                |          | L  | İ  |               | I 1500 kp hydrautic 48 cart, steel screen cart, 450t, cap bdy tank with electric pump, flammable  |                      |            |
| 10   10   10   10   10   10   10   10  | 30   | 37004401       |          | 2018   |  |               | storage cabinet, self dump hopper, banding cart, electrics 20 exper conveyor with Sew 3 ho drine, bottom drop out with date   |                      |            |
| 10   10   10   10   10   10   10   10  | 31   | 40204/01       |          | 2016   |  |               | 50' terin strand Byted ding type incline conveyor 7 5 hp, distribution shule, stone seperator, mannetic plats   |                      |            |
| Section 2015   1975     | 32   |                |          | 2018   | Promis   | Br4A 150      | Harrymer mill 63" X 45", p/w twin 400 hp 1450 rpm drivers, top feed bottom discharge, stainless   | 15930                |            |
| 1  |  |                | l i      | l  |  | 1             | suppression stations, 7 dia cyclone bin master bindicator, ACS bottom feeder v/n 35005 with Fibe  |                      |            |
| 15   15   15   15   15   15   15   15  | $\perp$  |                | <u> </u> | <u> </u>   |  |               |   |                      |            |
| 20   | 33   |                |          | 7018   | 1  | }             | If one day room cyclone with stand, Solids bottom rotary air lock feeder, surroventiatori type APRI<br>1251 LGO ES4 fan 150 hp 1590 rpm, temperature sensors, Square D disconnect   |                      |            |
| 1995      | 34   |                | -        | 2010   |  |               | L   |                      |            |
| 127716   7.1 by the price of may 1 for a global many store of many store of may 1 for a global many store of many 1 for a global many store o   | - 45   |                | -        | 2018   |  |               | 60HZ LGO E64 (an 120 ho 1620 rpm, Square D disconnect   |                      |            |
| 1807-07-10   1908   1   | ~  |                | i        | ""   |  |               | 12774185 7.5 hp beghouse mounted on a 21' dia cyclone, Baum 20X20 shr 8351 rotary airlock   |                      |            |
| April   Description   1995     | 1 1  |                |          |  |  |               | 14092-01 50,000 cfm main fan prie 200 hp 1785 rpm drive, secondary Claricus mod CP-41.42"   | ļ                    |            |
| Section   1985   |  |                |          | L_   | İ.   |               | abort gate s/n 037-010178-001, 40" abort gate s/n 1095003, 4" blow pipe, 2 hp Yurhit vacuum<br>pump, peessure transmitters, disconnects and electrics   |                      |            |
| 1985   1986      | 36   | 6BE-0R8-001    |          | 2018   |  |               | Buffer allo 15' X 11' dia bolt logether, 40m cap, bottom rotary sweep priv Bondofioli drive, 20'  |                      |            |
| 2016      |  |                |          |  | 0-4-   |               | 22' suger conveyor, 2 trp shuft mount   |                      |            |
| 4  | 30   | 50407401       |          |  | anas .   | CS000A15.0W   | [37 DUCKET EFENTION, 10 top drive, hitse explosion wents, Solids top diversing gate station with shutes,  | 1821085              |            |
| Company   Comp   |  |                | _        | 2018   | 8n/s   | CS800X17.8M   |   | 18210                |            |
| A  | 43   |                |          |  |  |               | 3,000 Kva pad Inunsformer   |                      |            |
| 1  | 42   |                | 2        |  | Lervick  | KGA180S4MS2G  |   | SRIGOTHINA SALECTION |            |
| descriptions with APP dates, (f) Private conclaiming any PEP 400, (f) are paid as his private for the private  |  |                | 2        |  |  |               | platform  |                      |            |
| Signature of the control of the cont | 43   |                |          | 2018   | Cumrere  |               | platiform Back up generator, natural gas, 45 tws, 45 km, 1207240v, transfer writch 50 frein stand fivied date chain conveyor, 7.5 hp. patiet mill on feeder   |                      |            |
| 60   1, 2014   1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,  | 43   |                |          | 2018   | Cumrere  |               | platherm Back up penderlor, natural pas, 45 km, 45 km, 120740v, transfer errich.  50 hein skand fried dies; chain commerc, 7,5 kp, petet mit ich leeder. Pater mits sech with (3) drop shutes with morotor probes, (3) 6 1/2 1 kp patet mit missers, (3) learn with (4) drop shutes with morotor probes, (3) 6 1/2 1 kp patet mit missers, (3) learn with APS decira with APS devel, (3) Promit public with section with APS decira with APS devel, (3) Promit public mits.   |                      |            |
| SQUADD   2018   Command    | 43   |                |          | 2018   | Cumrere  |               | jelebrin<br>Beit vip generation, natural gas. 45 tw. 45 tvs. 120740v, transfer ereich.<br>50 tvin jural (held dags chain commerc. 15 tp. gelet all 8 ch leeder.<br>Pellat milit sech with (3) dags chain with monote probes, (6) 1571 tap gelet milit missen, (3)<br>lotary articot leeden with APP delves, (3) from a contitionen type EPP 480, (3) Promit poellet milit<br>hype E8 (46 SER, 45 top) 1500 gm and 180771, (1874, 1875, 5), (6) beta pomp. (3) betas of (5) betas of (5)   |                      |            |
| Counted the part and counter spate (color spates), 115 Y 115 Y 190 year sent service, 167 3.1 Bit sele partium, selection  | 45<br>45<br>46   |                | 3        | 2018<br>2018<br>2018   | Comming Promit Sinfress  |               | jeldebm. St. (1) general (I. robust) (20. 4.5) v (3) v (3) v (3) v (3) v (3) v (4) v  | BJ80316197           |            |
| Section   Comment   Comm   | 45<br>44<br>45<br>45   | 5550-H01       | ,        | 2018<br>2018<br>2018<br>2018   | Comming Promit Sinfress  |               | jointhm. Sex 1,00 primarial contract pass, 45 to 45 to 1,00 250; inpurise series.  Sex 1,00 primarial contract pass, 45 to 45 to 1,00 250; inpurise series.  Sex 1,00 primarial pass, 45 to 1,00 primarial pass, 45 to 1,00 primarial misses, (2) primarial misses, (3) primarial pass, (4) primarial pass, (4) primarial pass, (4) primarial pass, (5) primarial pass, (6) primarial pass, (6) primarial pass, (7) primarial pass, (8) pr  | BJ80316197           |            |
| Section (Control of Control of    | 45<br>44<br>45<br>45   | 5550+O1        | ,        | 2018<br>2018<br>2018<br>2018<br>2018<br>2016<br>2016   | Company Promit Sinfrests Sydness   |               | platebom  Ber Ling general or, natural post, 45 tov. 45 tov. 1902 50v. Introduct particit.  25 to this justing 5 fired does climic incorrect. 25 top, politic all distribution.  25 to this justing 5 fired does climic incorrect. 25 top, politic all distribution.  15 to politic all distributions of the climic incorrect and  | BJ80316197           |            |
| Earl SP/ADDO SP Internation (1)   Earl SP/ADDO SP Internation (1)   2018   Square O described by Strychemic (1)    | 45<br>45<br>45<br>46<br>47<br>49<br>49   | 5550+01        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Comment Promit Sinfress Sinfress Sinfress  |               | plathom Sex up general or, natural gos. 45 tov. 45 tov. 1972 50v. Introduct particls.  25 to by layer 5 freed does other commerc. 15 to, policy in all of based on the commerce. 15 to go to the city of the commerce of the c  | BJ80316197           |            |
|  | 45<br>45<br>45<br>46<br>47<br>49<br>49   | 5550+01        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Comment Promit Sinfress Sinfress Sinfress  |               | plathom Sex up general or, natural pass. 45 tov. 45 tov. 1707.50v. Introduct particit.  25 to by played 5 first disea chair commerc. 15 to, polific it ill objected.  26 to by played 5 first disea chair commerc. 15 to, polific it ill objected.  26 to by played 5 to by 1500 per sex to be played 15 to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to by 1500 per sex to be 1500 per sex t  | BJ80316197           |            |
| Solid Compared Confidence of Compared Control (1) by degree of present project | 49<br>44<br>45<br>45<br>47<br>49<br>49<br>50<br>51   | 5550401        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Comment Promit Sinfress Sinfress Sinfress  |               | jedebun (2014) general (in coloral (in al. Str. 45 hr. 1907-05), lyrocks serich. In Str. 1909-064 (in coloral (in al. Str. 45 hr. 1907-05), lyrocks series. In Str. 1909-064 (in al. Str. 1909-064), lyrocks series. In physical series. In Str. 1909-064 (in al. Str. 1909-064), lyrocks series.   | BJ80316197           |            |
| Countries of 450 percent (in pressure place in 150 percent) (in the  49<br>44<br>45<br>45<br>47<br>49<br>49<br>50<br>50<br>50   | 5550401        | ,        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Conners Pronil Sinfrara Sinfrara General General   |               | plathom  Sex vg perserial or, natural post, 45 tov. 45 tov. 1707 50v. Inprofest periols.  Six holy buyers of herd does other borrows - 15 top, politic all do historia.  The history of herd does other borrows - 15 top, politic all do historia.  The history of historia of  | BJ80316197           |            |
| Countries of 450 percent (in pressure place in 150 percent) (in the  49<br>44<br>45<br>45<br>47<br>49<br>49<br>50<br>50<br>50   | 5550AD1        | ,        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Conners Pronil Sinfrara Sinfrara General General   |               | plathom Sex to generation, natural oper, 45 tov. 45 tov. 1707.50v. Introduct particip.  32 today invariable does define converse. 15 to, order it if it is beside Head in this work with Coll grap between intensity product, 60 pt.  33 today invariable (Coll grap between invariable product, 60 pt.  40 today in the converse of the collection of the collect  | BJ80316197           |            |
| Cases in the hand total, common one problem, Yenner are exchange, freshy screen, Soric at local auditories, metal  | 45<br>45<br>45<br>46<br>47<br>49<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50   | 5550401        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Conners Pronil Sinfrara Sinfrara General General   |               | plathom  Sex typ general or, natural ops. 46 tov. 45 tov. 1972 50v. Inprofest particle.  Six holy justed Shed does clear to convert. 25 top, offer all 6 histories.  The history of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex of (6) the pumps, (3) sexts of (6) they sex typ of the sex  | BJ80316197           |            |
| Auto-  | 50<br>45<br>47<br>49<br>49<br>40<br>40<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50<br>50   | 5550401        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Conners Pronil Sinfrara Sinfrara General General   |               | plathom  Sex typ general or, natural ops. 46 tov. 45 tov. 1972 50v. Inprofest particle.  Six holy justed Shed does clear to convert. 25 top, offer all 6 histories.  The history of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex typ of the sex of (6) the pumps, (3) sexts of (6) they sex typ of the sex  | BJ80316197           |            |
| Acc   Section (18) vid.s.   Services (18) v   | 52<br>45<br>45<br>47<br>47<br>49<br>50<br>51<br>51<br>52<br>53<br>53<br>55<br>57   | 5550±01        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Conners Pronil Sinfrara Sinfrara General General   |               | Josephin Description of the Community of  | BJ80316197           |            |
| 1  | 52<br>45<br>45<br>47<br>49<br>47<br>49<br>49<br>50<br>51<br>52<br>53<br>53<br>55<br>57<br>56   | \$\$\$\$A01    | 3        | 2018<br>2018<br>2018<br>2016<br>2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018         | Oznaria Pronii Sidrasia Sidrasia Sidrasia Sidrasia Sidrasia Aden Sidrasia Security Security  |               | plathorn  Sex v. op general or, outward ops. 46 tov. 45 tov. 1707 50v. Inprofest particle.  Sex v. v. op general or, outward ops. 46 tov. 45 tov. 1707 50v. Inprofest particle.  Six holds being of the sex others converse. 15 top. opfice all of histories.  The sex v. op general or outward of the sex others converse and the sex others are designed on the sex of 60 towns and the sex of 60 towns and 60 towns a  | BJ80316197           |            |
| 100    | 52<br>45<br>45<br>47<br>49<br>47<br>49<br>49<br>50<br>51<br>52<br>53<br>53<br>55<br>57<br>56   | 5550451        | 3        | 2018<br>2018<br>2018<br>2016<br>2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018         | Oznaria Pronii Sidrasia Sidrasia Sidrasia Sidrasia Sidrasia Aden Sidrasia Security Security  |               | Josephson (1994) Annie of the Common (1994) Anni  | BJ80316197           |            |
| Section 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1  | 52<br>54<br>54<br>54<br>54<br>54<br>54<br>50<br>51<br>52<br>53<br>53<br>54<br>56<br>57<br>59<br>60<br>60   | 5552401        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commina Promit Sinfrage Sofrage Sofrage Abin Sinfrage Secure D Prodesa   |               | platebom  Sex vig perioration, natural pass, 46 tov. 46 tov. 1972 50v. Inprofest particle.  25 In this lives of heric dives clear in convenior. 15 (a), policy in all in historia.  26 In this lives of heric dives clear in convenior. 15 (a), policy in all in historia.  16 In this lives of heric dives clear in convenior. 15 (a), policy in all in historia.  16 In this lives of historia in all in  | BJ80316197           |            |
| Section 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1  | \$2<br>\$4<br>\$4<br>\$5<br>\$2<br>\$2<br>\$2<br>\$2<br>\$2<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3<br>\$3         | \$\$\$\$A\$\$1 | 3        | 2016<br>2018<br>2018<br>2018<br>2016<br>2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018 | Oznaria Pronii Sidness Sidness Sidness Sidness Sidness Sidness Sidness Sidness Desire Sidness Desire Sidness Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire Desire   | C-5 1981      | jointhum. Sex vig general or, natural ops. 46 tov. 45 tov. 1907.650; Izanoles perich.  25 Indis Justino S held desc claim socrates. 2.5 top. 466 tol. 80 in helder.  26 Indis Justino S held desc claim socrates. 2.5 top. 466 tol. 80 in helder.  27 Indis Justino S held desc claim socrates. 2.5 top. 466 tol. 80 in helder.  28 Indis Justino S held desc claim socrates. 2.5 top. 466 tol. 80 in helder.  38 Indis Justino S held desc claim socrates. 40 X to inches period 9.00 helder period period 9.00 helder period 9.00 helder period 9.00 helder period 9.00 helder period 9.00 helder period 9.00 helder period 9.00 helder period 9.00 helder 9.00 helder period 9.00 helder period 9.00 helder 9.0  | BJ80316197           |            |
| Supplementary   Supplementar   | \$3 44 45 45 47 49 49 49 49 49 49 49 49 49 49 49 49 49   | 5550.491       | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-5 1981      | plathom  Sex up general or, natural cost. 46 tov. 46 tov. 1072 501, Ispanish serich.  Sit half bird of held disea clear to convert. 25 top, order all of histories.  Sit half bird of held disea clear to convert. 25 top, order all of histories.  Only an inches described with APP divises, (1) Provide continous high per PEP 400, (1) Provide point in the hype 6 till of SSR. 450 to 1000 pm in 10077, (1073, 1075, (1) bits pumper, (3) barks of (5) Argonization and the secondary of the   | BJ80316197           |            |
| Section   Sect   | 53<br>444<br>45<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47<br>47  | 5550401        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-5 1981      | jointhum. Sex. 1, 19 provinced or, colorization, and Sex. 1, 19 provinced or, colorization, colorization, and colorization of the colorization of   | BJ80316197           |            |
| Pirmite strayer   Content with brother, Incritroom, (7) offices with chairs, list acream monitors, chairs, like cabinets, fortunes, law rates, fridge, IV, bookstees, board room bolds, organis meters with chairs, like acreaming from the cabinets of the    | 43<br>443<br>45<br>46<br>47<br>47<br>49<br>50<br>50<br>51<br>53<br>53<br>55<br>56<br>57<br>56<br>68<br>68<br>67  | 5550±01        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-5 1981      | Joseph Times (1997) and the control of the control   | BJ80316197           |            |
| Contract, Instruction of Stage, IV, Societies, Burding on Isabel, organ matern with charges, meeting (1994)   Septiminary continues   Contract,    | 43<br>443<br>45<br>46<br>47<br>47<br>49<br>50<br>50<br>51<br>53<br>53<br>55<br>56<br>57<br>56<br>68<br>68<br>67  | 550401         | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-5 1981      | jedebum (2014) provided (in cubural Sec. 15) to - 50 to - 1007/05, jurnale serbit.   Sec. 10 general services are consistent of the consis  | BJ80316197           |            |
| 10   100     | \$43<br>443<br>443<br>443<br>443<br>449<br>449<br>500<br>501<br>502<br>503<br>504<br>504<br>505<br>505<br>506<br>506<br>506<br>506<br>506<br>506<br>506<br>506 | 5550401        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-5 1981      | jointhum (See 1) general siz, voluted loss, 45 to - 45 to - 15 to 2000, ignosts series. In Sec. 10 general siz, voluted loss, 10 to   | BJ80316197           |            |
| Content of preference   Cont   | \$43<br>443<br>443<br>443<br>443<br>449<br>449<br>500<br>501<br>502<br>503<br>504<br>504<br>505<br>505<br>506<br>506<br>506<br>506<br>506<br>506<br>506<br>506 | 5550401        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-S DRI       | Josephson 1992, Andread Sci., Andread Sci., 15 Nov. 55 Nov. 1972 (Sci. Ipprofest periols).  Best 1, 19 perioration, Andread Sci., 15 Nov. 55 Nov. 1972 (Sci. Ipprofest periols).  Best 1, 19 perioration, Andread Sci., 15 Nov. 55 Nov. 1972 (Sci. Ipprofest periols).  Best 1, 19 perioration, 1972 (Sci. Ipprofest Continues Nov. 1972 (Sci. Ipprofest Perioration Nov. 1972 (Sci. I  | BJ80316197           |            |
| Conferent such capitals, when all excepts applied of good, sheet pack  | 433<br>445<br>445<br>445<br>447<br>449<br>449<br>449<br>550<br>551<br>552<br>553<br>559<br>550<br>550<br>550<br>550<br>550<br>550<br>550<br>550<br>550         | \$50.101       | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-S DRI       | jointhm. The Set 1, 19 primarile of valuation past 4,5 to 4,5 to 4,5 to 1,10 primarile particle.  Set 1,19 primarile of valuation promoter. J. Solo, 19 of the III bit hadded.  Set 1,19 primarile of valuation promoter. J. Solo, 19 of the III bit hadded.  Set 1,19 primarile of valuation promoter. J. Solo, 19 of the III bit hadded.  John Set 1,19 primarile of valuation promoters. J. Solo, 19 of the III bit hadded.  John Set 1,19 primarile of valuation promoters. J. Solo, 19 of the III bit hadded.  John Set 1,19 primarile of valuation promoters. J. Solo, 19 of the III bit hadded.  J. Solo bit hadded primarile of the III bit hadded of valuation promoters. J. Solo, 19 of the III bit hadded of the III   | BJ80316197           |            |
| Cell   NSU-5022   Quark later, bedde forts   All (1900-2001)   A   | \$43 445 445 445 445 445 445 445 445 445 4   | 5550.491       | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-S DRI       | jedebom. Sex 1.9 general or. outstand loss. 45 to 45 tive 1.107.650; Izprofest earlie h.  Sex 1.10 general of control of the c  | BJ80316197           |            |
| Palled meltings and mice along container, 3 section 2 like palled and the with pealed and sparse chain, parting, lapped.    201  | \$3 44 45 45 447 45 45 45 45 45 45 45 45 45 45 45 45 45  | 5550H01        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Commercia Promit  Sinfrases Sinfrase | C-S DRI       | jedebom (2014) production chand past 45 for 45 for 1907-501 (product earlier).  Set 1,0 general or, chand past 45 for 45 for 1907-501 (product earlier).  Settle and better for the change of the chan  | BJ80316197           |            |
| Parting Speed   Parting Conference   Parting Conf   | \$3 443 453 4453 4453 4453 4453 4453 4453  | 5550401        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Connets Pronfil Sinfress Sinfr | CS DEL        | jointhm. The Set 19 general or, outstall past, 45 to 45 tives 1907-26/v. Ipprofest earlieft.  Set 19 general or, outstall past, 45 to 45 tives 1907-26/v. Ipprofest earlieft.  Set 19 general or, outstall past, 45 to 45 tives 1907-26/v. Ipprofest earlieft.  Set 19 general or 1907-26/v. Ipprofest earlieft.  Set 19 general or 1907-26/v. Ipprofest earlieft.  Set 1907-26/v. Ipprofest earlieft.  Set 29 general or 1907-26/v. Ipprofest earlieft.  Set   | 9,600/4/8/7          |            |
| Conference weed week intermediate professional and agrain by dainy drive, hose reveals, bits, accurate   Conference week agrain, metal clad, both leighter, included spirits, spirits, spirits, control of cont   | \$3 443 453 4453 4453 4453 4453 4453 4453  | 5552.101       | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Connets Pronfil Sinfress Sinfr | CS DEL        | jobelbom  Ext. tog profestion, outstand past, 45 tov. 45 tov. 1907_650, ispanight perich.  Set him jobelbom die des others commerce. 1,5 top, 666 all 8th histories.  Set him jobelbom die des others commerce. 1,5 top, 666 all 8th histories.  For him jobelbom die des others commerce. 1,5 top, 666 all 8th histories.  For him jobelbom die des others commerce. 1,5 top, 666 all 8th histories.  For him jobelbom die des others with APB dieter, 1,9 formed continues they per 64 all 9, 69 formed peaks make type 6 foll 65 555, 450 for 1500 group with 1907, 1907, 1807, 1807, 1908, per 64 all 9, 69 formed peaks make type 6 foll 65 555, 450 for 1500 group with 1907, 1907, 1807, 1807, 1807, 1908, per 64 all 9, 69 formed peaks make type 6 foll 65 555, 450 for 1500 group with 1907, 1807,  | 9,600/4/8/7          |            |
| 5 1 2010 Soors, man doors, 22 dishes detection. 5 2 20 Indicated in profession. 5 3 20 Indicated in profession. 5 3 20 Indicated in profession. 5 3 20 Indicated in profession. 5 4 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 6 Indicated in profession. 5 7 7 7 7 Indicated in profession. 5 7 7 7 7 7 Indicated in profession. 5 5 Indicated in profession. 5 5 Indicated in profession. 5 6 Indicated in profession. 5 7 7 7 7 Indicated in profession. 5 7 7 7 7 7 Indicated in profession. 5 7 7 7 7 7 7 7 1000 bioding. Clase apart, material call, total Ingelture construction, Indicated, form door, Indicated, Indicated In | 90<br>443<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>55<br>53<br>53<br>54<br>55<br>56<br>56<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66      | 5550401        | 3        | 2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebom. Sex. 1.0 general St., cubural Sex. 45 by — \$10°CAD. Jurnals sextic. 1.0 product sex. cubural Sex. 1.0 general Sex. 1.0 sex. 1.0 general Sex. 1.0 gener | 9,600/4/8/7          |            |
| 61 AV18   Test paskin in policy dark  52   23   Convertie built blocks  53   33 Seed brokklighter service allow the section peaks rack, she'ts, sings, chairs, pipe abands,  54   Pleate plant marker area allow allow the section peaks rack, she'ts, sings, chairs, pipe abands,  55   Pleate plant marker area only seed on the section peaks rack, she'ts, sings, chairs, pipe abands,  56   Verification of the section section section peaks rack, sings, chairs, pipe abands,  56   Verification of the section section section peaks rack, she'ts, sings, chairs, pipe abands,  56   Verification of the section section section section peaks rack, she'ts, sings, chairs, pipe abands,  57   Verification of the section section section section peaks rack, she'ts, sings, chairs, pipe abands,  58   Verification of the section section section section peaks rack, she'ts, sings, chairs, pipe abands,  59   Verification of the section section section section peaks rack, she'ts, sings, chairs, pipe abands,  50   Verification section section section section peaks rack, she'ts, sings, chairs, pipe abands,  50   Verification section section section section peaks rack, she'ts, sings, chairs, pipe abands,  50   Verification section section section section peaks rack, she'ts, sings, chairs, pipe abands,  51   Verification section section section section peaks rack, she'ts, sings, chairs, pipe abands,  52   Verification section  | 90<br>443<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>55<br>53<br>53<br>54<br>55<br>56<br>56<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66<br>66      | 5550401        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum (2014) profestion, outland lost, 45 Nov. 45 Nov. 1907,450. Ipprofest earliet.  Seek 1, 19 periorization, outland lost, 45 Nov. 45 Nov. 1907,450. Ipprofest earliet.  Seek 1, 19 periorization, outland lost, 45 Nov. 45 Nov. 1907,450. Ipprofest earliet.  Seek 1, 19 periorization, 10 periorization, 1907, 1907,150. Ipprofestion and microsome, (2) outland lost earliet.  Seek 1, 1907, 19  | 9,600/4/8/7          |            |
| 53   \$3.586 jubilityini sirunga cappy   | \$1 44 47 47 49 49 49 49 49 49 49 49 49 49 49 49 49  | 550.101        | 3        | 2016<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018<br>2018   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum  Edit 1.0 geninated in cubural Sec. 15. br. 4. St. vm. 1907-05. jurnala serich  Edit 1.0 geninated in cubural Sec. 15. br. 4. St. vm. 1907-05. jurnala serich  Edit 1.0 geninated in cubural Sec. 15. br. 4. St. vm. 1907-05. jurnala serich  Fadia critia serich verical serich series serich serich serich serich serich series serich series serich series serich series s  | 9,600/4/8/7          |            |
| Pallet (pilot markinence shop with Lacking pallet pilot price shorts, pilot pi | 50 447 497 500 500 500 500 500 500 500 500 500 50  | 550401         |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum (2014) policied (in. cubical Sec. 45) by - 45 har 1907-20, jurisdit septic.)  Ext. 10 general (in. cubical Sec. 45) by - 45 har 1907-20, jurisdit septic.)  False of this such with (1) drop to because (1) and policy (1) and   | 9,600/4/8/7          |            |
| 65 (Cpu table 66 (47 X 37 tho phabling), char span, milet club, basics, insulated, how door, insulated, him door, insulated, how door,  50 44 55 55 55 55 55 55 55 55 55 55 55 55  | 5550401        |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum (2014) production, undered also, 16 to 45 for 1907-1907-1909-1909-1909-1909-1909-1909-  | 9,600/4/8/7          |            |
| Cost bable  47 X XI whop balding, clear apan, metal clad, ball together construction, insulated, twin cloor, stripe lawy door stripe lawy door   | 50 44 55 55 55 55 55 55 55 55 55 55 55 55  | SSO.101        |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 1.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 1.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 4.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at the series of the series o  | 9,600/4/8/7          |            |
| 65 47 X 30 Who building , chair span, metal clad, boll together construction, insusted, fivin door, single bay door  | 50 44 55 55 55 55 55 55 55 55 55 55 55 55  | 5552.101       |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 1.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 1.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at, outland lost, 4.5 km, 4.5 km, 4.00 Zelv. Ipprobe serich.  Elect 1.0 gendred at the series of the series o  | 9,600/4/8/7          |            |
| laingle buy door   | 90 443 453 454 454 454 454 454 454 454 454   | 5550401        |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum (2014) profession under diese. 45 Nov. 45 Nov. 1907-200. Ignobel ereich.  Seek 1,0 genineties, underdiese, 45 Nov. 45 Nov. 1907-200. Ignobel ereich.  Seek 1,0 genineties (2) der publisse with monker produce, (2) 8 177-1 top publisse mit misses, (2) Nov. 1907-200.  Feder in this such with (2) der publisse with monker produce, (3) 8 177-1 top publiss mit misses, (3) Nov. 1907-200.  Feder in this such with (2) der publisse with monker produce, (3) 8 177-1 top publis mit misses, (3) Nov. 1907-200.  Feder in this such with (2) der publisse with monker produce, (3) 8 177-1 top publisse mit misses, (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (3) Nov. 1907-200.  Feder in this such with (4) Nov. 1907-200.  Feder in th  | 9,600/4/8/7          |            |
| IV-10  | 90 443 433 443 443 443 443 444 443 443 44  | 5550401        |          | 2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016<br>2016   | Connets Pronfil Sinfress Sinfr | CS DEL        | jedebum  Elect 1.0 gendred at, outland lost, 4.5 km - 1.00 z.60 i ipprofest earlich  Elect 1.0 gendred at, outland lost, 4.5 km - 1.00 z.60 i ipprofest earlich  Elect 1.0 gendred at, outland lost, 4.5 km - 1.00 z.60 i ipprofest earlich  Elect 1.0 gendred at the second  | [3]50314187          |            |

| SUMMAS | RY - SKEENA BI | DENE | RGY L | TD.                     | 1                       | [   | T                 |                     |
|--------|----------------|------|-------|-------------------------|-------------------------|---|-------------------|---------------------|
| item # | Voit#          | IQv. | Year  | Make                    | Model                   | Description   | Serial Humber     | Hex/Kma.            |
| 88     |                |      |       | Britespan               |                         | Portable storage building, 40° X 90°, clear span, metal frame, carross cover, approximatly 220 took block base  |                   | 13-13-14            |
| 89     |                |      |       |                         |                         | 40" X 4" vibratory conveyor, 50 hp 1770 rpm, Square O switch, Newyork Blower size 27Pt,R 25 hp<br>blower (yard storage)   |                   |                     |
| 90     |                |      |       |                         |                         | Palets of steel   |                   |                     |
| 91     |                | 100  |       |                         |                         | Concrete lock blocks  |                   |                     |
| 92     |                |      | 2018  | Almo Aligaier Morgenson | E1026C2                 | Vibratory screen/ policit sifter, 5 X 43 1/2, 1 kw 1152 rpm drive   | 03.01.2561        |                     |
| 93     | 5660 MO1       |      | 2018  |                         |                         | 80' bucket elevator, 15 hp drive, 158 bucket, (op distribution station  |                   |                     |
| 94     |                | 2    | 2018  |                         |                         | 100 (on cap bolt together storage gahanized sidns, 15" dia X 40" T load cell metering, bottom<br>Warmgroup Vt.C0000M1 bottom slide gates, hvin bottom telescopic unloading arms "Cascade",<br>approx. 173" W X 50" steel frame base building support frame with top clad enclosure, catwalk,<br>state, and disconnects. |                   |                     |
| 95     | 3720-8401      |      | 2018  |                         |                         | 55' Twin strand byted bypass conveyor with drive  |                   |                     |
| 96     |                |      |       | Schnell Industries Inc  |                         | Transloader, 4 wheel drive and steer, 35" belt, approx 30 tph, Belt way scale, Kubota mod V3307-<br>CA-TI-EFO2 60 hp engine   | 1113180106        | 5,113 hrs           |
| 97     | FL6K-113       |      |       | Doosan Darwoo           | G30P-3                  | Forkin, 4,600th cap, 3 stage mast, side shift, lpg, cushion tire  | KQ-02156          | 4.138 hrs           |
| 98     | 8165           | L    | 2014  |                         | 924K                    | Front and loader, quick attach, plumbed, fight package, 20 5725 tires, AMI mod Cat Fusion bucket WIL150H0500 s/n 201879-01-01, 5 yd cap   |                   | 12,516 tvs          |
| 99     | 16604          |      | 2014  | Freightliner            | Coronado SD CD122086SDT | Tri-sate walking foor truck, straight cab, Allison 4500 RDS, automatic transmission, Cummins<br>ISX15 SSD bp, 11R22 3 times, aluminum rims, JBS mod FBT762 rear box, 25 twin strand 8yted<br>floor drag takeout transfer, sut on 81 (op. 14 boyt package).  | 3AKRGNBGXEDFPS596 | 723,109 kms, 24,462 |

•

| SUMMA    | SUMMARY - LEASED EQUIPMENT | IPMENT   |  |   |             | - Anamaria   |
|----------|----------------------------|--|--|---|-------------|--|
| item #   | Item # Year Make           | Model  | Description  | Serial Number                           | Hrs/Kms     | Leased/Owned   |
| PARCELA  | T.A.                       |  |  |   |             |  |
|          | 1 2019 Cat                 | 924K   | Front end loader, quick attach, plumbed, light package, 20.5R25 tires, AMI mod Cat Fusion bucket WL150HD600 s/n 201879-01-01, 6 yd cap   | CAT0924KHSNZ00893                       | 15,900 hrs  | Leased - Skeena Bioenergy  |
| PARCEL B | LB.45 (Control of the      | the time of the contract of th | A CARLOS TO THE COME AND A SECTION OF THE SECTION O |   |             |  |
|          | 1 2021 AGI                 | BCX3 1549 FMD TM   | BCX3 1549 FMD TM   Self contained bundle conveyor, 3 wheel, 12" belt, Vanguard EFI 33 hp gas engine  | BA03B122102648                          |             | Leased - Skeena Bioenergy  |
| 8        | 2 2011 Lodeking            |  | Super B hopper bottom tri-axle lead lead with dandem axle pup trailer, rolltop, 38,000 gvwr, 63,000 gvwr   | 2LDHG2838BF051278/<br>2LDHG3029BF051279 |             | Leased - Skeena Bioenergy  |
| PARCELC  |                            | e esta de la companya  |  | Personal Control of the Control         |             | The second of th |
| -        | 1 2020 Cat                 | M996   | Front end loader, AIMI mod CAT 972G 12 yd bucket s/n 173642-01-01, 26.5 R 25 lires, brush guard, light package   | CAT0966MAEJA03350                       | 6,734 hrs   | Leased - Skeena Sawmills   |
| CV.      | 2 2012 Cat                 | 980K   | Front end loader, WBM grapple, quick attach, plumbed, 29.5R25 tires, brush guard, light package  | CAT0980KEW7K00716                       | 23,710 hrs  | Leased - Skeena Sawmills   |
| en .     | 3 2004 Cat                 | 988G   | Front end loader, WBM oversize full grapple, quick attach, plumbed, 35/65R33 tires, brush guard, light package, Cat rebuild 2012 with 15K hrs  | CAT0988GJBNH01119                       | 38,510 hrs  | Leased - Skeena Sawmills   |
| PARCELD  | LO                         |  |  |   |             | 年方里では大学を与るできたとうと   |
|          | 1 2020 Buick               | Encore   | SUV, automatic transmission, 4 door, awd   | KL4CJESBXLB062269                       | 18,900 kms  | Leased - Skeena Sawmills   |
| PARCE    | LESS                       | The second secon | and the second of the second o |   |             | 一年の日本の一年の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の日本の   |
| _        | 2012 Taylor                | TX330M   | Forklift, 33,000 lb cap, cushion tire, diesel, 2 stage, fork positioners   | S-GE-37856                              | 29,672 hrs  | Leased - Skeena Sawmills   |
| 2        | 2 2004 Cat                 | 924G   | Forklift, 15,000lb cap, 2 stage mast, side shift, 17.5R25 tires, light package, cracked window hyd leak  | CAT0924GCRTA00221                       | 29,751 hrs  | Leased - Skeena Sawmills   |
| (F)      | 3 2013 Hyundai             | H160D-7E   | Forklift, 35,273 lb cap, 2 stage mast, fork positioners, light package, cushion tire   | HHKHFT08HD0000955                       | 18,963 hrs  | Leased - Skeena Sawmills   |
| 4        | 4 2011 Cat                 | H056   | Forklift, 30,000 lb cap 2 stage mast, fork positioners, 23.5R25 tires, light package, CAT0950HTJAD01002 cracked window   | CAT0950HTJAD01002                       | 18,324 hrs  | Leased - Skeena Sawmills   |
| 5        | 5 2010 Doosan Daewoo       | woo  D90S-5  | Froklift, 17,850 lb cap, 2 stage mast, fork positioners  | PA-00818                                | 21,447 hrs  | Leased - Skeena Sawmills   |
| Ψ.       | 6 2013 Cat                 | 980K   | Front end loader, WBM grapple, quick attach, plumbed, 29.5R25 tires, brush<br>guard, light package   | CAT0980KCW7K01406                       | 23,327 hrs  | Leased - Skeena Sawmills   |
| 7        | 7 2012 Hyundai             | 110D-7E  | Forklift, 24,000 lb cap, 2 stage mast, fork positioners, led light package needs transmission and fuel pump  | HHKHFT05KC0000188                       | 17,859 hrs  | Leased - Skeena Sawmills   |
| 8        | 8 2013 GMC                 | Sierra 2500 HD   | Pickup truck, reg cab, tool box, Vortec, automatic transmission, 4X4   | 1GT02ZCGXDZ140209                       | 238,238 kms | Leased - Skeena Sawmills   |
| 6        | 9 2011 GMC                 | 2500 HD Silverado  | Pick up truck, crew cab, Vortec, automatic transmission, 4X4   | 1GC1KVCG5BF256025                       | 238,837 kms | Leased - Skeena Sawmills   |
| 10       | 10 2011 GMC                | 2500 HD Silverado  | Pick up truck, crew cab, Vortec, automatic transmission, 4X4   | 1GC1KVCG5BF263802                       | 141,995kms  | Leased - Skeena Sawmills   |
| -        | 11 2018 Dodge Ram          | 2500 HD  | Pick up truck, crew cab, automatic transmission, Hemi 5.7L, 4X4, side steps  | 3C6TR5CT4JG273565                       | 111,515 kms | Leased - Skeena Sawmills   |
| 12       | 2013 GMC                   | Sierra 2500HD  | Pickup truck, reg cab, tool box, Vortec, automatic transmission, 4X4   | 1GT02ZCG2DZ163547                       | 238,233 kms | Leased - Skeena Sawmills   |
| 13       | 13 2018 Dodge Ram          | 2500 HD  | Pick up truck, crew cab, automatic fransmission, Hemi 5.7L, 4X4, side steps  | 3C6TR5CT6JG273566                       | 120,045 kms | Leased - Skeena Sawmills   |
|          |                            |  |  |   |             | F  |